#### **INFORMATION TO USERS**

This reproduction was made from a copy of a document sent to us for microfilming. While the most advanced technology has been used to photograph and reproduce this document, the quality of the reproduction is heavily dependent upon the quality of the material submitted.

The following explanation of techniques is provided to help clarify markings or notations which may appear on this reproduction.

- 1. The sign or "target" for pages apparently lacking from the document photographed is "Missing Page(s)". If it was possible to obtain the missing page(s) or section, they are spliced into the film along with adjacent pages. This may have necessitated cutting through an image and duplicating adjacent pages to assure complete continuity.
- 2. When an image on the film is obliterated with a round black mark, it is an indication of either blurred copy because of movement during exposure, duplicate copy, or copyrighted materials that should not have been filmed. For blurred pages, a good image of the page can be found in the adjacent frame. If copyrighted materials were deleted, a target note will appear listing the pages in the adjacent frame.
- 3. When a map, drawing or chart, etc., is part of the material being photographed, a definite method of "sectioning" the material has been followed. It is customary to begin filming at the upper left hand corner of a large sheet and to continue from left to right in equal sections with small overlaps. If necessary, sectioning is continued again—beginning below the first row and continuing on until complete.
- 4. For illustrations that cannot be satisfactorily reproduced by xerographic means, photographic prints can be purchased at additional cost and inserted into your xerographic copy. These prints are available upon request from the Dissertations Customer Services Department.
- 5. Some pages in any document may have indistinct print. In all cases the best available copy has been filmed.



300 N. Zeeb Road Ann Arbor, MI 48106

.

#### 8418930

Cuevas, Carlos E.

# INTERMEDIATION COSTS AND SCALE ECONOMIES OF BANKING UNDER FINANCIAL REGULATIONS IN HONDURAS

The Ohio State University

Рн.D. 1984

University Microfilms International 300 N. Zeeb Road, Ann Arbor, MI 48106

## Copyright 1984

by

Cuevas, Carlos E.

All Rights Reserved

#### PLEASE NOTE:

In all cases this material has been filmed in the best possible way from the available copy. Problems encountered with this document have been identified here with a check mark  $\sqrt{}$ .

- 1. Glossy photographs or pages \_\_\_\_\_
- 2. Colored illustrations, paper or print
- 3. Photographs with dark background
- 4. Illustrations are poor copy \_\_\_\_\_
- 5. Pages with black marks, not original copy \_\_\_\_\_
- 6. Print shows through as there is text on both sides of page\_\_\_\_\_
- 7. Indistinct, broken or small print on several pages \_/\_\_\_
- 8. Print exceeds margin requirements \_\_\_\_\_
- 9. Tightly bound copy with print lost in spine \_\_\_\_\_
- 10. Computer printout pages with indistinct print \_\_\_\_\_
- 11. Page(s) \_\_\_\_\_\_ lacking when material received, and not available from school or author.
- 12. Page(s) \_\_\_\_\_\_ seem to be missing in numbering only as text follows.
- 13. Two pages numbered \_\_\_\_\_\_. Text follows.
- 14. Curling and wrinkled pages \_\_\_\_\_
- 15. Other\_\_\_\_\_

## University Microfilms International

## INTERMEDIATION COSTS AND SCALE ECONOMIES OF BANKING UNDER FINANCIAL REGULATIONS IN HONDURAS

. .

#### DISSERTATION

Presented in Partial Fulfillment of the Requirements for the Degree Doctor of Philosophy in the Graduate School of The Ohio State University

Ву

Carlos E. Cuevas, B.S., M.S.

\* \* \* \* \*

#### The Ohio State University

1984

Reading Committee:

Professor Douglas H. Graham Professor Dale W Adams Professor Francis E. Walker Approved By

Adviser Department of Agricultural Economics and Rural Sociology

© Copyright by Carlos E. Cuevas 1984 All Rights Reserved

•

To my wife, Sylvia, and my son, Carlos, for their love, support, and understanding

•

.

ii

#### ACKNOWLEDGMENTS

I wish to express my deepest appreciation to Professor Douglas Graham, my adviser and best friend during these four years at Ohio State. I learned from his example, and benefited from his guidance, encouragement, and support throughout this entire period. His insistence on comprehensiveness and completeness, and his thought-provoking questions and comments greatly contributed to the improvement of this dissertation. My family and I enjoyed the kind friendship of the Graham family during all these years. They made us feel respected and appreciated. Our sincere gratitude to all of them.

I am indebted to Professor Dale Adams, who was the man responsible for bringing me to Ohio State. I hope he does not regret it. He was a permanent source of insight and provided valuable criticism and advice during my work at OSU.

I wish to acknowledge the helpful advice of Professor Francis Walker during the course of this study. I also thank Professors Edward Kane and Warren Lee for their valuable comments and suggestions on my dissertation prospectus. Professor Kane's concepts in financial regulation had great influence on my thought and interests.

iii

I wish to thank all faculty and staff of the Agricultural Finance group at the Department of Agricultural Economics for their support and friendship. Special thanks to Mrs. Barbara Lee for her sympathy and cooperation in so many "rush hours" experienced in the fulfillment of project duties. I also wish to acknowledge the support received from all department chairmen during this period, specially Professor David Boyne.

Text and tables of this dissertation were diligently processed by Ms. Jill Loar. Ms. Janice Christensen helped me with the graphs. Thanks to both of them for their careful work.

I am indebted to many people in Honduras. They not only facilitated my data gathering but also contributed to making my work there an excellent professional experience. I am specially grateful to the Office of Food and Agriculture of the USAID Mission in Tegucigalpa, to the Economic Studies Unit of BANADESA, and to the Economic Studies Division of Banco Atlantida.

Finally, my family and I wish to express our gratitude to our relatives and friends in Chile for their permanent encouragement and support. We are also specially grateful to all our friends in Columbus, who have made these years of our lives so enjoyable and rewarding. Sylvia, Carlos and I, we did it.

iv

January	16,	1	.94	6	•	•	•	•	•	Born - Santiago, Chile
1968	••	•	•	•	•	•	•	•	•	B.S., Agronomy (Ingeniero Agronomo), Universidad de Chile, Santiago, Chile
1977	• •	•	•	•	•	•	•	•	•	M.S., Agricultural Economics, Universidad Católica de Chile, Santiago, Chile
1969–197	1.	•	•	•	•	•	•	•	•	Staff member, Department of Technical Assistance and Credit, Instituto de Desarrollo Agropecuario, Puerto Montt, Chile
1971-197	3.	•	•	•	•	•	•	•	•	Head, Department of Liyestock Development, Corporacion de la Reforma Agraria, Santiago, Chile
1974–197	75.	•	•	•	•	•	•	•	•	Advisor, Cooperativa Rupanco Ltda., Osorno, Chile
1976–198	30.	•	•	•	•	•	•	•	•	Research Associate, Department of Agricultural Economics, Universidad Católica de Chile, Santiago, Chile
1980-198	34.	•	•	•	•	•	•	•	•	Research Associate, Department of Agricultural Economics and Rural Sociology, The Ohio State University, Columbus, Ohio

#### PUBLICATIONS

- "Intermediation Costs in an Agricultural Development Bank: A Cost Function Approach to the Measurement of Scale Economies," ESO 1089, Ohio State University, February 1984.
- "Coffee and Basic Grains: Review of Sectoral Performance, Pricing and Marketing Margins, and Recent Policy Changes". A Report to USAID-Honduras, January 1984 (co-author).

### VITA

V

- "Basic Grains: Marketing Channels and Financing at the Farm and Wholesale Levels". A Report to USAID-Honduras, January 1984 (co-author).
- "Level and Structure of IHCAFE Costs Associated with Coffee Loan Activities". A Report to USAID-Honduras, January 1984 (co-author).
- "Loan Targeting and Financial Intermediation Costs in Honduras", ESO 1072, Ohio State University, December 1983. Presented at the <u>ASSA Meetings 1983</u> (co-author).
- Bernormannia and State State
- 7. "Desarrollo del Sistema Financiero y Movilización de Ahorros: La Experiencia de Honduras en el Período 1970-1981." (Development of the Financial System and Savings Mobilization: The Honduran Experience in the Period 1970-1981). ESO 1020, Ohio State University, April 1983 (co-author).
- 8. "Lending Costs and Rural Development: Is Cheap Credit Really Cheap?" ESO 1002, Ohio State University, March 1983, Selected Paper <u>AAEA Meetings 1983</u> (co-author). (Also in Spanish). Forthcoming in <u>Savings and</u> <u>Development</u>, 8(2), 1984.
- 9. "Interest Rates Policies and Borrowing Costs in Rural Financial Markets." ESO 952, Ohio State University, 1982, Selected Paper <u>AAEA Meetings, 1983</u> (co-author). (Also in Spanish).
- "BANCO ATLANTIDA: Nivel y Estructura de los Costos de Prestamos" (Level and Structure of Lending Costs). A Report to USAID-Honduras, October 1982 (co-author).
- 11. "BANADESA: Nivel y Estructura de los Costos de Préstamo: Implicaciones para Política Crediticia y Organización Interna" (Level and Structure of Lending Costs: Implications for Credit Policy and Internal Organization). A Report to USAID-Honduras, September 1982 (co-author).
- 12. "BANADESA: The Delinquency Question and Lending Costs," Chapter V in Ohio State University Report to USAID-Honduras, An Assessment of Rural Financial Markets in Honduras. Dec. 1981 (co-author).

- 13. "Borrowing Costs and Formal and Informal Credit Activity at the Farm Household Level," Chapter VI in Ohio State University Report to USAID-Honduras, <u>An</u> <u>Assessment of Rural Financial Markets in Honduras</u>. Dec. 1981 (co-author).
- 14. "The Role of Women as Borrowers of Agricultural Credit in Honduras: Selected Case Studies," Ohio State University Report to USAID-Honduras. Dec. 1981 (co-author).
- 15. "Sectoral Demand for Credit and Credit Policy (Chile)." Paper presented at the <u>Second International Conference</u> on Rural Finance Research Issues. Calgary, Alberta, Canada. August, 1979.
- 16. "Oferta Tradicional y Comercial de Granos Básicos. Comentarios" (Traditional and Commercial Supply of Basic Grains. Comment). <u>Cuadernos de Economía</u> No. 46, December 1978).
- 17. "Un Modelo Sectorial para la Agricultura Chilena" (A Sectoral Model for Chilean Agriculture), with Emilio Francisco and J.C. Fernandez. XV Meeting of the Pan American Union of Engineers (UPADI). October 1978, published in <u>Ciencia e Investigacion Agraria</u>, March 1979).
- "Credito Agropecuario en Chile" (Agricultural Credit in Chile). Thesis for the Mag-Sci. Degree, Universidad Catolica de Chile, 1977.

FIELDS OF STUDY

- Major Field: Agricultural Economics
  - Economic Theory. Professors Stephen A. McCafferty, Gene E. Mumy, and Edward J. Ray.

Econometrics and Quantitative Methods. Professors Richard G. Anderson, Leroy J. Hushak, Francis E. Walker, and D. Ransom Whitney.

- Development Economics and International Trade. Professors Douglas H. Graham, Albert E. Keidel, Marie C. Thursby, and Thomas A. Wolf.
- Agricultural Finance, and Money and Banking. Professors Dale W Adams, Douglas H. Graham, and Edward J. Kane.

### TABLE OF CONTENTS

	Page
DEDICATION	, ii
ACKNOWLEDGMENTS	, iii
VITA	. v
LIST OF TABLES	. xii
LIST OF FIGURES	, xvii
Chapter	
I. INTRODUCTION	. 1
Overview	. 1
Statement of the Problem	. 3
Objectives	. 6
Organization of the Study	. 7
II. BACKGROUND: THE FINANCIAL SECTOR OF HONDURAS	. 9
2.1. Size and Composition of the Financial System	. 11
2.2. Development of the Financial Sector	. 15
2.3. Indirect Policy Impacts on Financial Deepening	. 21
2.4. Financial Sector Regulations and Financial Deepening	. 25
III. THEORETICAL FRAMEWORK	. 34
3.1. The Functions of Financial Intermediaries and Transaction Costs	. 36
3.2. Lender's Intermediation Costs	. 42

.

## Chapter

		3.3.	Cr	edi	t F	۱at	:i	oni	ing	3,	I	mp	li	ci	t	Pr	ic	in	ıg	an	đ		
			Bo	rro	wir	ıg	C	ost	ts.	•	•	•	•	•	•	•	•	•	•	•	•	•	46
IV.	MEI	THODOL	IOG	¥.	• •		•	•	• •	• •	•	•	•	•	•	•	•	•	•	•	•	•	54
		4.1.	1	Len	der	:' :	3 (	Cos	зt	Fu	in	ct	io	n	•	•	•	•	•	•	•	•	54
		4.1.1	•	Spe	cif Ecc	ic onc	ca om:	tic ies	on s c	of	S	• ca	le	•	•	•	•	•	•	•	•	•	55 59
					of Ela	Sc Sc ast	co coj	pe) cit	) . ) .	of	=11 • E :	ιa • Su	bs	ti	.es .tu	ti	.on	• •	ind		•	•	63
					Ela	ist	ti	cit	tie	es	0	£	Fa	ct	or	D	em	an	ıđ	•	•	•	64
		4.1.2	2.	Out	put	: [	)e:	fir	nit	tic	on	•	•	•.	•	•	•	•	•	•	•	•	65
		4.1.3		Pro	xie	es	f	or	Re	egι	11	at	io	n	•	•	•	•	•	•	•	•	69
					Int	:ei	cea ma	st-	-Ra	ate -ir	) 2 2	Ce	il nð	in	igs	ci	• al	•	• 'ro	đi	•	•	69
					Pro	oj€	9C	ts.	•	• •	•	•	•	•	•	•	•	•	•	•	•	•	71
		4.1.4	<b>.</b> 1	Del	inc	ļu€	en	су	ar	nđ	D	ef	au	lt		•	•	•	•	•	•	•	74
		4.1.5	<b>.</b> 1	Dat	aa	ind		Est	tin	nat	ti	on	•	•	•	•	•	•	•	•	•	•	76
		4.2.		Bor	rov	iei	:'	s (	Cos	sts	3	•	•	•	•	•	•	•	•	•	•	•	80
		4.2.1	- •	Tra	nsa	act	ti	on	Co	ost	ts		•	•	•	•	•	•	•	•	•	•	81
		4.2.2	2.	A T	rad	le-	-0	ff	Ec	qua	at	io	n nd	be	tw	ee	n	~ ~					
			1	Rat	inse ie d		•	•			•	.а	•	•	•	•	•	•	•	•	•	•	82
					Cha	ara	aC.	tei	cis	sti	ic	S	of	- t	:he	L	oa	n					~ ~
					Ope	era eci	if:	101 102	n. ati	i or	• 1	•	•	•	•	•	•	•	•	•	•	•	82 86
										- • ·	-	•	-	•	•	•	•	•	•	•	•	•	
		4.2.3	3.	Dat	aa	and	i E	Est	tir	na t	ti	on	•	•	•	•	•	•	٠	•	•	•	96
۷.	LEN	IDER'S	S I	NTE	RME	DI	E A!	TIC	DN	CC	)S	TS	•	•	•	•	•	•	•	•	•	•	98
		5.1.		Ove	rvi	Lev	1	of	Re	esi	11	ts	•	•	•	•	•	•	•	•	•	•	99
		5.1.1	•	The Ban	Co	ost	t :	Fui	nct	tic	on	0	f	th	ne	De	eve	10	pm	en	t	_	99
					Sir	ng l	le	-E¢	qua	ati	io	n	an	d	Sy	st	en	•	•	•	-	-	102
					Efi	fec	ua ct:	5 ( 5	of	L	• ba	• n–	Si	.ze	e a	nc	ľ	•	•	•	•	•	105
					net	208	31	C-3	512	ze.		•	•	•	•	•	•	٠	•	•	•	•	TOD

іx

Page

#### Chapter

#### 5.1.2. The Cost Function of the Private 112 5.1.3. General Findings and Contrasts, A 123 5.2. Economies of Scale, Cost Complementarities and Factor Substitution . . . . . . . . . . . . . . . 131 Economies of Scale. . . . . . . 131 Cost Complementarities (Economies of Scope), and Elasticities of Factor Substitution and Factor 140 5.3. Interest-Rate Regulations and Lender's Intermediation Costs . . . . . . . . . 143 Loan-Targeting, Portfolio Composition 5.4. and Lender's Intermediation Costs. . . 153 Sources of Funds and Portfolio Composition . . . . . . . . 154 Effects of Loan-Targeting on the Costs of the Development Bank . . . 157 5.5. Loan Delinquency and Intermediation Costs in the Development Bank. . . . 160 Delinquency Records and Measurement Problems. . . . . . . . . . . . . . . 161 Effects of Delinquency on Intermediation Costs. . . . . . . 165 VI. TRANSACTION COSTS AND BORROWING COSTS. . . . . 171 171 6.1. Overview of Survey Results . . . . . 6.1.1. Overall Sample Results . . . . . . . 173 6.1.2. Different Loan Sizes . . . . . . . . . 177 179 6.1.4. Different Lenders. . . . . . . . . . . . . 181 6.2. The Transaction-Costs Function . . . . 183 6.2.1. Transaction Costs and Loan Size. . . . 194 6.2.2. Trade-off between Transaction Costs 204

Page

Chapter																			Page
VII. SU	JMMARY	AND	CON	ICLUS	ION	s.	•	•	•	6	<b>o</b>	. 0	•	•	•	•	•	•	215
	Resul	ts a	and	Impl	ica	ti	ons	5.	•	•	•	•	•	•	•	•	٠	•	217
	Concl	usid	ons	• •	•••	•	•	•	•	•	•	•	•	•	•	٠	•	•	227
BIBLIOGE	RAPHY .	•	••	••	•••	•	•	•	•	•	•	•	•	•	•	•	٠	•	229
APPENDIX	KES																		
A. Ag	ppendix	to	Cha	pter	11	•	•	•	•	•	•	•	•	•	•	•	•	•	238
B. Ag	ppendix	to	Cha	pter	٠IV	•	•	•	•	•	•	•	•	•	•	•	٠	•	245
C. Ag	ppendix	to	Cha	pter	v.	•	•	•	•	•	•	•	•	•	•	•	٠	•	253
D. Ag	ppendix	to	Cha	pter	vı	•	•	•	•	•	•	•	•	•	•	•	٠	٠	264

•

## LIST OF TABLES

Table		Page
1.	Ratio of Money Stock (M2) to Gross National Product (GNP): Honduras and Other Selected Countries, 1960-1982	17
2.	Deposit-Rate Ceilings Adjusted by Reserve Requirements, Lending-Rate Ceilings, and Gross-Spread Implicit in Interest-Rate Regulations and Reserve Requirements, 1970-1982	31
3.	Development Bank: Summary of R-square Values Obtained in the Estimation of the Cost Function, Under Different Output Definitions and Functional Forms	101
4.	Development Bank: Estimated Parameters of the Cost Function. Output Defined as Number of Loans $(q_1)$ and Number of Deposit Accounts $(q_2)$ . Single Equation versus System Estimation	103
5.	Development Bank: Estimated Parameters of the Cost Function. Output Defined as Value of Loans $(q_1)$ and Deposit Balances $(q_2)$ . Single Equation versus System Estimation.	104
6.	Development Bank: Estimated Parameters of the Cost Function, Including Loan-Size and Deposit-Size Effects. System Estimation with Two Output Definitions	106
7.	Private Bank: Summary of R-square Values Obtained in the Estimation of the Cost Function, Under Different Specifications	113
8.	Private Bank: Estimated Parameters of the Cost Function, Single-Equation <u>versus</u> System Estimation	115
9.	Private Bank: Estimated Parameters of the Cost Function, including Deposit-Size Effects. Single-Equation <u>versus</u> System Estimation	119

10.	Lender's Intermediation Costs and Costs of Mobilizing Deposits. Summary of Findings for the Development Bank and the Private Bank	124
11.	Development Bank: Scale-Economies Estimates (ES), Under Different Output Definitions, Functional Forms, and Estimation Procedures. Tests of Null Hypothesis ES=1, and R-squares of Underlying Estimated Equations	132
12.	Private Bank: Scale Economies Estimates (ES), Under Different Functional Forms, and Estimation Procedures. Tests of Null Hypothesis ES=1, and R-squares of Underlying Estimated Equations	135
13.	Estimated Values of Partial and Overall Economies of Scale (ES), at the Sample Means and at Different Branch-Sizes: Development Bank and Private Bank	138
14.	Cost Complementarities in Production (Economies of Scope), Elasticities of Factor Substitution, and Price-Elasticities of Demand for Factors of Production, Derived from Cost-Function Estimates: Development Bank and Private Bank	142
15.	Effects of Interest-Rate Regulations on Lender's Intermediation Costs: Estimated Parameters of the Real-Deposit Rate, and the Real Lending-Rate in Different Equations. Cost-System Estimates for the Development Bank and the Private Bank .	148
16.	Effects of Loan-Targeting on the Intermediation Costs of the Development Bank: Estimated Coefficients for Different Sources of Funds	159
17.	Delinquent Balances and Refinanced Loans as a Proportion of Total Loans Outstanding in the Development Bank, 1970-1982	162
18.	Composition of Delinquent Balances by Age in the Development Bank, 1972-1982	166
19.	Effects of Delinquency on the Intermediation Costs of the Development Bank: Estimated Coefficients for Current and Lagged Effects	168
20.	Mean and Median Values of Transaction Costs and Total Borrowing Costs (Per Loan and Per Lempira) for Total Sample of Borrowers	175

21.	Components of Total Borrowing Costs, by Loan Size	178
22.	Components of Total Borrowing Costs, by Farm Size	180
23.	Borrowing Costs Associated with Different Institutional Sources of Credit	182
24.	Estimated Coefficients of the Transaction-Costs Function, Under Different Model Specifications. Dependent Variable: lnT	186
25.	Tests of Significance of the Effects of Loan Source and Loan Use on Borrower's Transaction Costs. Values of the F-Statistics for Joint Null Hypothesis, Under Different Model Specifications	192
26.	Transaction Costs per Lempira $(\tau)$ : Estimated Elasticities of $\tau$ with respect to Changes in the Explicit Interest Rate (i). Differences Between Lenders and Loan-Size Categories	206
27.	Transaction Costs per Lempira $(\tau)$ : Estimated Values of $\tau$ , Changes in $\tau$ and in Total Borrowing Costs (i+ $\tau$ ) with Increases in the Explicit Interest Rate (i), by Lender and Loan Size	209
Append	lix A	
28.	Honduras: Size and Composition of the Financ'al System. Number of Institutions and Number of Offices of Main Financial Institutions. Selected Years, 1950-1982	240
29.	Relative Importance of Commercial Banks, Development Banks, and Other Financial Institutions in the Banking System. Proportion of Total New Loans Granted, and Shares in Overall Deposit Mobilization. Selected Years, 1960-1982	241
30.	Interest-Rate Ceilings: Nominal and Real Rates on Savings and Time Deposits, 1970-1982.	

Page

31.

32.

Appendi

33.

34.

35.

36.

37.

•

Interest-Rate Ceilings: Nominal and Real Lending Rate Ceilings, 1970-1982. Real Rates Computed for Two Deflators: The Consumer Price Index (CPI) and the Implicit GDP Deflator (ID).	243
Central Bank Rediscount Lines: Number of Rediscount Lines, Discount Rates and Associated Lending Rates of Main Rediscount Lines, 1970-1982	244
lix C	
Development Bank: Estimated Parameters of the Cost Function, for Different Output Definitions and Functional Forms. Dependent Variable: Administrative Costs (lnC)	254
Development Bank: Estimated Parameters of the Cost Function, for the Single-Output Definition (Loans + Deposits), and Different Functional Forms. Dependent Variable: Administrative Costs (lnC)	255
Development Bank: Estimated Parameters of the Cost Function, Including Loan-Size and Deposit-Size Effects. Single-Equation Estimation with Two Output Definitions	256
Private Bank: Estimated Parameters of the Cost Function, for Different Functional Forms. Dependent Variable: Administrative Costs (lnC).	257
Private Bank: Correlation Coefficients Between Selected Variables of the Cost Function. All Variables in Natural Logarithms	258

- 38. Means, Standard Deviations, and Correlation Coefficients for Interest-Rate Ceilings and Implicit Rates Paid on Deposits. Nominal and Real Terms. Results for the Development Bank 259
- Shares of Different Sources of Funds in the 39. Portfolio of New Loans of the Agricultural Development Bank (BANADESA), and Average Loan Size by Source, 1971-1982..... 260
- Shares of Agricultural Loans in the Portfolio 40. of New Loans of the Development Bank, 261

Page

.

-

41.	Correlation Coefficients Between Sources of Funds and New Loans to Agriculture in the Development Bank	262
42.	Correlation Coefficients Between Shares of Sources of Funds in the Loan Portfolio of the Development Bank	263
Append	lix D	
43.	Distribution of Loans by Loan Size, Mean and Median Values by Loan-Size Category, by Source of Loan. Total Sample	265
44.	Farm-Size Distribution of Loans Granted by the Development Bank, by Loan Size	266
45.	Farm-Size Distribution of Loans Granted by Private Banks, by Loan Size	267
46.	Farm-Size Distribution of Loans Granted by Credit Unions, by Loan Size	268
47.	Estimated Coefficients of the Transaction-Costs Function, with Alternative Proxies for Firm Size. Depndent Variable: lnT	269
48.	Transaction Costs per Lempira $(\tau)$ : Estimated Values of $\tau$ , Changes in $\tau$ and in Total Borrowing Costs $(i+\tau)$ with Increases in the Explicit Interest Rate (i). Results by Loan Use and Loan Size for the Development Bank	270
49.	Transaction Costs per Lempira $(\tau)$ : Estimated Values of $\tau$ , Changes in $\tau$ and in Total Borrowing Costs $(i+\tau)$ with Increases in the Explicit Interest Rate (i). Results by Loan Use and Loan Size for Private Banks.	271

xvi

## LIST OF FIGURES

Figure	Page
1. Transaction Costs per Loan by Loan Size, for Different Loan Sources	197
2. Transaction Costs per Lempira by Loan Size, for Different Loan Sources	199
3. Transaction Costs per Lempira by Loan Size, for Different Levels of the Interest Rate (i)	201
<ol> <li>Trade-off Between Transaction Costs per Lempira and the Interest Rate, for Different Loan Sizes (L)</li></ol>	203
5. Trade-off Between Transaction Costs per Lempira and the Interest Rate, for the Development Bank and Private Banks, by Different Loan	
Sizes (L)	213

#### CHLPTER I

#### Introduction

Financial markets play a crucial role in resource allocation and distribution. The efficiency with which financial markets operate determines the magnitude of the contribution of this sector to economic development. The intrinsic nature of finance, however, makes it difficult to assess the efficiency of financial-market performance. Transaction costs of financial intermediation provide an analytical tool to evaluate this performance. Transaction costs are an appropriate measure of the degree of "friction" existing in the functioning of financial markets. The higher the transaction costs of financial intermediation, the less efficient the performance of financial markets, and the more constrained their contribution to development.

This study is a comprehensive analysis of the costs of financial intermediation in a small, less developed country, Honduras. The transaction costs of intermediation borne by lenders and borrowers are measured and analyzed with particular (though not exclusive) attention devoted to agricultural lenders and borrowers. Particular emphasis is placed on the cost structure and scale economies of

financial institutions. The effects that financial market regulations have on the magnitude and the distribution of financial intermediation costs are also investigated.

This Introduction presents an overview of the issues involved, states the problem addressed and the objectives pursued. The organization of the study is set forth at the end of this chapter.

#### Overview

Over the last decade, much effort has been devoted to the clarification and understanding of the roles of money, capital, and finance in development. In less developed countries (LDCs), the contribution of money to development relies primarily on its role as a channel of resources from savers to investors [56]. This function has been the focus of many research efforts.

With the clarification of the contribution that finance makes to development and of the roles performed by financial intermediaries, concern has surfaced regarding the effects of economic policies on the financial sector. Since financial intermediaries channel claims on real resources between savers and investors, policies that affect the functioning and performance of the financial sector will have an impact on the allocation of real resources and on the distribution of income [1,32,57,74]. As a consequence of the non-neutrality of the incidence of financial policies, it is in the interest of different groups in society to

attempt to influence the direction of these policies. Thus government intervention in financial markets tends to be ultimately decided by the relative strength of different political constituencies, constraining the economist's role to that of "sorting out" the likely consequences of alternative intervention schemes [52].

"To intervene or not to intervene" has been a question addressed by many researchers, both with respect to financial markets in general, and to credit allocation schemes in particular. Selective credit policies have been utilized extensively in developed countries and advocated for LDCs, on the assumption that market imperfections and fragmentation exist, and that there is a need to induce important shifts in economic relationships and technological change [57,77,86]. The results of these policies in LDCs have generated various controversies. These debates usually contrast the costs and effects of financial market regulation against alternative policy measures such as the establishment of specialized financial institutions that would presumedly perform the allocational and developmentpromoting roles attributed to selective credit controls [2,57,86].

#### Statement of the Problem

Financial intermediaries in LDCs are usually highly regulated. Controlled interest rates creating fragmented

markets for loans and deposits are among the preferred regulations utilized by governmental authorities. At the same time, domestic central banks and international agencies impose various targeting and end-use requirements for funds lent to different sectors of the economy. Special credit projects and specialized lending institutions complete the typical financial scene in these countries [2,23]. Specialized lending institutions (development banks) have received special attention particularly from international funding agencies, that have seen them as a financial innovation capable of increasing internal and external finance to selected sectors such as agriculture [20,23,41,93].

A number of studies have analyzed several features of intervention schemes in rural financial markets, and their effects on resource allocation and income distribution. $\frac{1}{2}$ These studies suggest that very often the intended effects of regulations are rarely attained, and that many unintended and undesirable results occur. The diffused nature of financial activities and the limited development of institutional and informational systems in LDCs have made it difficult for advocates and detractors of governmental intervention in rural financial markets to provide strong empirical support for their views.

Financial intermediation is far from costless, and financial policies impinge upon these costs in several ways

<sup>1/</sup> A collection of essays on this subject is forthcoming in Adams, Graham, and Von Pischke [5].

ſ11. Financial intermediaries use the services of factors of production according to particular technologies, therefore resource costs are necessarily involved in the process. Moreover, the quantity and quality of resources as well as the technologies utilized are subject to changes. Financial policies in general, and regulatory policies in particular, create incentives for innovations and organizational adjustments in the financial sector. These innovations and adjustments are justified more often by their "productivity in regulatory avoidance" than by their cost-reducting effectiveness (Kane, [53]). In fact, many technological and organizational changes in the financial system may have increased intermediation costs instead of reducing them. However, few attempts have been made to evaluate these costs, estimate the effects of policies on their magnitude and the incidence of these costs among participants [79].

This study analyzes the costs of financial intermediation in a small, less-developed economy, Honduras, where most of the characteristics of the typical environment depicted above are present [36]. The chain that links savers with investors is conceived as a sequence - savings mobilization, lending, borrowing - that demand the use of real resources and therefore involve costs. It is hypothesized that both the magnitude of these costs as well as their relative incidence among lenders and borrowers are affected by the regulated environment. The characteristics

and behavior of commercial banks and development banks will be contrasted, with the aim of providing additional insights into the behavior of specialized lending institutions, particularly those engaged in agricultural lending.

#### Objectives

The aims of this work are two-fold: first, to document and evaluate the costs involved in the process of financial intermediation. These include the costs incurred by lenders in mobilizing funds, the operational costs of on-lending these funds to clients, as well as the costs incurred by final borrowers securing loans from institutional sources. The second objective is to determine the impact of regulatory policies on the costs of financial intermediation. Interest-rate regulations and selective credit policies will receive special attention.

Specific objectives of the study are as follows:

- 1. To measure the average costs and the marginal costs of loans granted and deposits mobilized, as well as the overall intermediation costs for the National Agricultural Development Bank (BANADESA) and a large private bank in Honduras;
- 2. To assess the scale economies and scope economies of banking, and other related characteristics of the underlying technology of financial services provided by these institutions;

- 3. To estimate the impact of interest-rate policies and loan-targeting schemes on the intermediation costs of lenders;
- 4. To examine to what extent 2, and 3 above differ between a private commercial bank and a public-sector agricultural development bank;
- 5. To investigate the nature and magnitude of borrowing costs borne by agricultural borrowers in Honduras, specifically the non-interest transaction costs involved in obtaining a loan;
- 6. To ascertain the effects of the explicit interest rate charged by lenders, the loan size, and risk characteristics of the farmer-client on his non-interest transaction costs; and
- 7. To analyze the differences in borrowing costs (explicit and implicit) associated with different loan sources (i.e. the agricultural development bank vs. other lending institutions).

#### Organization of the Study

An overview of the Honduran financial system from 1960 to 1982 is presented in chapter 2. Policies and regulations that have affected financial deepening in the Honduran economy are reviewed and analyzed. Chapter 3 develops the theoretical framework for the study. The functions of financial intermediation, the theory of the banking firm, and theoretical elements of credit rationing models make up this conceptual framework. The methodology used to measure and evaluate the costs of intermediation for lenders and borrowers is presented in chapter 4. This includes econometric analysis of the lenders' cost functions and the transaction-costs of borrowing.

The principal results are presented in chapters 5 and 6. Chapter 5 sets forth the analysis of lenders' costs, the estimation of cost functions, and the estimated effects of regulations on financial intermediaries. The magnitudes of deposit-mobilization and lending costs are documented and discussed here for the agricultural development bank of Honduras and the largest private commercial bank in the country. Scale-economies and scope-economies are then estimated and contrasted between the two banks. Chapter 6 concentrates on the results obtained in the evaluation of the transaction costs of borrowing. The magnitude of these costs, and the factors affecting this magnitude and their incidence in the agricultural loan portfolio are discussed. Emphasis is placed on the trade-off that exists between implicit, non-interest transaction costs and the explicit interest rate. Finally, chapter 7 summarizes the major findings and outlines important policy implications and conclusions.

#### CHAPTER II

Background: The Financial Sector of Honduras

Honduras, with a population of about 4 million inhabitants and a size comparable to the state of Ohio, is one of the countries with the lowest income per capita in Latin America (US\$ 660 in 1982). $\frac{1}{2}$  Like many other countries in the area, Honduras is an open economy, with agricultural exports the basis of the country's foreignexchange earnings. $\frac{2}{}$  Agriculture contributes roughly 30% of GDP and receives about one-fourth of the total credit of the banking system. Also, Honduras exhibits the main features of the financial structures of low income countries as pointed out by McKinnon [66]: (1) limited issuance of primary securities (bonds, stock) by individual economic units, and consequently a greater reliance on self-finance by these economic units, as compared to more developed countries; (2) most primary securities are acquired by financial intermediaries, rather than being purchased directly by savers; and (3) most claims on financial institutions held by the public are accounted for by the 1/ Selected indicators are presented in appendix A.

<sup>2/</sup> A thorough description of the Honduran economy and financial markets is found in Graham <u>et al</u>. [36], specially chapters 1 and 2 [33].

liabilities of the monetary system, i.e, the central bank and deposit banks. Under these conditions, the major role in the intermediation of financial assets between savers and investors corresponds to the financial sector. These financial assets consist mainly of currency and deposits, i.e., claims on the central bank and the depository institutions comprising the financial system.

This chapter reviews the main features of the Honduran financial system, emphasizing developments in the 1970s and early 1980s. $\frac{3}{}$  The first section describes the historical evolution of the size and composition of the banking system. The moderate expansion and subsequent stagnation of the financial sector after 1976 stand out in the discussion presented in this section. Section 2.2 analyzes the development of the financial system of Honduras on the basis of the most commonly accepted indicators of financial deepening. A pattern of growth, stagnation, and decline of the financial sector emerges as a result of examining the behavior of the ratio of the sector's liabilities (M2) to the Gross National Product during the period 1970-1982. However, financial deepening in Honduras compares favorably to other Central American or other Latin American countries. Finally, the regulatory environment under which the Honduran financial

<sup>3/</sup> Even though informal credit markets exist in Honduras, particularly in the rural areas (see Graham et al. [36]), in what follows the term financial sector is applied to the complex of institutions that comprise the Honduran "formal" financial system.

system has evolved is reviewed in sections 2.3 and 2.4. Interest-rate regulations, rediscount policies, and reserve requirements are emphasized as indicators of the degree of regulation and fragmentation of the financial system. These regulations provide various incentives for the financial institutions to engage in regulatory-avoidance behavior. Thus the costs of avoidance or adjustment to the changing regulatory environment should be considered in addition to the costs of complying with these financial-market regulations.

2.1. Size and Composition of the Financial System

The Honduran banking system is less than a century old. The first financial institutions, created in 1888, were comprised of two private banks authorized to issue paper currency subject to the supervision of the Ministry of Finance [97].4/ These banks merged in 1889 to become the "Banco de Honduras", with official endorsement to engage in currency-emission. In 1912, "Banco Atlantida" and "Banco de Comercio" were created, the latter being acquired by the former in 1917. From this period until 1950 Banco de Honduras and Banco Atlantida were the only two banks with authority to issue paper money in the country. This authority was then transferred to the Central Bank, founded in 1950. At that time, only four financial institutions

<sup>4/</sup> A comprehensive review of the history of the Honduran banking system until 1968 is found in Yu-Shan [97]. These initial paragraphs rely heavily upon this work.

were recognized, the two banks mentioned above with a total of seven offices, and two single-office savings banks. These offices were located in the two main cities of the country (Tegucigalpa and San Pedro Sula), and in port-towns along the northern coast, where banana operations were (and still are) concentrated. $\frac{5}{}$ 

The continuing evolution of the Honduran banking system in terms of the number of financial institutions and the number of offices, from 1950 through 1982, is summarized in appendix A, table 28. Financial institutions (not including the central bank) have been classified here between private commercial banks, public development banks, and other (private or public) financial institutions that comprise savings and loans associations and "financieras". The ratio between total population and the total number of offices of all financial institutions is presented in this table, to serve as a crude proxy for real growth of the (physical) size of the financial network.

Between 1950 and 1968 the system experienced a 9-fold expansion in the number of offices, with the creation of 6 new private commercial banks, and the formation of the first public financial institutions. Of the latter, the most important was the "Banco de Fomento" (currently the "Banco Nacional de Desarrollo Agricola", BANADESA). With this

<sup>5/</sup> In fact, Banco-Atlantida's main office was established in La Ceiba, then the most important port for banana exports.
expansion, the ratio of total population to the number of offices of financial institutions dropped from 160 thousand inhabitants per office in 1950 to roughly 28.2 thousand in 1968. The total number of offices of the financial system continued growing between 1968 and 1976, from 82 offices to 212, further reducing the population-to-institution ratio to 15.1 thousand inhabitants per office in 1976.

Overall, the financial system has experienced rather limited growth in terms of number of offices since 1976, becoming practically stagnant after 1980. The lowest ratio of population to number of offices was attained in 1980, 13.3 thousand inhabitants per office, before the failure of "Banco Financiera Hondureña" at the end of this year. Private banks accounted for most of this expansion in the second half of the 70's, showing an increase in the number of offices from 175 in 1976 to a peak of 218 in 1980. Only two new offices of development banks were created during this period, while the most striking growth in relative terms corresponds to other financial institutions. These institutions show a 3-fold increase in the number of offices between 1976 and 1980, thus accounting for one third of the overall increase in the size of the financial system between 1976 and 1980. The expansion of these other financial institutions finally terminates, as does the rest of the system, after 1980. The total number of offices has slightly decreased between 1980 and 1982. As a consequence

the ratio of population to the number of offices in 1982 remains at the same level as in 1978.

As the foregoing discussion suggests, the expansion of private commercial banks have been the predominant source of growth within the financial institutions since its incep-Table 29 in appendix A provides further evidence of tion. this dominance, both on the asset and the liability side of the system's balance sheet. Private commercial banks have accounted for more than 80 per cent of total lending since 1970, with the exception of 1978 (79%). These banks have also accounted for more than 86 per cent of total deposits mobilized in the system. The development banks' share of total lending in the 1970s and early 1980s has fluctuated between 8 per cent (1976) and 16.6 per cent (1980). However, the development banks' share in total lending in 1982 (11.7 percent) is only slightly larger than their share in 1970 (10.4%), and about half the level attained in 1965 (21.2%).

The role of development banks in deposit mobilization has been of little importance and stagnant, accounting for only 6 percent of total deposits mobilized in the system in the period 1970-1982, despite the fact that development banks are the "preferred" depository institution for public sector entities. Between 40 and 50 percent of public-sector deposits go to development banks. In turn, for these development banks as a group, reliance on public-sector deposits

has increased from 6.4 percent of total deposits in 1970, to 47 percent of the deposit-portfolio in the early  $1980s.6^{-/}$ The increased role of other financial institutions in total deposit-mobilization observed in table 29 has occurred mainly at the expense of the savings-accounts balances of private commercial banks. $7^{-/}$  As a group, these other financial institutions accounted for an average of 8 percent of total deposit mobilization in recent years.

In summary, the Honduran financial system experienced sustained growth in the network of branches since 1950 until the mid 1970s. This expansion reduced to a moderate pace during the second half of the 1970s and came to a stop and remained stagnant after 1980. The private commercial banks dominated the system throughout the period. Therefore, the development of the financial sector discussed in the following section corresponds essentially to changes experienced in the private-bank area.

## 2.2. Development of the Financial Sector

In this section the ratio of the money stock (M2) to the Gross National Product (GNP) is utilized as an indicator of the degree of monetization or "financial deepening" [66,83]. This ratio also represents an alternative way of defining the size of the financial sector, in terms of its

<sup>6/</sup> Figures computed from data reported in the Central Bank Statistical Bulletin [9].

<sup>7/</sup> See Cuevas and Vogel [27].

total "output" instead of the physical size of the "plant" (i.e., branch network), considered in the preceding section. More importantly, the behavior of the ratio of M2-to-GNP is an indicator of the degree of financial repression; the more repressed the financial sector of an economy, the more constrained is the growth of the stock of financial assets relative to income [53].

The behavior of the ratio of M2-to-GNP in Honduras is contrasted against those of other countries in table 1. In this table, the money stock (M2) as a percent of GNP is presented for Honduras, row 1, other Central American countries, rows 2 through 4, other Latin American countries, rows 5 through 8. A textbook case of succesful financial reform and substantial growth of the financial system (South Korea) is included in row 9 of the same table, and two industrialized economies are included in rows 10 and 11 for comparative purposes. Ratios of M2-to-GNP are presented in this table for 1960, 1965, and for the period 1970-1982.

The degree of monetization of the Honduran economy, and all less-developed economies included in table 1 (rows 1 through 9) have always been below those of Japan and the United States(rows 10 and 11). In the early 80's, the ratios of M2 over GNP of the less-developed economies shown in table 1 were between 23% (Guatemala), and 41% (South Korea), whereas those of the United States were around 62%,

•••••							Yei	ar, M2	as Perc	ent of (	GNP					G	NP per capita in 1982
Country		1960	1965	1970	1971	1972	1973	1974	1975	1976	1977	1978	1979	1980	1981	1982	(US\$ 1982)_d/
1.	Honduras	14.61	17.55	21.61	22.74	24.65	26.82	24.60	26•55	30.11	30.14	31.75	26.70	25.98	26•72	30.64	660•1
2.	Guatemala	13.63	17.36	18.79	20.07	23.65	23.61	21.85	23.25	25.45	23.96	24.83	23.36	22.66	23.19	N•@•	1144.4
3.	Nicaragua <mark>a/</mark>	12.72	17.29	15.75	17.01	20.70	23.73	19.80	18.94	21.69	18.94	18.43	n.a.	n.a.	n•a•	n.a.	850.0
4.	Costa Rica <sup>b/</sup>	19•77	20.36	19.84	25.73	27•20	26•73	26.86	30.16	32.87	33.74	37•78	44.64	43.72	£.5•	∏+ā+	1032.1
5.	Braził	27•41	21.40	23.35	22.10	25.23	27.97	27.06	29.33	28.45	29.02	30.16	28•25	23.24	24.73	<b>N</b> +ā+	1661.9
6.	Mexico	19.14	23.69	25.99	26.97	28.52	28.53	25.65	25.75	27•30	28•92	32.97	34.23	34.11	37.57	38•93 <sup>0</sup>	2231.6
7.	Argentina	n.a.	n.a.	28.74	27.42	24.76	27.84	32.99	25.28	22.68	29.41	33.02	35.83	33.79	39.06	27.58	2554.6
8.	Chlle	n+ā+	16.32	17.89	26.77	35.29	40.05	22.10	18.43	14.02	14.66	18.49	19.66	22.33	25.39	34.94	1331.2
9.	South Korea	10.53	12.05	33.45	32.93	36.03	37.68	33•21	31.83	31.14	33.81	34•04	33.98	36.52	36•96	41.24	1652.8
10.	Japan	66•66	79.18	93.05	104.65	114.13	111-25	105.61	111.54	114.26	116.19	141.71	145.29	147.62	154.60	160.37	9119.2
11.	United States	61.66	66.43	63.25	66.14	67.91	64.93	63.35	66.03	67•72	67.06	64.17	61.94	61.95	60.76	63.77	13303.0

#### Table 1. Ratio of Money Stock (M2) to Gross National Product (GNP): Honduras and Other Selected Countries, 1960-1982

- Source: International Monetary Fund [42]. M2 computed as money <u>plus</u> quasi-money (row 351 in int<sup>1</sup>). Fin. Statistics), <u>plus</u> deposits in other financial institutions (rows 44 and 45 of the same publication), except in the U.S. (domestic M2 definition is used).
- a/ National Accounts figures not available for 1979-82
- b/ Monetary figures not available 1981-82
- c/ GDP used instead of GNP (not available)
- d/ Or latest year available. Exchange-rate conversion may distort cross-country comparisons.

7.Ľ

.

and Japan showed an impressive 154 percent average for these years. In this same period (1980-1982), Honduras shows a relatively high degree of monetization among the low-income countries included in table 1. Despite being the poorest country among this set, money stock as a proportion of GNP in Honduras was higher than in Guatemala and even higher on average than in Brazil and Chile with a per-capita income more than double the Honduran level. In other words, the degree of monetization of the Honduran economy and the relative size of this country's financial sector appear relatively higher than the country's overall development level would suggest.

Another interesting comparison between the countries included in table 1 refers to the evolution of the M2/GNP ratio over time. Overall, South Korea is the most striking case of financial development with a 4-fold increase in the ratio of M2-to-GNP between 1960 and 1982. A second group of countries showing a degree of monetization (ratio M2/GNP) in the early 80's more than double the level observed in 1960 includes Japan, Mexico, Chile, Costa Rica, and Honduras. Finally, the other countries included in table 1 have experienced little or only moderate growth in their levels of monetization during the 23-year period under analysis. In summary, both the recent level of the ratio of money stock to national income, as well as the behavior of this ratio over time, do not place Honduras among the most

"financially repressed" economies of the countries included in table 1. In order to better assess the extent to which financial regulations may have affected financial growth in the Honduran economy, it is necessary to analyze changes in the ratio of money stock to GNP over time. Thus the remainder of this section will discuss the particular features of the evolution of the degree of monetization in Honduras. $\frac{8}{7}$ 

The figures reported in table 1 for the ratio of M2-to-GNP for Honduras show three sub-periods in the development of the country's financial system. First, a period of sustained growth from 1960 to 1973, during which the M2/GNP ratio grew from less than 15% to almost 27%. Second, a period of fluctuation with a moderate-growth trend between 1974 and 1978, the year in which the ratio M2/GNP reaches its highest level in the time series (about 32%). The third sub-period, after 1978, shows a declining trend in the degree of monetization of the economy until 1981, and then a partial recovery in 1982. Despite this recovery, the level of the M2/GNP ratio in 1982 is similar to that existing in 1977, lower than the 1978-level, and far lower than the

<sup>8/</sup> It is clear from table 1 that there are important differences in the growth paths followed by the M2/GNP ratios of different countries. However, a cross-country analysis of these different paths is beyond the scope of this chapter.

level that would have been reached in 1982 had the growing trend prevailing up to 1978 been maintained. $\frac{9}{}$ 

Quarterly growth-rates estimated in a separate study [27] support the foregoing analysis by sub-periods. The growth-rate of the M2/GNP ratio for the period between 1970 and the first quarter of 1974 was almost 1% per quarter. Between the second guarter of 1974 and the end of 1978 this growth-rate was approximately 0.4% per quarter. Finally, for the period 1979-1981 the rate of growth of the ratio of M2-to-GNP was negative. During this period, the M2/GNP ratio was falling at an estimated rate of 1% per guarter. According to this same study, the trends for the ratio M2/GNP can be attributed mainly to the pattern followed by the interest-bearing components of the money stock, i.e., savings and time deposits. Currency and demand deposits as a whole remained at a fairly constant level, as a percent of GNP, between 1970 and 1982. Savings and time deposits on the other hand, which accounted for 50 to 56% of M2 in the 1970-1982 period, marked the growth-stagnation-decline pattern shown by the overall M2/GNP ratio during this period. This performance of interest-bearing deposits suggests a role for interest rates in the explanation of

<sup>9/</sup> The ratio M2/GNP would have reached a level close to 34% in 1982, if the trend observed in the sub-period 1974-78 had been maintained after 1978. Had the average trend of the 1970-78 period been sustained, the 1982-level of the M2/GNP ratio would have reached approximately 37%. These projections are based on quarterly growth rates reported in Cuevas and Vogel [27].

financial growth. A discussion of this role is included in the following sections.

Many factors explain the pattern of financial development experienced in Honduras. Of particular interest is the explanation of the financial decline since 1978. It is not the purpose of this background chapter to engage in a discussion of all phenomena involved in the process of financial development. $\frac{10}{}$  However, some of the main factors affecting this process are outlined below, emphasizing their relevance to the Honduran case.

2.3. Indirect Policy Impacts on Financial Deepening

Policies affecting the development of the financial system can be classified into two categories. The first set are those policies not specifically aimed at the financial sector, but that exercise a significant (indirect) effect on the performance of the system. Typical examples of these policy-effects are inflation that arise from fiscal deficits, and currency-overvaluation resulting from exchangerate policies. The second set are those policies that are specifically designed to affect the operation of the financial sector, such as interest-rate regulations, reserve requirements, rediscount policies, and selective credit allocation policies. A brief discussion of the indirect policies follows. The second type of sector-specific policies is considered in more detail in the next section.

<sup>10/</sup> A discussion of the factors affecting financial development in Honduras is found in Gonzalez-Vega [34].

Inflation affects the growth of the financial sector in several ways. It reduces the real size of the monetary system, will induce the public to substitute non-financial assets for monetary assets, will alter relative prices (specially over time), thus increasing the uncertainty of financial-market operations and shortening the term structure of loans and deposits. Inflation in Honduras arises mainly from a disproportionate expansion in domestic credit aimed at financing fiscal deficits, as documented in Gonzalez-Vega [34]. Important increases in the fiscal deficit during the 1970s were matched by a growing share of the public sector in total domestic credit during this period. As a consequence, the rate of inflation increased particularly in the second half of the 1970s and in the early 1980s.

At this point, it is important to consider the measurement of the rate of inflation in Honduras. This is normally reported in Honduras as the average rate of change of the consumer price index (CPI). A wholesale price index was not available in Honduras until 1982. Therefore the only other deflator available for comparative purposes is the implicit GDP deflator (ID) computed from national-accounts data. Even though the correlation between the two deflators is very high, the rates of change of the two indices show significant differences in some years.<u>11</u>/

II/ Cuevas and Vogel [27] estimated a correlation coefficient of 0.98 between the two indices for the period 1970-81.

Despite these differences, the two time series of inflation-rates show a similarly unstable and fluctuating behavior throughout the period 1970-1982. Likewise, in both cases an upward trend can be identified when the rates of inflation in each series are averaged over the same subperiods reflecting the behavior of the M2/GNP ratio. First, in the sub-period 1970-1973 the average annual rate of inflation was 3.5% according to the CPI, or 3.2% if the changes in the implicit deflator are used to estimate the rate of inflation. The average calculated for the 1974-1978 sub-period jumps to 8% per year in the CPI-based series, whereas in the ID-based series this annual average is 9.9%. Finally, the upward trend in the inflation-rate continues during the period 1979-1982 according to the calculations based on the CPI, that give an average rate of 12.3% per year for this period. The average inflation-rate computed for this same period using the implicit GDP deflator remains at a moderate level (8.7%).

In summary, the first period of rapid monetization of the Honduran economy (1970-1973) is associated with a lowinflation environment. The rather stagnant second period in terms of the growth of the M2/GNP ratio (1974-1978) coincides with an average inflation-rate significantly higher than the one existing in the first period. This relatively high-inflation scenario remains the same or worsens,

depending on the deflator selected, during the period of financial decline after 1978.

The foregoing discussion suggests the existence of a relationship between the formation of inflationary pressures in the economy on the one hand, and the decline of financial-sector growth on the other hand. However, the inflationary process in Honduras is moderate as compared to other Latin American countries and inflationary expectations may not have developed in Honduras until after 1980. The tradition of a stable price level in Honduras matches the even more traditional stability of the exchange-rate. This rate has remained at the same level, 2 lempiras per dollar, for more than forty years. Furthermore, only in very recent years has a serious disequilibria appeared in the balance of payments inducing devaluation expectations, thus capital flight and currency substitution. $\frac{12}{}$  These conditions may have contributed to the financial decline observed after 1978.

An overall assessment of the (indirect or unintended) effects of fiscal and foreign-exchange policies on the development of the financial sector of Honduras indicate that: (a) low-inflation and exchange-rate stability are associated with sustained growth of the financial system, and (b) the appearance of moderate inflation and currency-overvaluation seem to induce a decrease in the rate of growth and 12/ See Gonzalez-Vega [34].

eventually a contraction of the financial system. The incidence of sector-specific policies and their interactions with inflationary phenomena are the subject of the following section.

2.4. Financial-Sector Regulations and Financial Deepening This section reviews the main financial regulations prevailing in Honduras during the period 1970-1982. Interest-rate regulations, rediscount policies, and reserve requirements are considered. Emphasis is placed on the likely effects of these policies on deposit-mobilization, and on the operation of financial intermediaries.

Interest-rate ceilings on deposits and loans have existed during most of the period 1970-1982. Interest-rates paid on time-deposits were freed in 1978, and the ceilings for rates paid on savings deposits were lifted in 1981. However, loan-rate ceilings have remained in effect throughout the entire period, thus imposing indirect ceilings on deposit-rates even after these were officially eliminated. Tables 30 and 31 in appendix A describe the evolution of interest-rate ceilings on deposits, table 30, and loans, table 31, both in nominal and real terms. As expected, fairly stable nominal administered rates with a fluctuating rate of inflation generate unstable and fluctuating real interest-rates, regardless of the deflator selected (the CPI, or the implicit GDP deflator, ID). In order to capture the existence of possible trends in real

rates, it is useful to smooth these fluctuating rates by averaging them for different sub-periods. The discussion that follows relies upon these average real interest-rate ceilings.

In the first four years of the series, 1970-1973, deposit rate ceilings show, on average, low though positive levels, of less than 1% for savings deposits, and about 3.5% for time deposits. Despite an increase in the nominal ceiling for interest rates on savings deposits after 1973, the average real rate for these accounts in the period 1974-1978 is -2% using the inflation rate of the CPI, or -3.6% if the ID-based rate of inflation is utilized. The real rate on time deposits falls in this same period to a near-zero average according to the CPI based rate of inflation, or -1.3% according to the inflation-rate of the implicit deflator. The two years with explicit ceilings for interest rates on savings deposits after 1978 show high negative real rates with an average of -6.6% or -5.2% depending on the deflator chosen for the calculations. After the elimination of all ceilings in 1981, the nominal effective interest rates paid by banks, according to central-bank estimates for late 1981, imply negative real returns for depositors in savings accounts (-2.1%) and very low positive rates for holders of time deposits (2.8%). This reflects the indirect effects of the ceilings on lending rates on deposit-rates.

Ceilings on loan-rates cause financial-market fragmentation. Seven or more different levels of the lending-rate ceilings were in effect at the same time during the period under study. Among these, the maximum lending-rate (ranging from 16% to 19%) was usually allowed only for selected operations using own-resources mobilized from the public. The average lending-rate ceiling was normally well below this maximum rate, thus providing a permanent incentive to lenders to reclassify loans and reallocate funds internally in order to maximize the amounts lent at the maximum rate.

In real terms, lending-rate ceilings have been predominantly positive between 1970 and 1982. However, a declining trend is observed when comparing sub-period averages. The average maximum lending-rate ceiling in real terms was higher than 14% between 1970 and 1973. This average declined to 7.8% or 6%, depending on the deflator, for the period 1974-1978. After this year, the CPI deflator indicates a further decline to a 5.5%-average in the period 1979-1982, whereas the ID-based inflation rate gives an average of 9.1% in real terms for this same period. In general, lending-rate ceilings have implied positive though declining real rates, a situation that contrasts with the strongly negative real rates of interest characteristic of credit markets in other less-developed economies [2].

The different lending-rate ceilings pointed out above are closely associated with the rediscount policies of the

central bank. These policies are represented in table 32 (appendix A), where the number of rediscount lines in different years is reported, along with the discount rates and associated lending rates of the main lines existing during the period. The number of discount rates associated with central-bank lines of credit grew from a total of 3 in 1973, to 9 different discount rates in 1982, varying according to the specific targets of the rediscount lines. This reflects increased fragmentation in the financial sector. Furthermore, the low discount rate is offset by upward pressures on the operational costs of financial institutions using rediscount funds. Additional accounting, monitoring, and reporting schemes have to be implemented by these institutions to comply with the requirements attached to the use of rediscount funds.

The increase in the number of rediscount lines indicated above is in a sense "consistent" with the interestrate policies applied to deposits. As pointed out before, the real levels of the interest-rate ceilings on deposits provided a growing disincentive for depositors to hold financial savings. For the depository institutions this means increasing costs of mobilization, since they need to offer non-interest rewards to their depositors in order to at least maintain the existing levels of deposit balances. Under these circumstances, central-bank rediscount lines appear as a convenient source of low-cost funds, especially

appealing to banks with difficulties in mobilizing deposits from the general public. There are however, trade-offs involved in the use of rediscount funds, as this study will show later. Use of these funds avoids the traditional costs associated with deposit-mobilization, but do represent a source of additional costs for banks due to the creditallocation constraints (i.e., loan targeting) associated with rediscount lines, with the reporting and documentation requirements mentioned earlier.

The main lines of rediscount funds established lending-rate ceilings that imply a gross margin for loan administration between 4 and 7 percentage points. In order to estimate the gross spread or margin implicit in the structure of interest-rate ceilings, it is necessary to incorporate into the analysis the reserve requirements prevailing for Honduran financial institutions. Reserve requirements fluctuated between 25% and 30% for demand, savings, and time deposits in banking institutions during the period between 1970 and 1982. These requirements were lower for deposits in savings and loans associations, and higher for deposits denominated in foreign currency, thus adding to the fragmentation in financial markets.

Table 2 presents the estimated gross-spread or margin implicit in the interest-rate and reserve policies of the central bank during the period 1970-1982. First, the weighted average of deposit-rate ceilings adjusted by

reserve requirements is included in column 1 of table  $2.\frac{13}{2}$ The average lending-rate ceiling, as well as the maximum lending-rate ceiling, are seen in columns 2 and 3. The implicit gross margin allowed for financial intermediaries by these financial regulations is estimated for the average lending-rate ceiling and then for the maximum level among these ceilings in the last two columns of table 2. It is clear from this table that the gross spread implicit in the central-bank's financial policy has systematically shrunk. If banks had lent all funds at the maximum lending rate, and paid the maximum allowed rate on deposits, their margins would have shown a decrease from a "generous" average spread of ll.1% in the earlier years of the period (1970-1973), to a narrow 5%-spread in the post-1978 period.

A more reasonable assumption is that banks have lent on average at rates between the maximum and the average ceilings. On the other hand, it is likely that these banks have paid interest-rates on deposits close to their ceiling levels, since these have been already sufficiently low in real terms to discourage depositors. Under these conditions, the implicit spread set by policy-makers for financial institutions lie between the margin computed using the average ceiling, column 4 in table 2, and the spread associated with the maximum lending-rate ceiling. This range

<sup>13/</sup> The (low) interest-rate paid on the proportion of total reserves held in government bonds was not considered in the calculations. However, this represents a minor bias in the estimated average effective deposit-rate ceiling.

	(1)	(2)	(3)	Implicit Gross-spread			
	Average Deposit-rate	Average	Maximum	With Average	With Maximum		
	Ceiling Adjusted by ,	Lending-rate	Lending-rate	Lending-rate	Lending-rate		
	Reserve Requirements <sup>a</sup> /	Ceiling	Ceiling	(2) - (1)	(3) - (1)		
Year			<u> </u>	£	8		
1970	6.85	_	18	_	11.15		
1971	6.85		18	-	11.15		
1972	6.85		18	-	11.15		
1973	. 7.22	12.0	18	4.78	10.78		
1974	8.50	12.8	18	4.30	9.50		
1975	9.40	12.1	16	2.70	6.60		
1976	9.40	12.1	16	2.70	6.60		
1977	9.40	12.1	16	2.70	6.60		
1978	10.19	12.1	16	1.91	5.81		
1979	11.93 <u>b</u> /	12.0	16	0.07	4.07		
1980	13.20b/	13.3	19	0.10	5.80		
1981	14.00 <u>c</u> /	14.3	19	0.30	5.00		
1982	n.a.	15.0	19		-		

Table 2. Deposit-rate Ceilings Adjusted by Reserve Requirements, Lending-rate Ceilings, and Gross-spread Implicit in Interest-rate Regulations and Reserve Requirements, 1970-1982

Sources: Tables 31 and 32 in appendix A, and Central Bank (Superintendence) internal memoranda

- a/ Weighted average of interest-rate ceilings on savings deposits and time deposits, adjusted by the reserve rate. Reserve rates: 25% in 1970-72, and 1975-77, 28% in 1973-74, 30% in 1978-82.
- b/ Interest-rate ceilings on time deposits were lifted in Dec. 78. Averages for 1979 and 1980 are based on Central Bank (Superintendence) estimates of effective interest-rates paid on time deposits
- c/ All interest-rate ceilings on deposits eliminated after May 1981. Average for 1981 based on Central Bank (Superintendence) estimates of effective interest-rates paid.

n.a.: not available.

results in an average implicit spread between 2.9% and 7% for the period 1974-1978. For the subsequent period, 1979-1982, the average margin implicit in the financial policy would fall between 0.16% and 5%. These margins implicit in the interest-rate and reserves policies are in general consistent with those indicated above for centralbank rediscount lines.

The foregoing discussion raises two important points. First, sector-specific regulatory policies affecting the financial system have resulted in low (usually negative) and unstable real deposit-rates, thus penalizing and discouraging savers. The downward trend observed in these real deposit-rate ceilings can be associated with the slowdown and decline in the growth of the Honduran financial sector discussed earlier. Second, these combined policies have caused decreasing spreads for financial intermediaries operating with own-resources mobilized from the general public. Regardless of the source of loan funds, these gross spreads implicit in interest-rate, reserve, and rediscount policies range from roughly zero up to 7% in the period after 1974. Therefore, if intermediation costs are in effect larger than the gross margins implicit in the regulatory setting then there will be strong incentives for the financial institutions to avoid these regulations in order to minimize operational losses. Financial institutions will allocate funds in their loan portfolio to maximize the use

of high-interest lines of credit, and will try to pass-on transaction costs to other participants in the system, primarily to the ultimate borrowers. The level, structure, and behavior of these intermediation costs borne by lenders and borrowers are the subject of the following chapters.

# CHAPTER III Theoretical Framework

This chapter reviews the theoretical elements involved in the study of financial intermediation costs. First, the role of the financial system as a service sector that creates specialized commodities or services is discussed. The transaction costs involved in the provision of these financial commodities are emphasized. Second, the theory of the banking firm provides a basis for the analysis of lender's intermediation costs. The ways in which financial regulations affect the use of real resources by financial institutions are discussed in this section. Finally, the theoretical foundations of credit rationing constitute the framework for the study of borrowing transaction costs.

The working assumptions adopted are: (i) lenders are cost-minimizing firms that behave as price-takers in input markets; and (ii) lenders are able to exercise loan-rate differentiation through the non-interest component of the price vector associated with loan operations, passing on transaction costs to borrowers and substituting the discriminatory application of loan procedures among borrowers for explicit loan-rate differentiation.

There is no a priori reason to modify the foregoing assumptions for the case of development banks. As long as financial viability and institutional survival are objectives pursued by these specialized institutions, revenues and costs must be taken into account. Therefore, the only additional assumption that is required with respect to development banks is that the goals of financial viability and institutional survival are components of their objective function. Some literature on specialized lending institutions imply that the latter assumption may not be valid for publicly-owned institutions, given the political role that these entities allegedly perform [2,23,93]. However, it has also been argued that managers of these institutions must be concerned about costs and profitability, in order to improve their standing with higher-ranked government authorities concerned with fiscal deficits [34].

In general, development goals and political objectives predominate over profit considerations in public development banks [20,41]. However, these goals and objectives are attained through the selective provision of banking services (primarily loans) to specific groups, under special terms. Therefore, given these objectives, it is still in the interest of the bank's administration to minimize the costs of providing these services. This allows them to reach a larger set of target groups, and to service a broader scope of political interests than would be the case if cost

considerations were neglected. In summary, even though long-term profit maximization conventionally defined is a questionable assumption for public development banks, cost minimization seems to be an appropriate framework for the analysis of these institutions.

### 3.1. The Functions of Financial Intermediaries and Transaction Costs

A financial system provides four services in an economy: (1) provision of a medium of exchange and a store of value, money; (2) financial intermediation, i.e., mobilizing funds from savers (surplus units) to investors (deficit units); (3) provision of a means of transforming and distributing risk across the economy; and (4) provision of a set of golicy instruments that serve as stabilization tools. The provision of these services require the use of real resources, and therefore involve costs of production [65]. Benston and Smith [14] conceive financial intermediaries as producers of specialized financial commodities. These commodities are sold for prices that are expected to cover the direct costs and the opportunity costs of production.

Demand for financial commodities derive from the existence of transaction costs in performing transfers between individuals and over time. Benston and Smith summarize their discussion of the demand for financial commodities as follows:

"...financial intermediaries meet consumers' demands for time-dated consumption by supplying units of generalized purchasing power that can be converted into goods or services at minimal transaction costs in the amounts and at the times demanded" [14, page 219]

Financial intermediaries allow borrowers the opportunity to acquire the services of real goods <u>now</u> in exchange for reduced consumption in the future. At the same time, financial intermediaries provide depositors with investment opportunities that match their preferences in terms of liquidity and maturity. In short, loans and deposits are among the commodities produced by financial intermediaries. Labor and capital goods are utilized in the production of financial commodities, primarily through resources devoted to documentation, information and monitoring activities. Financial intermediaries can achieve economies of scale and economies of diversification in performing these tasks. Furthermore there are economies enjoyed by consumers from the provision of several financial services by a single institution [14].

In summary, the existence of transaction costs in inter-temporal and intra-temporal transfers explains the existence of financial intermediaries. The provision of financial commodities such as loans and deposits by these intermediaries reduces overall transaction costs for society in performing these transfers. However, the transaction costs remaining in the process of financial intermediation

are still significant. Their level and distribution among the participants in the process are affected by changes in technology, changes in consumer preferences and more importantly, by financial regulations.

Surplus units in the economy (depositors) incur search and information costs before selecting a depository institution. Once this selection is made, there are further costs associated with performing transactions, i.e., deposits, withdrawals, and transfers. The opportunity cost of time is likely to be the most important transaction cost for depositors. Transaction costs incurred by financial intermediaries may be classified into costs of mobilizing deposits and costs of lending. The former correspond to resources utilized in handling deposit accounts, documentation, record-keeping, and issuing statements. Costs of lending refer to costs associated with loan processing, monitoring, and loan recovery. Gathering information about potential borrowers, assessment of collateral and documentation are among these lending costs. Beyond these direct costs of intermediation, the return to the owners of financial intermediaries (i.e., profits) should be considered a part of the overall margin needed by these intermediaries to stay in business. Finally, borrowers incur transaction costs in negotiating, obtaining and repaying a loan.

The foregoing discussion can be summarized in the following set of simplified relationships:

where, d is the interest rate paid on deposits,  $c_{g}$  is the (per unit) non-interest transaction costs incurred by depositors; (b) for borrowers, effective or total costs of borrowing = i + c\_b where, i is the interest rate charged on loans,  $c_{b}$  is the (per unit) non-interest transaction costs of borrowing; (c) for financial intermediaries,  $d + c_{m} + c_{1} + b = i$  (3.1) where,  $c_{m}$  is the (per unit) cost of mobilizing deposits,

for savers, effective return =  $d - c_s$ 

(a)

c1 is the (per unit) operational costs of lending, and

b denotes per unit profits.

It is clear from the identity (3.1) that in the absence of regulations on interest rates any changes in the return to savers (d) or in the direct costs of intermediation ( $c_m$ ,  $c_1$ ), or in the opportunity cost of capital invested in financial intermediation (b) will be reflected in the interest rate paid by the ultimate borrower (i). Different types of financial regulations will affect different components of equation (3.1), or will directly affect the transaction costs borne by savers ( $c_n$ ) and borrowers ( $c_b$ ).

Regulations on financial intermediaries may be classified into four groups [14]: (1) licensing, (2) price

controls, (3) credit allocation, and (4) supervision. A brief discussion of the first three groups follows. Supervision is closely related to credit allocation policies in terms of its effects on transaction costs, therefore it will not be treated separately here.

- (1) Licensing, i.e., entry-restrictions or restrictions to branching, results in consumers of financial services (depositors and borrowers) bearing higher transaction costs than would be the case with free-entry or branching. The problem of scale is important here, particularly in small markets, thus branch expansion could be limited even in the absence of licensing restrictions, at the expense of high transaction costs for consumers in small communities.
- (2) Price controls are probably the most important type of regulation. These take the form of ceilings imposed on rates paid on deposits and/or ceilings on lending rates. The former affect the ability of financial intermediaries to mobilize deposits, the latter constrain their ability to discriminate between borrowers of different risk or creditworthiness. As a result, the costs of mobilizing  $(c_m)$  will be adjusted to compensate depositors beyond the ceiling rate. At the same time, the costs of lending associated with high-risk customers cannot be fully reflected in the level of the explicit interest rate due to the

existence of lending-rate ceilings. Therefore, equation (3.1) tends to become an inequality:

 $\overline{d} + c_m + c_1 + b > \overline{1}$  (3.2) where  $\overline{d}$  and  $\overline{1}$  denote administered ceilings on the deposit and the lending rate. In order to restore the original identity, either profits have to go down and eventually become operational losses, or a proportion of the costs of intermediation ( $c_m$ ,  $c_1$ ) must be transferred to other participants in the market.

Of these two possibilities, borrowers are more likely to have their transaction costs increased as a result of the interest-rate regulations. They will have to provide more information and documentation that otherwise would have been gathered by the intermediary, and follow additional and more complicated procedures. Borrowing transaction costs will increase, and "honest" borrowers will be discouraged by these increased costs. Investors with high yield-high risk alternatives, and "dishonest" borrowers determined to default will still be willing to bear the increased level of transaction costs. Thus a process of adverse selection may result from the imposition of lending-rate ceilings.

(3) Selective credit policies, or mandatory credit allocation, have been a popular form of policy intervention, particularly in low-income countries. Substitutability (fungibility) of credit both in lender portfolios as

well as in borrower funds tend to make these policies ineffective [86]. At the same time, these selective credit policies will impose additional transaction costs on the economy [14].

The foregoing discussion offers a conceptual framework for the following sections. These sections formulate the basic models for the analysis of the transaction costs of intermediation borne by financial intermediaries (section 3.2), and by borrowers (section 3.3). The models take into account the effects of two types of financial regulations: interest-rate controls, and selective credit policies. Specification and estimation procedures applied to these models are discussed in chapter 4.

### 3.2. Lender's Intermediation Costs

Several different approaches to modeling the behavior of financial intermediaries are found in the literature. Baltensperger [8] and Sealey [81] have surveyed the main developments and have attempted to integrate them in more comprehensive models at the cost of introducing further complexities.

Two main approaches discussed by Sealey [81] are the Markowitz-Tobin portfolio approach (e.g. Kane and Malkiel [55]) and the firm-theoretic models (e.g. Klein [58]). The latter present the advantages of allowing the explicit consideration of market conditions, resource costs involved in intermediary operations and deposit-rate setting behavior.

The portfolio-theory approach omits these aspects of banking behavior, however, it has the advantage over the firmtheoretic models of incorporating uncertainty and particularly non-linear risk preferences in intermediary behavior.

Since the main concern here is with the behavior of the resource costs involved in financial intermediation, the theory-of-the-firm approach will be followed. It will be assumed that the financial intermediary minimizes cost subject to the constraint of a function that relates the production of banking services to the use of productive factors and inputs. The model can be summarized as follows: minimize  $C = \sum_{j=1}^{n} p_j X_j$ , cost equation, (3.3)

subject to

 $F(q_1, \dots, q_m, X_1, \dots, X_n) = 0$ , production function, (3.4) where, C is operational costs of production

- q<sub>i</sub> is the quantity produced of the ith output, i=1,...,m
- $x_j$  is the quantity of the jth input, j=1,...,n p<sub>j</sub> is the price of the jth input, j=1,...,n.

The solution of the system formed by equations (3.3), (3.4) and the first-order conditions for cost minimization yields a cost function that depends on the output levels and factor prices, p;.

$$c = \Phi (q_1, \dots, q_m, p_1, \dots, p_n) .$$
 (3.5)

This same approach underlies other empirical studies on banking costs and economies of scale in banking [15,16,17, 18,30,69].

In general, the production technology represented by the implicit form (3.4) is not known. However, the duality relationships between cost and production functions allow inferences about the production technology from the knowledge of the cost function [61,91]. Varian [91] has shown that an input-requirement set that approaches the true (unknown) technology can be defined given a cost function with input prices and output levels as arguments. This cost function will be the same as the cost function associated with the original technology, since it will include only economically efficient points of this production technology. Furthermore, if the original production technology is "well-behaved", i.e. continuous, monotonic, convex and closed in output and input levels, then the production technology derived from the cost function is the true technology. $\frac{1}{}$ 

The foregoing duality relationships allow the study of different properties of the underlying technology, starting with the cost function. Among these properties, economies of scale, economies of scope, elasticities of factor substitution and elasticities of factor demand are of particular importance. Their definitions and formulations will be considered in chapter 4.

<sup>1/</sup> See Varian [91]. Lamberte [62] reviews many dual relationships of cost and production functions, and applies them to the analysis of banking costs.

Regulation-related variables are introduced in the cost function (3.5) by assuming that the total demand for every factor of production X<sub>i</sub>, can be decomposed into two parts: (a)  $x_{i1}$ , which corresponds to the level of  $x_i$  consistent with an unregulated environment; and (b),  $x_{i2}$ , an additional quantity or a differential skill that is required by existing regulations. Examples of these are additional personnel or special mechanisms devised to provide customer services that compensate for deposit-rate ceilings, and teams hired and trained to deal with specific project funds and clientele. Also, additional accounting and recordkeeping personnel become necessary to comply with the reporting requirements of special credit programs. Finally, in the case of public lending institutions there are usually expanded personnel costs of featherbedded employment within the institution and additional workload (i.e. costs) of the existing staff associated with servicing political-patronage clients.

The level of input j can then be written as

$$x_{j} = h_{j}(x_{j1}, x_{j2})$$
, (3.6)

where the magnitude of  $X_{j2}$ , the regulation-induced portion of  $X_j$ , is assumed to depend on the degree of regulation. At this stage, this relationship for the case of interest rate regulations can be formulated as follows:

$$x_{j2} = g_j(d_e - d) , g_j > 0 , d_e > d .$$
 (3.7)

where  $(d_e - d)$  is the difference between an equilibrium doposit rate  $d_e$  and the statutory rate d. It is expected that the closer d is to  $d_e$ , the lower the level of  $X_{j2}$ , and therefore the lower the level of non-financial costs. In other words, assuming that  $d_e$  is approximately constant the existence of a trade-off between the explicit interest rate paid on mobilized funds and the operating costs of the financial intermediary is hypothesized. The fact that the equilibrium rate  $d_e$  is usually unobservable introduces a difficulty in the measurement of the difference  $(d_e - d)$ . In this respect, the assumption that  $d_e$  is constant makes the statutory rate d itself a good proxy for the differential since  $\Delta(d_e - d) = -\Delta d$ . This assumption is kept for the time being, postponing the discussion about the measurement of  $(d_e - d)$  until the next chapter.

Interest-rate regulations are not the only variables to be included in the cost function. Many other indicators of, or proxies for, financial-market regulations can be utilized in the empirical assessment of the cost function (3.5). This study emphasizes the effects that interest-rate ceilings, and loan targeting have on the costs borne by financial intermediaries. Chapter 4 discussed the specifications of these effects in the cost function.

3.3. Credit Rationing, Implicit Pricing and Borrowing Costs The literature on credit rationing has been abundant over the last two decades [6,45,46,47,48,49,87,88]. Fairly

comprehensive reviews can be found in Baltensperger [7] and Gonzalez-Vega [31]. A credit-rationing framework is utilized in this section to introduce the relationship between the implicit interest charged in loan operations and the characteristics of these operations associated with borrower's riskiness and demand conditions.

Borrowers are seen by lenders as essentially nonhomogeneous. Each borrower has a different demand function for loanable funds and, more importantly, different borrowers have different risk characteristics and therefore different probabilities of repaying their loans. Since lenders are concerned with the expected return on loans which is a decreasing function of risk, they will be interested in using various "screening devices" for their borrowers [88], of which the interest rate would be the most important.

Thus loan-rate differentiation is a necessary element of lenders' behavior if they are to maximize profits. Some literature on credit rationing has approached this issue by considering lenders as price-setting entities that optimize along the borrower's demand function [47,49], even though this price-discriminating behavior is not necessarily determined only by different demand elasticities. Different risk characteristics of customers, and, in this sense, different costs associated with the loan operation also play a role [7]. The price-setting analytical model set forth below

closely follows those presented by Jaffee and Modigliani [47] and Jaffee and Russell [49], with some extensions relevant for the purposes of this study. The main implication of these modifications is that credit rationing will be exercised primarily through implicit-price adjustment, rather than through quantity restrictions.

It is assumed that lenders maximize the expected value of their profits,  $\pi$ , which in a loan operation are given by:

 $\pi = LR[P] - LC ,$ 

where,

LR is the size of the loan contract given by:

R, the interest-rate factor R = 1 + r, and L is the loan size, a point on the borrower's loan demand function faced by the lender. This function,  $L = L(R, \overline{W})$ , derives from a multi-period optimization in which the borrower behaves as price-taker and where  $\overline{W}$  represents the individual's resource endowment that influences the potential size of his/her investment projects. It is assumed that L' =  $\partial L/\partial R < 0$ 

P = P (L,R) is the likelihood of repayment (the  $\lambda$  function in Jaffee and Russell, [49]) which is conditional on the value of a minimum cost of default Z accruing to borrowers, that determines the range of contract sizes over which default is observed.

P(L,R) = 1 if  $LR \leq Z$ ,

48

(3.8)
$P(L,R) < 1, P'_L < 0, P'_R < 0, \text{ for } LR > 2^{2/}.$ 

where the prime denotes the partial derivative with respect to the variable that appears as a subscript.

C is the cost of funds for the lender which is assumed constant for simplicity (i.e. the marginal cost is equal to average cost).

Maximization of (3.8) with respect to the loan rate factor R gives the following equilibrium condition, stated in terms of the elasticities of the loan demand function ( $\eta$ ) and of the likelihood function ( $\epsilon$ ) with respect to the loan rate factor:

RP 
$$[1 + \frac{1}{\eta} (1 + \varepsilon)] = C$$
. (3.9)

In other words, the expected marginal revenue is set equal to the marginal cost, therefore the optimal loan rate is determined by:

$$R^* = C/P[1 + \frac{1}{\eta} (1 + \varepsilon)] . \qquad (3.10)$$

That is, the loan rate would be optimally set considering the probability of repayment (P), the borrower's demand elasticity ( $\eta$ ) and the response of P to changes in R ( $\varepsilon$ ). In general form, R (and thus r) will be a function of loan

2/Note that  $P'_R = \partial P/\partial R < 0$  does not ensure dP/dR < 0 since the latter is given by  $dP/dR = P'_L + P'_R$  where the first term to the right of the equal sign is positive (since L' < 0) thus making the sign of dP/dR indetermined.

demand and the probability of repayment, together with the perceived response of those functions to variations in R.

Note that under certainty of repayment (P=1,  $\epsilon$ =0) condition (3.9) reduces to the familiar result in monopolistic equilibrium:

$$R (1 + \frac{1}{\eta}) = C$$
 (3.11)

The two reasons for loan-rate differentiation are summarized in equation (3.10): first, as the likelihood of repayment, P, diminishes, i.e. <u>ceteris paribus</u> the loan becomes riskier, the interest-rate factor R (and therefore r) will go up. Second, customers with different demand elasticities will be charged (everything else constant) different rates. An additional element in (3.10) is the response of the probability of repayment to changes in R. However, its behavior will not be discussed at this stage, considering the simplifying assumptions made with respect to the P function.

Pure monopoly price setting is not a necessary condition for this loan-rate differentiation process. As asserted in Stiglitz and Weiss [88], many banks can compete by selecting prices (interest rates) that maximize their profits. However, the typical environment in which lenders perform their activities in LDCs is characterized by institutional arrangements that constrain price-setting or loan-rate differentiation. These restrictions are particulary strong in rural financial markets in LDCs where the targeting of credit to specified groups or end-use requirements for loan funds at concessionary (and controlled) interest rates prevail [2].

Under regulated conditions, facing constraints on loan rate differentiation, lenders will engage in "regulatory avoidance" or implicit-price setting (Kane, [54]). This involves establishing different procedures for credit allocation, monitoring and supervision that create both lender and borrower transactions costs (see [2,59]). This amounts to exercising price-setting through the non-interest component of the price vector. Lenders are substituting the discriminatory application of loan procedures among borrowers for explicit loan rate differentiation. Also, to the extent that different sources of funds (international donors, and government) allow lenders to charge slightly different loan rates, lenders will use their limited discretionary power on these rates to set their prices. This price-setting procedure forces different borrowers onto different "tracks," where the number and height of the obstacles to negotiate loans in each track (i.e., transaction costs) are controlled by the lending institutions, enabling them to ration-out unwanted (risky) clients and ration-in desired clients.

Furthermore, the lender can transfer the burden of transaction costs from himself to the borrower in the form of administrative charges, fees, documentation requirements

and charges, and compensatory balances. Borrowers will experience a rise in their total borrowing costs equivalent to the implicit charges passed on by the lender. There is however, as Kane [52,54] points out, some degree of waste embodied in implicit pricing since this effort diverts economic resources from other uses. When this occurs the total borrower's costs will differ from the actual revenue or total price perceived by the lender by the amount of that "waste." This wedge is neglected in the following analysis assuming that borrower's costs reflect accurately the differential loan rates (inclusive of implicit charges) that lenders impose.

In terms of the model developed above, the interestrate factor R should now be interpreted in the broad sense of including explicit and implicit interest, i.e., the rate r will consist of an explicit rate (i) and an implicit element ( $\tau$ ) which result from expressing borrowing transaction costs per loan on a percent basis. It is precisely this component of the total price that will be affected by the variables involved in equation (8), i.e. borrower's riskiness and demand conditions, since the explicit rate is bounded by the existing regulations.

In summary, it is argued here that lenders in rural financial markets in LDCs are price-setters (of explicit and implicit interest charges) that take as given the profile of loan demand such as farm size, loan amount, enterprise type,

and other characteristics of the borrower. Lenders then set explicit interest charges and, more importantly, establish differential administrative procedures that are in effect transformed into implicit charges (i.e., transaction costs for the borrower) according to these loan demand characteristics.

A general formulation that derives from the foregoing discussion is as follows:

$$T = T(B, i)$$
, (3.12)

where,

- T is the borrowing (non-interest) transaction costs per loan
- B is a vector of risk-related characteristics of the loan operation (loan size, firm size, and loan use).
- i is the explicit interest rate that can be charged on loans by the lender.

The main concern here is to investigate the trade-off between explicit interest and transaction costs or implicit interest, passed-on or charged by lenders to borrowers. Also, it is important to examine the incidence of different characteristics of the loan operation on the level of transaction costs, as well as the interactions between these characteristics and the trade-off between explicit and implicit interest.

# CHAPTER IV

## Methodology

Econometric analyses of time-series and cross-sectional data are the methodological tools used in this study. This chapter reviews the specification and variable definition associated with the empirical estimation of the relationships formulated in the preceding section. The lender's cost function is reviewed first (section 4.1), followed by a discussion of the approach to be adopted for the borrower's cost analysis (section 4.2).

#### 4.1. Lender's Cost Function

Lender's costs are composed of the opportunity cost of loanable funds, operational costs, and risk costs, usually measured as the losses due to default [64]. Since this study is concerned with resource costs involved in financial intermediation, the lender's administration or operational costs comprise the dependent variable in the cost function (3.5):

 $C = \Phi(q_1, ..., q_m, p_1, ..., p_n, \Gamma) , \qquad (4.1)$ 

where  $\Gamma$  stands for a vector of regulatory indicators.

The (per unit) explicit costs of funds will enter the function as one of the proxies for the degree of regulation,

i.e., one element of vector  $\Gamma$ . On the other hand, administrative or operational costs include the resource costs involved in routine loan-recovery activities, but they do not include the default rate or a related measure as an ex ante premium for risk. Accounting provisions for unrepaid loans should reflect this ex ante risk-premium, but that is not necessarily the case in regular accounting practices. The approach utilized here will not include these accounting provisions in the measurement of the dependent variable in the cost function. Instead, the delinquency rate will be considered as a variable that influences the allocation of real resources in the financial institution, and therefore affects loan administration costs (Lee and Baker [64]). In other words, the treatment of delinquency will focus on the effects that variations in the delinquency rate have on costs of financial intermediation.

Four issues will be discussed below in relation to the lender's cost function: (1) specification, (2) output definition, (3) proxies for regulation and (4) data and estimation.

#### 4.1.1. Specification

The specification of a mathematical form for the cost function (4.1) takes into account certain conditions suggested by theoretical considerations [38,55,58], earlier empirical work [15,17,68,69], and the specific objectives of this study. These conditions refer to the shape of the

average cost curves, and in general to the characteristics of the underlying technology, particularly economies of scale and factor-substitution. Also, allowing for interactions between the variables entering the function will help in testing different hypotheses with respect to their effect on intermediation costs.

Earlier studies in less-development countries (e.g. Gheen [16], Nyanin [43]) have provided very limited insights into the characteristics of the cost structure and underlying technology of these institutions, due to the choice of very restrictive functional forms for the cost function. In general, the use of Cobb-Douglas or CES specifications implies the adoption of highly restrictive assumptions about the technology utilized by financial intermediaries. Under these specifications, scale economies are forced to remain constant, regardless of the output level, therefore the corresponding average cost curves are either downward or upward sloping throughout the entire output domain. In other words, under these constrained functional forms, the existence of U-shaped average cost curves is ruled out a priori.

In this study, the cost-function specification will be flexible enough to allow some specific features of the underlying technology such as economies of scale and factor substitution to be tested. Furthermore, it is important to allow for interactions such as the effect of loan size on

the marginal cost of lending, or the effect of deposit size on the marginal cost of mobilizing deposits.

The foregoing arguments call for the specification of a generalized form of the cost function. Within this class of mathematical functions, the trascendental logarithmic (translog) form has been utilized in a number of earlier banking cost and profit-function empirical studies [15,17,68,69,72], as well as in cost analyses of multipleoutput, multiple-input firms [18,19,21,25,78]. The translog cost function is essentially a second-order approximation to an arbitrary cost function. It is quadratic in the logarithms of quantities and input prices, and linear in the parameters. This function has flexible properties with respect to the characteristics of the underlying technology discussed above, and allows multiple interactions between the variables in the model. Many assumptions imposed by other functional forms, such as homogeneity or unitary elasticity of factor-substitution, become testable hypotheses under the translog specification. The use of this functional form is particularly pertinent in multi-output production, as is the case of financial institutions producing at least two different outputs, loans and deposit services, in varying proportions. $\underline{1}^{/}$ 

In what follows, the specification of the implicit cost function (4.1) is developed for the two-output, two-input

<sup>1/</sup> For a detailed characterization of the translog function see Binswanger [21], Christensen, Jorgenson, and Lau [25], and Ray [78].

case. The vector of regulatory indicators  $\Gamma$  is omitted for simplicity. The specific forms in which these indicators will enter the cost function are discussed in section 4.1.2 below.

For two outputs and two inputs, the translog function is written as follows:

$$\ln C = \alpha_{0} + \alpha_{1} \ln q_{1} + \alpha_{2} \ln q_{2} + \beta_{1} \ln p_{1} + \beta_{2} \ln p_{2} + + \frac{1}{2} \gamma_{11} (\ln q_{1})^{2} + \frac{1}{2} \gamma_{22} (\ln q_{2})^{2} + \gamma_{12} \ln q_{1} \ln q_{2} + + \frac{1}{2} \delta_{11} (\ln p_{1})^{2} + \frac{1}{2} \delta_{22} (\ln p_{2})^{2} + \delta_{12} \ln p_{1} \ln p_{2} + + \eta_{11} \ln q_{1} \ln p_{1} + \eta_{12} \ln q_{1} \ln p_{2} + \eta_{21} \ln q_{2} \ln p_{1} + + \eta_{22} \ln q_{2} \ln p_{2} ,$$

$$(4.2)$$

where,  $q_i = quantity$  of the ith output,

The cost-share equations for the two factor inputs derive from equation (2) as:

$$S_{j} = \beta_{j} + \Sigma_{h} \delta_{jh} \ln p_{h} + \Sigma_{i} \eta_{ij} \ln q_{i}, \quad j,h = 1,2, \quad (4.3)$$
  
 $i = 1,2,$ 

where S denotes the cost share of factor j,

$$s_{j} = \frac{p_{j}x_{j}}{C} = \frac{\partial lnC}{\partial lnp_{j}}.$$

Cost function (4.2) should be homogenous of degree one in input prices. This condition imposes a set of restrictions on the parameters of equation (4.2) that is also consistent with the requirement that the sum of the cost shares (4.3) must equal one:

 $\Sigma_{j}\beta_{j} = 1$ ,  $\Sigma_{j}\delta_{jh} = 0$ ,  $\Sigma\eta_{ij} = 0$ , j,h = 1,2, i = 1,2. Several properties of the cost structure and the

underlying production function can be investigated using the translog cost function defined in equation (4.3). These properties are summarized below.

#### Economies of Scale

Overall economies of scale, ES, are defined as the percentage change in cost when all outputs increase by a common factor. In equation (4.2), scale economies are measured as:

$$ES = \frac{\partial lnC}{\partial lnq_1} + \frac{\partial lnC}{\partial lnq_2}$$

i.e.,  $ES = \alpha_1 + \alpha_2 + \gamma_{11} \ln q_1 + \gamma_{22} \ln q_2 + \gamma_{12} (\ln q_1 + \ln q_2) + (\eta_{11} + \eta_{21}) \ln p_1 + (\eta_{12} + \eta_{22}) \ln p_2.$  (4.4)

Note that scale economies are a function of the output levels,  $q_1$  and  $q_2$ , therefore the ES measure is not invariant to scale and is dependent on the output mix. If ES is less than 1, there are economies of scale since costs increase proportionately less than output. Values of ES equal to or greater than 1 imply constant returns or diseconomies of scale respectively. Partial economies of scale,  $ES_i$ , and marginal costs of each output,  $MC_i$ , can be computed from equation (4.2) as:

$$ES_{i} = \frac{\partial \ln C}{\partial \ln q_{i}} = \alpha_{i} + \gamma_{ii} \ln q_{i} + \gamma_{ik} \ln q_{k} + \Sigma_{j} \eta_{ij} \ln p_{j}, \text{ and}$$

$$MC_{i} = \frac{C}{q_{i}} \left(\frac{\partial \ln C}{\partial \ln q_{i}}\right) = \frac{C}{q_{i}} \left(\alpha_{i} + \gamma_{ii} \ln q_{i} + \gamma_{ik} \ln q_{k} + \sum_{j} \eta_{ij} \ln p_{j}\right) \quad i, k = 1, 2. \quad (4.5)$$

Equation (4.5) requires an important gualification. When there is joint production of two (or more) outputs, it is not possible to measure the cost of the ith output "because this cost is also attributable to other outputs" [61]. The production of loan and deposit services by a bank is not exactly analogous to the typical wool-and-mutton example of jointness in production. However, it can be argued that some degree of technological interdependence exists in the production of loans and deposit services. In some cases, this interdependence approaches the wool-mutton analogy when bank-client relationships are such that the provision of one of the services (e.q., a loan) is tied to the sale of the other (i.e., the opening of a deposit account). Furthermore, the case of financial intermediaries will also fit the definition of jointness that arises from the existence of fixed or quasi-fixed allocatable inputs in multi-output production [85]. Jointness in production, due either to technological interdependence or to the existence of binding constraints in the allocation of some inputs, invalidates the derivation of individual cost functions for

each output from the multi-product form of the cost function [85].2/

In terms of equation (4.5), the foregoing considerations imply that the ratio  $(C/q_i)$  is not an appropriate measure of the average cost of producing output i, since total costs C are attributable to both outputs  $(q_1 \text{ and } q_2)$ . Furthermore, it is not possible to derive separate cost functions for each output for analytical purposes (e.g., finding optimum levels of production). In fact, all analytical results obtained for one output will be conditional upon the level of the other output. Therefore, the use of equation (4.5) is constrained here to two purposes: first, to assess the effects of different variables on the marginal costs of producing each output, by looking at the signs and levels of significance of the coefficients involved in the equation. Second, to make point-predictions of the level of marginal costs for each output, at given levels of all variables involved in the expression for marginal costs (4.5). This second purpose has yet to overcome the problem of "allocating" total costs between the two outputs, in order to estimate the value of the average-cost ratio that enters expression (4.5). A procedure to overcome this allocation problem is described below.

Even though exercises in cost allocation among outputs are feasible in financial institutions [26,37], the output- $\overline{\frac{2}{\text{ This limitation also applies to multi-output production}}$ and profit functions.

cost-shares resulting from these allocation studies are conditional upon the output levels at which they were performed. An expression for the share of each output in total marginal costs that takes this conditionality into account has been defined by Laitinen [61]:

$$g_{i} = \frac{q_{i}}{\rho} \frac{\partial C}{\partial q_{i}} , \qquad (4.6)$$

where, g is the share of the ith output in total marginal costs,

- q, is the quantity of output i,
  - $\boldsymbol{\rho}$  is Laitinen's definition of total or overall marginal costs,

$$\rho = \Sigma_{i} q_{i} \frac{\partial C}{\partial q_{i}} , \qquad (4.7)$$

 $\partial C/\partial q_i$  is the individual marginal cost of output i. Expression (4.6) can be easily transformed into:

$$g_{i} = \frac{\partial \ln C}{\partial \ln q_{i}} \sum_{i} \frac{\partial \ln C}{\partial \ln q_{i}} = \frac{ES}{ES} , \qquad (4.8)$$

i.e., the share of the ith output in total marginal costs is equal to the ratio of the partial scale-economies measure associated with this output ( $ES_i$ ) over the value of overall scale-economies (ES). It is evident from (4.8) and (4.5) that the output-cost-share  $g_i$  depends on both the scale of production and the output mix. However, it can be easily computed using the values of  $ES_i$  and ES resulting from the estimated cost function.<sup>3</sup>/ The expression (4.5) for 3/ See B.1 in appendix B. marginal costs of production of output i can now be written as follows:

$$MC_{i} = \frac{C_{i}}{q_{i}} \frac{\partial lnC}{\partial lnq_{i}} = \frac{C_{i}}{q_{i}} (B_{i} + \gamma_{ii}lnq_{i} + \gamma_{ik}lnq_{k}),$$
  
$$i_{i}k = 1,2 \qquad (4.9)$$

where,

$$B_{i} = \alpha_{i} + \Sigma_{i} \eta_{ij} \ln p_{j}$$

#### Cost Complementarities (Economies of Scope)

.

Cost complementarities exist in multi-output production when the marginal cost of producing one output declines with increases in production of another  $output^{4/}$ . In the case of a banking firm producing loan and deposit services, cost complementarities exist if the marg\_nal costs of lending decrease as a result of increasing deposit activity, or vice versa. For example, an expansion of deposit services may reduce information costs in loan evaluation. In general, cost complementarity exist if:

$$\partial(MC_i) / \partial q_k = \frac{\partial^2 C}{\partial q_i^{\partial} q_k} < 0$$
.

This condition can be expressed in terms of the logarithms of the variables as:

<sup>4/</sup> Murray and White [69] refer to this relationship as "economies of scope". However, Benston, Berger, Hanweck and Humphrey [18] give a more strict definition for the concept of economies of scope. See also Panzar and Willig [73].

$$\frac{\partial^2 C}{\partial q_i \partial q_k} = \frac{C}{q_i q_k} \left( \frac{\partial^2 \ln C}{\partial \ln q_i \partial \ln q_k} + \frac{\partial \ln C}{\partial \ln q_i} \frac{\partial \ln C}{\partial \ln q_k} \right) < 0 . (4.10)$$

Since C, q<sub>i</sub>, q<sub>k</sub> are all positive, the sign of this second derivative is determined by the expression in parenthesis. Murray and White [69] indicate that in terms of the parameters of the cost function, a necessary condition for the existence of cost complementarity between loans and deposits is:

$$\gamma_{12} + \alpha_1 \alpha_2 < 0$$
 (4.11)

#### Elasticity of Substitution and Elasticities of Input Demand

The flexibility of resource allocation in the banking firm, as well as the responsiveness of management to price signals, can be assessed through the magnitudes of the elasticity of factor-substitution and the price-elasticities of factor demand. Uzawa [89] has shown that the Allen partial elasticity of substitution between factors of production,  $\sigma_{jh}$ , can be written in terms of the (dual) cost function as:

$$\sigma_{jh} = \left(\frac{\partial^2 \ln C}{\partial p_j \partial p_k} / \frac{\partial \ln C}{\partial p_j} \frac{\partial \ln C}{\partial p_h}\right) + 1 .$$
(4.12)

This expression can be transformed and expressed in terms of the parameters of the translog cost function (4.2) and the factor shares (S<sub>j</sub>), so that the Allen partial elasticities of substitution can be computed as: $\frac{5}{}$ 

<sup>5/</sup> See B.2 in appendix B for details on the transformation of equation (4.12) into (4.13).

$$\sigma_{jh} = (\delta_{jh} + s_j s_h) / s_j s_h, \ \sigma_{jj} = (\delta_{jj} + s_j (s_j - 1)) / s_j^2,$$
  
j,h = 1,2. (4.13)

In addition, the price elasticities of demand for inputs,  $e_{jh}$ , can be obtained using the estimated values of  $\sigma_{jh}$  and the factor shares (see Binswanger [21]).

$$e_{jj} = \sigma_{jj}S_{j}, e_{jh} = \sigma_{jh}S_{h}, j, h = 1,2$$
 (4.14)

It is clear from (4.13) that if all  $\delta_{jh} = 0$ , then the elasticities of substitution are independent of factor prices, and equal to one for  $j \neq h$ .

#### 4.1.2. Output Definition

It has been shown that the estimation of cost functions is particularly sensitive to the definition of output of banking institutions [15,17,18]. However, the difficulties associated with this definition have induced some researchers to avoid the cost-function approach for the analysis of banking behavior and concentrate their efforts on alternative approaches, such as profit-function analysis [68].

The use of the (dollar) value of loans and deposits as a measure of a bank's output has been criticized on the basis that this definition amounts to the use of a total sales concept instead of a value-added concept to measure firm output [15,30]. A preferred conceptual approach states that banks essentially produce services in the form of deposits and loan accounts [8,15,30].

A number of studies have used the number of loans as the definition of output [30,71]. Gheen [30] has argued that loan accounts can be considered a homogeneous measure since lending operations tend to be standardized. However, there are a number of attributes of each particular account that may create sharp differences in their administrative costs. Among these characteristics of loan accounts, loan size can reflect the degree of riskiness involved in the loan operation and therefore should be another variable entering the cost function. Whether loan risk increases or decreases with loan size is a controversial issue. "Everything else constant" it could be expected that the riskiness of a loan increases with the loan size. A larger loan-size implies a higher leverage on the borrower's wealth thus reducing the borrower's expected ability to repay. Therefore, if lenders perceive large loans as riskier ventures they will tend to require additional information, more evaluation, and closer supervision, as compared to smaller loans.

Additional requirements will create further loanadministration costs associated with loans of large amounts. It is hypothesized here that the marginal cost of a loan is an increasing function of loan size. However, it can be expected that the increase in costs is less than proportional to the increase in loan size. Therefore, on a per

unit of money basis the marginal cost of lending will be a decreasing function of the loan size.

As for deposit accounts, their size can play a discriminatory role, similar to that assumed for loan size in the preceding discussion, though through different mechanisms. Large deposit accounts may be associated with "preferred" customers who receive special or additional services, i.e., involve higher costs for the financial intermediary. Also, large accounts may be associated with numerous transactions, thus making them more costly to service as compared to less active accounts. Furthermore, the risk of withdrawal becomes higher as the size of deposit-balances increase. It is thus expected that the marginal cost of handling depositaccounts increases as the deposit-size increases. However, as in the case of loans, the marginal cost per lempira mobilized in deposits is expected to decrease with deposit-size.

Summarizing, two measures of output are used in the estimation of the cost function (4.2), (a) the number of loans and deposits, and (b) the value of loans and deposits. Loan-size and deposit-size variables are included in the model to account for the heterogeneity of loans and deposit transactions. These variables are included in the cost function (4.2) in interactive form with the output levels:

 $\theta_1 \ln q_1 \ln LS + \theta_2 \ln q_2 \ln DS$ , (4.15) where, LS is the average loan size,

DS is the average size of deposit-balances.

In this way, scale-economies indicators and the marginal costs of production become dependent upon the average size of loans and deposits.

An alternative to the foregoing two-output definition of the production of banking services is the use of an index of bank output that combines the number of loans and deposits with different weights. It has been argued that the use of an index gives a more comprehensive view of the overall "bank productive efficiency" [17], particularly when dealing with scale-economies questions. It must be pointed out however, that the interpretation of other results obtained in the estimation of a cost function, such as marginal costs of loans or deposits, becomes less clear when the firm's output is measured through an index that combines loans and deposits. A discussion of these indices is found in Benston, Hanweck, and Humphrey [15,17]. These authors have found that the use of a Divisia index, instead of a simple sum of loans and deposits, generates differences in the estimates of economies of scale and the optimum size of banking institutions, even though these differences are admittedly not substantial [15]. A Divisia index corresponds to the sum of the number of loans and depositaccounts adjusted (weighted) by the quantity shares and unit costs of each type of account [17].

As will be discussed later (section 4.1.4), a problem in implementing this index-number approach in the case of

the Honduran banks, is the absence of a data-reporting system such as the FCA in the United States. While quantity shares by type of account could be computed with individual banks' data, unit costs by type of account are not recorded by Honduran financial institutions. Here the approach will be to use a simple sum of loans and deposits as a crude measure of aggregate output. This simple sum is equivalent to a Divisia index when quantity shares and unit costs are the same across all observations. The results obtained with this single-output definition will be contrasted against those associated with the two-output approach, in terms of the magnitudes and significance of the estimates of scale economies.

#### 4.1.3. Proxies for Regulation

#### Interest-Rate Ceilings

The statutory deposit rate is one of the variables that will represent the degree of repression on interest rates. As stated before (section 3.1), the statutory deposit rate d would be a good proxy for the difference between an unobserved equilibrium rate  $d_e$ , and the level of d itself, under the assumption that  $d_e$  is constant. However, while this assumption may be valid for the  $d_e$  rate in <u>real</u> terms, it cannot be maintained when <u>nominal</u> rates are considered in an inflationary environment. A constant equilibrium rate in real terms results in a fluctuating nominal equilibrium rate that varies with the rate of inflation. As a result, the

inflation rate can be utilized as a proxy-variable that will capture the variation of the equilibrium nominal rate  $d_e$ . The difference  $(d - d_e)$  will then be measured as the difference between the statutory deposit rate d and the rate of inflation, i.e., the ex-post real deposit rate (d-p), where  $\dot{p}$  denotes the rate of inflation. As discussed before, restrictions on the level of the deposit rate create costs of regulation-avoidance in order to compete for deposit mobilization. Therefore, an increase in the real depositrate should induce the substitution of explicit interest for implicit premia to depositors, thus decreasing the administrative costs of mobilizing deposits.

Interest-rate regulations include ceilings on lending rates. These constraints on the interest rates that can be charged on loans generate costs of implementing loan procedures that allow lenders to discriminate among borrowers. The higher the ceiling on the lending rate, the more flexible and less constrained are these lending operations. Again, explicit interest charges can take the place of implicit charges, and lenders' costs can be reduced through the adoption of less-complicated loan procedures. This effect will not only reduce lenders' costs but, as will be argued later, will also benefit borrowers since transaction costs associated with borrowing will be reduced as well. The ex-post real lending rate (1-p) computed using the interest-rate ceiling 1 and the rate of inflation p, will be specified in the cost function as another indicator

of interest-rate regulations, along with the real deposit rate defined above. The relevant expression that will be included in the cost function (4.2) can be written as:

 $\lambda_1(d-\dot{p}) + \lambda_2(l-\dot{p})$  (4.16) where, d,  $\dot{p}$ , and l have been defined above. The foregoing discussion about the effects of interest-rate regulations indicates that the signs of  $\lambda_1$  and  $\lambda_2$  should be negative.

#### Loan Targeting and Special Credit Projects

The most important financial intermediary dealing with targeted funds and special credit projects in Honduras throughout the period considered in this study (1971-1982) has been the National Agricultural Development Bank (BANADESA). Only recently have some private banks participated in externally-funded projects sponsored by the World Bank, and to a lesser extent by the U.S. Agency for International Development (USAID). Therefore, the analysis of the effects of loan-targeting on intermediation costs will concentrate on the development bank. A separate study has reported and analyzed the lending costs associated with different sources of funds in both the development bank and a private commercial bank dealing with agriculture [26]. The findings of this cross-sectional comparative study will be referred to when discussing the results obtained in the econometric analysis of targeting in the development-bank case.

The effects of targeted funds and special credit projects on intermediation costs were discussed in section 3.1. These programs have a direct effect on lenders' costs due to the additional accounting and record-keeping personnel and materials necessary to comply with the reporting requirements of these programs. Typical sources of targeted funds in Honduras are the central bank, and donor agencies. Central-bank funds correspond mainly to crop-specific lines of credit designed to provide short-term financing to small and medium-size farms. Foreign funds usually come in the form of special projects targeted to specific activities, and tend to include a larger proportion of long-term loans. In what follows, the term "external funds" will be used to refer to both central-bank and foreign funds. The other, non-targeted, source of funds for BANADESA are demand, savings and time deposits from public-sector institutions, and from the public at large.

It is further hypothesized that the effect of targeted funds on costs in the development bank includes a "ratchet" effect. That is, the increased level of costs growing out of a new credit project contracted by the bank does not decline to the previously existing cost level once the loan funds have been disbursed to the ultimate borrowers. Additional resources are employed or purchased at the beginning of the project in order to comply with the project's targeting requirements, but these resources are

not laid-off or sold once the funds are disbursed. The cost function will thus incorporate a set of variables that capture the effect of targeted funds under this "ratchet" effect hypothesis. Three indicator variables  $(S_i, i=1,2,3)$ are defined to account for the effect of the three different sources of funds: deposits, central bank, and foreign funds. In order to capture the influence of targeted funds under the "ratchet" effect hypothesis,  $S_i$ 's are defined so that  $S_i > 0$  if the value of funds coming from source i has increased over the level observed in the previous year, otherwise  $S_i = 0$ . Specifically, the value of  $S_i$  in year t  $(S_{i+})$  will follow a three-point distribution, such that:

 $S_{it} = 0, \text{ if } \Delta_{it} \leq 0$   $S_{it} = 1, \text{ if } 0 < \Delta_{it} \leq (1/2) \Delta_{itm}$   $S_{it} = 2, \text{ if } (1/2) \Delta_{itm} < \Delta_{it} \leq \Delta_{itm},$ 

where, Δ<sub>it</sub> stands for the difference between the amount of funds coming from source i in year t, and the amount of these funds in year t-1, Δ<sub>itm</sub> is the maximum value of this difference observed

over the period covered by the data (1971-1982). A combined variable, S<sub>23</sub> is similarly defined to account for the effect of all external funds combined (central-bank and foreign funds together). The "ratchet effect" hypothesis implies that a positive sign is expected in the coefficients of the S<sub>i</sub> variables that capture the

effects of targeted funds, i.e., central-bank and foreign funds. The estimation will consider the possibility that these effects may be lagged, particularly for foreign-donor funds, since this source of funding is often in the form of special projects with delayed period of disbursement and expenditures. Consequently, external funds combined, and foreign funds alone are also specified with a one-year lag, to capture the lagged effect increases in these sources of funds are likely to have on costs.

Summarizing, the set of indicator variables that will enter the cost function to address the issue of loantargeting can be written as:

 $\omega_1 S_1 + \omega_2 S_2 + \omega_3 S_3$ , (4.17) or alternatively:

 $\omega_1 S_1 + \omega_{23} S_{23}$ , where,  $S_i$ 's have been defined above and  $\omega_i$ 's are the corresponding parameters. The variables  $S_2$  and  $S_{23}$  are also included with a one-year lag in various regressions.

#### 4.1.4. Delinquency and Default

The treatment of delinquency will focus on the effects that variations in the delinquency rate are likely to have on loan-administration costs. These cost-increasing effects of delinquency are due to the allocation of additional resources to loan recovery, and costs associated with legal

procedures to handle loans in default. Special departments or task-forces are often created, or existing units reinforced with additional personnel and resources to deal with delinquency.

A likely second-round effect of increases in loan delinquency on the costs of financial institutions emerges from the tightening of lending procedures. Banks tend to become more cautious in loan-approval, and upgrade their loan-evaluation departments in the presence of a growing delinquent portfolio. This second-round effect of delinquency is likely to operate with a lag, whereas the allocation of additional resources to loan recovery discussed above may be an almost contemporaneous reaction to increasing delinquency.

An important consideration in the treatment of delinquency and default is the classification of delinquent balances by age. Banks usually will not take actions on balances until they are 30 days overdue, nor will they actively pursue loans more than 4-years delinquent. Therefore, the age classification of delinquent balances should be taken into account in defining the variables that account for the effects of delinquency on administrative costs. Delinquent balances between 90 days and 4 years overdue will be considered here as "active delinquency". The expression included in the cost function to capture the effects of delinquency can be written as follows:

- where, DR is the delinquency rate computed as the ratio of total overdue balances over total loans outstanding,
  - DA is the proportion of total delinquent balances that is more than 90 days and less than 4 years overdue.

Both DR and DA will be also included with a one-year lag to test for the second-round effects of delinquency. In all cases, cost-increasing effects of delinquency should be reflected in positive signs of the estimated parameters involved in expression (4.18).

#### 4.1.5. Data and Estimation

The estimation of the cost function will draw upon two separate data sets. The first data base corresponds to 28 branches of the National Agricultural Development Bank (BANADESA) over the 12-year period 1971 through 1982. This bank is referred as the "development bank". The second data set was obtained from the largest private commercial bank of the country (Banco Atlantida), that will be referred to as the "private bank". This bank has a network of over 50 agencies and offices throughout the country, that is organized into 16 main branches with independent accounting The same 12-year period (1971-82) is covered by records. this data set. Data were gathered through the Economic Studies departments of both banks, and in many cases directly from the branches. Financial-sector and national-income variables were recorded from Central Bank publications.

76

(4.18)

All variables have been expressed in real terms (lempiras of 1966) using the country's implicit GDP deflator. Exceptions are the number of loans, the number of deposit accounts, and several dummy or categorical variables already defined in previous sections. Variable definitions are briefly outlined below.

(a) Costs. Total non-financial operating expenses, net of depreciation and provisions for bad debt. The sources of these data in both banks were the revenue-expenditure statements of the branches, produced by the accounting divisions. (b) Outputs. Alternative definitions of output utilized here are: (i) the number of loans, and the number of deposit accounts (as separate outputs), (ii) the value of loans, and the total amount of deposit balances at year-end (also as separate outputs), and (iii) an index of aggregate bank output computed as the sum of loans plus deposits. $\frac{6}{}$ Information on the number of loans was not available from the private bank, excepting the last two years of the series. Therefore, the definition of output in this case was constrained to those stated in value terms. (c) Factor Prices. Two factors are considered here: labor, and capital goods. The price of labor services  $(p_1)$  is measured as total personnel costs including benefits and social security payments divided by the total number of employees. A unit price of capital services (p<sub>2</sub>) is proxied 6/ See section 4.1.2 for a discussion of these definitions.

by the ratio of depreciation plus rents paid over the total value of loans plus deposit balances. This proxy was found positively (and significantly) correlated with the implicit deflator of gross domestic capital formation in the national accounts. A posteriori support for this proxy selection is the fact that the factor-price homogeneity condition is met in most (unrestricted) estimations of the cost function. (d) Loan Size and Deposit Size. These variables are included in the model to account for the heterogeneity of loans and deposit accounts (see section 4.1.2). Loan size is computed as the ratio of total value of loans over the total number of loans. Deposit size is calculated as the ratio of total deposit balances at year-end over the total number of deposit accounts. Since the number of loans was not available in the case of the private bank, loan size could not be computed for this bank.

(e) Real deposit-rate and real lending-rate. Two approaches were used to compute the nominal deposit rate (d). First, it was calculated as the arithmetic mean of all deposit-rate ceilings established by the central bank. This average deposit-rate will be denoted  $d_c$ . Second, the implicit rate paid on deposits by each bank was computed as the ratio of total interest payments over total deposit balances. This second variable is denoted  $d_p$ , and can be interpreted as the bank's average cost of funds under regulated conditions. In both cases, the real deposit-rate is obtained by

substracting the inflation rate  $\dot{p}$  from the nominal deposit rate. The real lending rate is proxied substracting the inflation rate from the overall ceiling established for the nominal interest rate on loans, l. The rate of inflation is calculated as the l2-month variation in the implicit GDP deflator.

(f) Delinquency rate. Only the development bank provided information on delinquency. The ratio of overdue balances over total loans outstanding was computed as a measure of the delinquency rate. The age-classification of delinquent balances in the development bank does not break down balances more than 1 year overdue. Therefore, the variable that accounts for "active delinquency" (DA in expression 4.13) is redefined to include all delinquent balances over 90 days overdue.

Estimation of the translog cost function (4.2) was undertaken independently for the two banks, both as a single equation (by OLS), and as a cost system with the cost-share equations (4.3). Since cost shares must add to 1, one of these equations is redundant and therefore is dropped from the system. The remaining equations in the system, the cost function and the labor-share equation, are seemingly unrelated and the estimation of this two-equation system utilizes a generalized-least-squares procedure. Joint estimation of the cost system should improve the efficiency of the parameter estimates [21,69,70]. However, Benston,

Hanweck and Humphrey [17] argue that these gains in efficiency are relatively small and undertake their estimations using OLS on the cost function above. As will be shown later, aside from efficiency gains, there may be important differences in the magnitude of the estimated parameters resulting from different estimation procedures. As a consequence, the scale-economies measure (and other parameters) will differ depending on the estimation technique.

#### 4.2. Borrower's Costs

The methodological issues involved in the empirical assessment of the general model set forth in section 3.3 are discussed here. This model was formulated in general form as:

$$T = T(B, i)$$
, (4.19)

where,

- T is the borrowing (non-interest) transaction costs per loan
- B is a vector of risk-related characteristics of the loan operation (loan size, firm size, loan use, etc.)
- i is the explicit interest rate that can be charged on loans by the lender.

The definition of transaction costs that will be used as the dependent variable in this general model is outlined below. The elements involved in a trade-off equation between transaction costs and explicit interest rate are

then briefly discussed. This discussion includes the selection of proxies for loan risk, and the specification of a mathematical form for the general expression (4.19) above. Since this section of the empirical work focuses on agricultural loan operations, the expressions farmers, clients, and borrowers will be used interchangeably.

#### 4.2.1. Transaction Costs

Transaction costs are defined here as all those noninterest explicit and implicit expenses incurred by the borrower in the process of obtaining a loan. These costs occur at different stages in the sequence of procedures established by the lending institution, i.e., application and documentation, approval, and disbursement. Explicit expenses refer basically to the following:

- (a) Cost of transportation, lodging and meals when travelling to the office of the institution granting the loan, or to other places with the purpose of obtaining related documents;
- (b) Fees, taxes or other charges associated with the issuing of documents, registration of guarantees or collateral, contracts and the like; and,
- (c) Other explicit charges imposed by the lending institution in the process of handling the loan application.

The implicit transaction costs directly related to borrowing correspond to the value of the time foregone by clients in negotiating and securing their loans.

When transaction costs per loan, T, are expressed as a percent of the loan amount, we obtain the implicit element  $(\tau)$  involved in the interest-rate factor discussed in section 3.3.

# 4.2.2. A Trade-off Equation between Transaction Costs and the Interest Rate

The general model defined by expression (4.19) corresponds essentially to a trade-off equation between transaction costs of borrowing and the explicit interest rate. In this equation, the risk-related elements of vector B will play the role of shift and/or interaction variables. Likewise, the model will account for behavioral or managerial differences between loan sources (development bank, private banks, credit unions) that may affect both the level of borrowing-transaction costs and the magnitude of the trade-off between these and the explicit interest rate charged on the loan.

### Characteristics of the Loan Operation

The components of vector B in the model refer primarily to variables associated with the risks of lending from the point of view of the lender. Key elements underlying these risks can be summarized as follows:

(a) Firm size. This factor is related to the client's wealth and loan-repayment capacity. <u>Ceteris paribus</u>, a wealthier client represents a larger debt capacity and a better ability to repay. Therefore, as the size of the firm increases, transaction costs of borrowing should decrease.

In the case of agricultural loans, a logical proxy for firm size is the area of the farm. However, the heterogeneous nature of land quality in Honduras might affect the validity of this variable as a proxy for size. Other size measures available in the data will be specified as alternatives to farm area. These other proxies for size include the number of hired laborers, total labor costs, cultivated cropland, and the value of collateral declared by the borrower.

(b) Loan amount. The relationship between this characteristic and the level of risk has been discussed before (see section 4.1.2.). It is expected that, everything else constant, the riskiness of a loan increases with loan size, thus creating additional transaction costs for both the lender and the borrower. At this point, it is important to distinguish between transaction costs per loan and transaction costs per lempira. If transaction costs per loan increase in exactly the same proportion in which the loan amount increases, transaction costs per lempira will remain constant. The hypothesis set forth here is that as loan size increases, transaction costs per loan increase at a

decreasing rate; therefore the magnitude of transaction costs per lempira borrowed will decrease as the loan amount increases.

(c) Loan use. This feature refers to enterprise-type characteristics associated with different levels of production risks, marketing risks and built-in collateral. The end-use of the loan may be considered relevant by the lender in any of the following cases. First, if the lender is not aware of diversion and fungibility, and/or believes that loan-monitoring will guarantee that loan funds are devoted to the stated end-use. Second, if the stated end-use is known to be the single most important activity of the borrower. Third, if the activity allegedly being financed with the loan is developed under a three-party marketingrepayment agreement between the borrower, the lender, and a marketing firm. If none of these conditions is present, loan-use in general should not be a factor in the lender's assessment of loan risk. The discussion that follows assumes that some consideration is given to the stated enduse of loans in evaluating the risks of lending.

The most important end-uses of agricultural loans are represented by a set of dummy variables in the model set forth below. Among these end-uses, basic grains may be considered one of the riskiest activities in Honduran agriculture. This consideration is due to the crops' exposure
to weather-related phenomena (droughts in the south, floods in the north) that frequently reduce yields and profitability. Furthermore, despite government efforts to raise farm-level prices these have not improved significantly for producers, due in part to increasing marketing margins and imperfect marketing structures. $\frac{7}{}$  This adds uncertainty of marketing to the production risks mentioned above, thus making basic grains possibly a less-preferred lending activity. It follows from this reasoning that loans for basic grains will carry higher transaction costs than loans for other uses.

Transaction costs (and risks) associated with loans to export-crops can be expected to be lower-than-average. These crops, such as sugar, tobacco, cotton, and others, are grown under marketing-repayment schemes agreed upon between processing firms, banks and producers. These agreements minimize marketing risks and in many cases eliminate default risks for the lender. Furthermore, the processing/marketing agent usually provides technical assistance that reduces production risks through the adoption of improved inputs and practices. Finally, the behavior of transaction costs in the case of livestock activities in Honduras is more difficult to predict. On the one hand, tropical diseases, inefficient cattle-raising techniques, and poor animalhealth resources make livestock a rather unreliable source 7/ See Pollard, Graham, and Cuevas [75].

of income for farmers. Furthermore, dairy products have controlled prices at retail, and beef markets have been subject to depressed price conditions for several years. These production and marketing conditions mean that livestock should be seen as a "bad risk" by lenders. However, livestock (as well as machinery) is a typical end-use with built-in collateral, since the animals purchased can be made a part of the loan guarantee. This particular feature makes the expected response of lenders to these loans ambiguous, and therefore the expected sign of the differential transaction costs associated with these loans is also ambiguous.

### Specification

The trade-off equation between transaction costs and interest rate can now be written more explicitly as:

T = f(A,L,i,D,U) (4.20)

where,

A is the area of the farm,
L is the loan amount,
i is the explicit interest rate charged on the loan
D is a set of dummy variables representing different loan sources for the borrower,
U is a set of dummy variables accounting for different end-uses of loans.

Most previous studies on borrowing transaction costs have been descriptive in the presentation and treatment of their results [3,40]. Therefore, these studies do not

provide a priori elements to select a specific mathematical form for the trade-off equation (4.20). In other cases, the market of loanable funds has been modeled as a supply-demand system specifying linear forms for the equations involved but neglecting transaction costs [44]. The issue of simultaneity requires particular attention, and is discussed below.

Even under the price-setting framework set forth in section 3.3, the loan-amount variable L on the right-hand side of equation (4.1) may be considered endogenous rather than pre-determined. This loan amount would be a point on the borrower's loan-demand function outlined in section 3.3, which, together with the trade-off equation (4.20), would conprise a two-equation system with T and L as endogenous variables. The system will be recursive if it is assumed that borrowers neglect transaction costs when making borrowing decisions, i.e., the loan demand is a function of the explicit interest rate, and the individual's resource endowment,

$$L = h(i,W)$$

(4.21)

where,

i is the explicit interest rate,

W represents the borrower's resource endowment that influences the potential size of his investment projects,

 $L_{i}^{1} < 0, L_{M}^{1} > 0.$ 

If on the other hand it is assumed that borrowers do consider transaction costs as part of the loan price, the system should be specified in a simultaneous-equations model. In this case the loan-demand function can be written as:

$$L = g(i,T,W)$$
 (4.22)

where,

 $L_{i}^{*} < 0, L_{m}^{*} < 0, L_{W}^{*} > 0.$ 

As will be explained in the following section, the data base utilized to analyze borrower's transaction costs do not include information to measure the borrower's resource endowment. Therefore the estimation of a system of equations with T and L as endogenous variables is not attempted in this study. The empirical approach undertaken here estimates the single equation (4.20), recognizing the shortcomings involved if the "true" model is the one that considers the loan amount L as endogenous. Under the hypothesis that loan demand does not depend on transaction costs, as in equation (4.21), the problem of estimating the single equation (4.20) would be multicollinearity, with no bias in the parameter estimates. However, if the true model involves a loan-demand function such as (4.22) where the loan amount depends on the magnitude of transaction costs, then the trade-off equation (4.20) includes a variable (i.e., loan amount) that is correlated with the error term. In this case, the single-equation estimation of (4.20) would

yield biased and inconsistent estimates of the parameters in the model. It can be shown that the bias is negative for the coefficient associated with the loan-amount variable, and positive in the case of the interest-rate variable (see appendix B, section B.3). As will be clear in chapter 6, these potential biases do not affect the essence of the conclusions derived from the results obtained in the estimation of the transaction-costs function (4.20). The mathematical form of this function is specified below.

The trade-off equation between transaction costs and interest rate is specified as a generalized power function [28], in log-linear form:

$$\ln T = a_0 + a_1 \ln A + a_2 \ln L + a_3 \ln(i) + b_1 D_1 + b_2 D_2 + c_1 U_1 + c_2 U_2 + c_3 U_3$$
(4.23)

where,

•

- T is the borrowing (non-interest) transaction costs per loan, A is the area of the farm,
- L is the loan amount,
- i is the explicit interest rate that can be charged on the loan by the lender,
- D1 and D2 are dummy variables that account for deviations of T in private banks and credit unions with respect to the development bank, that is used as the base or level of reference,
  - D1=1 if the lender is a private bank,
  - $D_1=0$  otherwise,
  - $D_2$ =1 if the lender is a credit union,

 $D_2=0$  otherwise,

U1, U2, and U3 are dummy variables defined to capture the effects on transaction costs of different loanuses: basic grains, export crops, and livestock, as deviations with respect to a miscellaneous end-use category conformed by all other end-uses in agriculture (land purchases, trade, vegetable crops, and others),

 $U_1$ =1 if the stated end-use of the loan is basic grains,  $U_1$ =0 otherwise,

 $U_2=1$  if the stated end-use of the loan is export crops,  $U_2=0$  otherwise,

 $U_3=1$  if the stated end-use of the loan is livestock,  $U_3=0$  otherwise.

The generalized power function was chosen in order to obtain direct estimates of the elasticities of transaction costs with respect to the explanatory variables. Also, this functional form guarantees that transaction costs will be zero if there is no loan operation (L=0), and that predicted values will be non-negative.

From the discussion about the characteristics of loan operations and their relationship with risk, it follows that the expected signs of the partial derivatives in equation (4.23) are:

(a) Area of the farm, A, proxy for borrower's wealth,
 lnT/lnA=a1 < 0,</li>

as borrower's wealth increase transaction costs of borrowing decrease.

(b) Loan amount, L

 $0 < \partial \ln T / \partial \ln L = a_2 < 1,$ 

as the loan amount increases, transaction costs per loan increase but at a decreasing rate. If this holds then  $\partial \ln \tau / \partial \ln L = (a_2 - 1) < 0$ , i.e., transaction costs per lempira  $(\tau = T/L)$  are a decreasing function of loan amount. Note that, from the definition of elasticity, the slope of the transaction costs (per loan) curve can be written as:

 $\partial T/\partial L = a_2(T/L) = a_2\tau$ , 0 <  $a_2\tau$  < 1.

Since in general the ratio of transaction costs to loan amount (transaction costs per lempira borrowed) is less than 1, the slope of the transaction costs per loan curve will be positive but less than 1. The value of this slope will decrease as loan amount increases.

In turn, the slope of the transaction costs per lempira curve can be written as:

 $\partial \ln \tau / \partial \ln L = (a_2 - 1)(\tau / L) = (a_2 - 1)(T / L^2)$ , < 0. In other words, the absolute value of the slope of this function is inversely related to the loan amount. This downward sloping curve will be very steep for small loans, and relatively flat for large loans.

(c) Interest rate, i

 $\partial \ln T / \partial \ln i = a_3 < 0,$ 

transaction costs (implicit interest) and the (explicit) interest rate charged on loans are substitutes. In other words, a trade-off exists between transaction costs and the interest rate. Note that under this functional form, the

elasticity of transaction costs per lempira with respect to the interest rate is also a<sub>3</sub>,

 $\partial \ln \tau / \partial \ln i = a_3 < 0.$ 

Therefore, the absolute change in transaction costs per lempira induced by a one-unit change in the explicit interest rate is given by

 $\partial \tau / \partial i = (\partial \ln \tau / \partial \ln i) (\tau / i) = a_3 (T / iL), < 0.$ 

In short, the absolute magnitude of the change in transaction costs due to a change in the interest rate is directly proportional to the ratio of transaction costs over interest costs (T/iL). Also, the response of  $\tau$  to a change in the interest rate is inversely proportional (in absolute value) to the level of the explicit interest rate, and to the loan amount. This trade-off between transaction costs and interest rate will be discussed further below.

(d) Loan sources,  $D_1$ ,  $D_2$ 

The hypothesis that in general the loan source (type of lender) is a factor in determining the level of transaction costs can be tested under the joint null hypothesis  $b_1=0, b_2=0$ . It is expected that this hypothesis will be rejected, i.e., that loan source is a significant factor in the transaction-costs function.

Since private banks may be expected to be more efficient in "passing-on" intermediation costs to the borrowers, it is anticipated that

 $\partial \ln T / \partial D_1 = b_1 > 0$ .

In other words, it is expected that transaction costs of borrowing from private banks will be higher than those associated with development-bank loans, which serve as level of reference in equation (4.24).

Also, given that loan procedures are in general less complicated in credit unions it can be expected that borrowing from these institutions will imply lower transaction costs, i.e.,

 $\partial \ln T / \partial D_2 = b_2 < 0.$ 

(e) Loan uses,  $U_1$ ,  $U_2$ ,  $U_3$ 

The significance of end-use as a determinant of the level of transaction costs can be analyzed testing the joint null hypothesis  $c_1=0$ ,  $c_2=0$ ,  $c_3=0$ . The conditions under which end-use would matter in the determination of loan procedures and transaction costs have been discussed before. This discussion does not lead to any specific expectation about the result of the joint test indicated above. As regards the signs of the parameters associated with specific end-uses, the discussion in the preceding section suggests that

 $\partial \ln T/\partial U_1 = c_1 > 0$ , for basic grains,  $\partial \ln T/\partial U_2 = c_2 < 0$ , for export crops, and  $\partial \ln T/\partial U_3 = c_3 \ge 0$ , for livestock.

The specification of the transaction-costs equation (4.23) determines that the magnitude of the elasticity of transaction costs with respect to changes in the explicit

interest rate will be a constant parameter  $(\partial \ln T/\partial \ln i = a_n)$ . However, it can be expected that this elasticity will be affected by other variables in the model. Specifically, lenders may react differently to changes in the interest rate when dealing with very small loans or very small firm sizes. In these cases, the loan procedures that create borrower's transaction costs tend to be more rigid, therefore the magnitude of the trade-off or substitution between transaction costs (implicit interest) and the explicit interest rate may be smaller than average, in absolute terms. Also, the response of different lenders to changes in the interest rate may differ due to different institutional structures, overall flexibility of loan procedures, and managerial goals and capabilities. These considerations suggest the inclusion of interaction effects in equation (4.23), in order to account for the effect of very small loan sizes, firm sizes, and different lenders, on the elasticity of transaction costs with respect to the interest rate. The term agln(i) in equation (4.23) is then substituted by the following expression:

 $a_{30}ln(i) + a_{31}Sln(i) + a_{32}D_1ln(i) + a_{33}D_2ln(i)$ +  $a_{34}Fln(i)$  (4.24)

where,

S is a dummy variable for loan-size category, S=1 if the loan amount is less than or equal to L.2,000, S=0 otherwise,

- F is a dummy variable for farm-size category,
  - F=1 if the area of the farm is less than or equal to
     20 hectares,

F=0 otherwise,

i, the interest rate, and the dummy variables for loan source  ${\rm D}_1$  ,  ${\rm D}_2$  have been defined before.

With this change in specification, the trade-off between transaction costs and the interest rate in elasticity form is given by

 $\partial \ln T / \partial \ln i = a_{30} + a_{31}S + a_{32}D_1 + a_{33}D_2 + a_{34}F.$  (4.25)

It is expected that in all cases the sign of the elasticity will be negative, i.e.,  $\partial \ln T/\partial \ln i < 0$ . The interaction variables now included in expression (4.25) will only affect the magnitude of this elasticity. The discussion above is consistent with expected signs of the coefficients in (4.25) as follows:

 $a_{31} > 0, a_{34} > 0$ 

i.e., the absolute value of the elasticity is reduced when dealing with very small loans or very small farm sizes.

 $a_{32} < 0, a_{33} > 0,$ 

Private banks will be more responsive to changes in the interest rate, due to a greater flexibility of procedures and better managerial abilities. Even though loan procedures are less complicated and fairly flexible in credit unions, their management is not likely to respond actively to changes in the economic environment, therefore the expected sign of a<sub>33</sub> is ambiguous. Finally,  $a_{30}$  is also expected to be negative. This is the value of the elasticity in the case of loans larger than L.2,000 (S=0), granted by the development bank ( $D_1 = D_2 = 0$ ) to borrowers with farms of more than 20 hectares (F = 0).

### 4.2.3. Data and Estimation

The estimation of the borrower's transaction-costs function (4.23) in its original form, and with the modifications introduced by expression (4.24), will draw upon field-level data obtained in a survey undertaken in July-August 1981. This survey included some 200 observations of farmerclients of the development bank (BANADESA), private banks, and credit unions. The interviews were conducted in four important agricultural regions of Honduras, by an unusually competent survey team under the author's supervision. The survey questionnaire was designed to measure the costs incurred by borrowers when applying for and obtaining loans from institutional sources. These costs include the costs of transportation, lodging and meals incurred in trips to the office of the institution or other places, with purposes related to the loan application or to its disbursement. Fees, taxes or other charges associated with the issuance of documents, registration of guarantees, and contracts, were also documented, as well as miscellaneous charges imposed by the lending institution. The time spent by the farmer in negotiating and securing the loan was also recorded, in order to compute an imputed cost that is added to the other

(explicit) costs of borrowing, to obtain the total borrower's transaction costs.

The explicit interest rate paid on the loan, the loan amount, farm size, hired labor, and end-use of the loan are among the variables documented in the survey. However, the emphasis on measuring transaction costs of borrowing and other related variables constrained the measurement of different components of the resource endowment available to the farmer. This restriction makes it difficult to attempt the specification and estimation of a loan-demand function that completes the system of equations referred to in section 4.2.2 above. Instead, a single-equation approach will be used to estimate the transaction-costs function (4.23), and its extended version with expression (4.24). The econometric problems that might affect this estimation procedure have been discussed above (see section 4.2.2).

## Chapter V

### Lender's Intermediation Costs

This chapter first reviews the estimates of the cost functions for the development bank and the private bank obtained (section 5.1). This overview addresses the methodological issues of output definition, functional form, and estimation procedure in terms of their effects on the overall statistical performance of the cost-function estimates. This section also provides estimates of the average and marginal costs of lending, as well as the average and marginal costs of mobilizing deposits, for the two banks under study. Second, the results obtained for the measures of economies of scale, economies of scope, and factor substitution are presented and contrasted between the two banks (section 5.2). Section 5.3 deals with the effects of interest-rate regulations on intermediation costs in the two institutions. Then, section 5.4 presents and discusses the effects of loan-targeting and special credit projects on portfolio composition and costs of the development bank. The results obtained for the effects of delinquency on this bank's costs are reviewed and discussed in the final section.

### 5.1. Overview of Results

This section is organized in three parts: the first and second parts report the separate results of the costfunction estimates of each bank. The methodological issues and problems associated with the estimation are also discussed. Estimates of average costs and marginal costs are obtained using the functional specifications and estimation procedures that are judged best among the different alternatives considered. The results for the two banks are summarized and contrasted in the final part of this section, emphasizing their economic interpretation and policy implications.

### 5.1.1. The Cost Function of the Development Bank

The results obtained from the estimation of the bank's cost function, with different output definitions and two functional forms, are summarized in table 3. Detailed results of the estimated equations referred to in table 3 are reported in appendix C, tables 33 and 34. These results correspond to the (unrestricted) single-equation estimation of different model specifications. $\frac{1}{}$  The issue of system estimation versus single-equation estimation will be considered later.

Three general results are clear from the observation of table 3. First, the output definition that considers loans 1/ Factor-price homogeneity restrictions were not significant when imposed on any of the equations.

and deposits as separate outputs performs better in terms of statistical results than the equations including the singleoutput definition (loans plus deposits). Comparison of rows (1) and (3) in each column of table 3 indicates that the R-square values for the equations with the two-output definition are consistently higher than those obtained with the single-output (aggregate) specification. Second, output defined in value terms (value of loans, deposit balances) provides a better statistical fit than estimations using number of loans and number of deposit accounts. This is evident from the comparison of columns (1) and (2) against columns (3) and (4) in table 3. All R-square values reported in the two right-hand columns of this table (value of loans, deposit balances) are higher than those shown in the first two columns (number of loans, number of deposit accounts), regardless of the functional form utilized in the estimation of the cost function. Third, the translog specification performs statistically better than the Cobb-Douglas form, under all different output definitions. This result is clear from the comparisons of column (1) versus column (2), and column (3) against column (4) in table 3. The R-square values associated with the translog form are significantly higher than those obtained with the Cobb-Douglas specification. The F-ratios reported in table 3 indicate that these differences are statistically significant at the 1% level in all cases.

# Table 3. Development Bank: Summary of R-square Values from the Estimation of the Cost Function, Under Different Output Definitions and Functional Forms

· •

	(	Output Definition:		Number of Loans and Deposit Accounts <u>versus</u> Value of Loans and Deposit Balances			
	1	Functional Form:	Cobb-Doug	glas <u>versus</u> Tra	nslog		
Output Definition: Two-Output versus		(1) Number of L Number of Deposi	(2) Joans, It Accounts	(3) Value of L Deposit Ba	(4) oans, lances		
Single-Outpu	t	Cobb-Douglas	Translog	Cobb-Douglas	Translog		
Two-output:	loans, q <sub>l</sub> , and deposits, q <sub>2</sub>						
(1)	R <sup>2</sup>	0.7567	0.7922	0.8208	0.8598		
(2)	F-test of functional for	<u>ma</u> /	2.47*		7.60*		
Single-outpu	t: loans plus deposits $(Q = q_1 + q_2)$						
(3)	R <sup>2</sup>	0.7325	0.7650	0.7946	0.8167		
(4)	F-test of functional for	m	3.45*		5.08*		

Source: Tables 33 and 34 in appendix C.

a/	' E	? =	[SSEC	-	SSEmp)/(k <sub>r</sub>	-	$k_{C}$ )]/(SSE <sub>m</sub> /N- $k_{m}$ ),	where	SSE	=	error sum of squares,
-			•						k	=	number of estimated parameters,
									С	de	enotes Cobb Douglas form,
									т	de	enotes Translog form.
*	:	si	gnifica	ant	t at 0.01	le	vel.				. –

.

Overall, the "best" specification for the cost function of the development bank is the translog form that includes output as two separate variables expressed in value terms, value of loans and deposit balances (column 4 in table 3, and table 33 in appendix C). The remainder of this section will concentrate on the results obtained with the two-output specifications, both in terms of numbers of accounts and in value terms. The implications of using the sum of loans <u>plus</u> deposits as a measure of aggregate output will be discussed in section 5.2, in reference to the value of the scale-economies parameter.

### Single-Equation and System Estimation

Tables 4 and 5 present the results obtained for the cost function using single-equation estimation (OLS) and system estimation (GLS). Table 4 reports the results for the cost function using numbers of loans and depositaccounts as bank outputs, whereas table 5 deals with outputs defined as value of loans and deposit balances. It is clear in both tables that system estimation improves overall goodness-of-fit of the estimations, and specially the statistical significance of individual coefficients. Significance of individual coefficients appears weak in the case of single-equation estimation due to multicollinearity. The additional degrees of freedom gained through the adoption of the system approach provide a solution for this low level of statistical significance.

		(1) Single Equation (OLS)		(2) System of Equations (GLS)		
Para	ameter (Variable)	Estimate	t-ratio	Estimate	t-ratio (asymptotic)	
α0	(intercept)	9.3081	2.382*	9.0209	4.005*	
α1	(lnq <sub>1</sub> , no. of loans)	-0.6827	-1.222	0.0274	0.080	
α2	(lnq <sub>2</sub> , no. of deposit	-0.6293	-0.486	-1.5806	-2.238†	
<sup>β</sup> 1	accounts) (lnp <sub>1</sub> , price of labor)	0.6978	1.077	0.8020	8.073*	
β <b>2</b>	(lnp <sub>2</sub> , price of capital)	0.3021	0.466	0.1980	1.994†	
γ11	(1nq <sub>1</sub> ) <sup>2</sup>	-0.0931	-1.748°	0.0620	1.845°	
Υ <u>22</u>	(lnq <sub>2</sub> ) <sup>2</sup>	-0.0700	<del>-</del> 0.255	0.3796	2.547*	
γ <sub>12</sub>	(1nq <sub>1</sub> 1nq <sub>2</sub> )	0.1973	1.618°	0.0365	0.481	
δ <sub>11</sub>	(1np <sub>1</sub> ) <sup>2</sup>	-0.0111	-0.149	0.1007	13.686*	
δ <mark>22</mark>	(lnp <sub>2</sub> ) <sup>2</sup>	-0.0111	-0.149	0.1007	13.686*	
<sup>δ</sup> 12	(lnp <sub>1</sub> lnp <sub>2</sub> )	0.0111	0.149	-0.1007	-13.686*	
η <b>11</b>	(lnq <sub>l</sub> lnp <sub>l</sub> )	0.0257	0.561	-0.0604	-7.482*	
η <b>12</b>	(lnq <sub>1</sub> lnp <sub>2</sub> )	-0.0257	-0.561	0.0604	7.482*	
η <b>21</b>	(lnq <sub>2</sub> lnp <sub>1</sub> )	0.0499	0.366	-0.0730	-4.004*	
η <sub>22</sub>	(lnq <sub>2</sub> lnp <sub>2</sub> )	-0.0499	-0.366	0.0730	4.004*	
$R^2$		0.7749		0•7898 <u>b</u> ,	/	
F-va	alue	57.37*		38.64*		
Weighted R <sup>2</sup> (system)				0.4238		

Table 4. Development Bank: Estimated Parameters of the Cost Function. Output Defined as Number of Loans  $(q_1)$  and Number of Deposit Accounts  $(q_2)$ . Single Equation versus System Estimation.<sup>a</sup>/

<u>a</u>/ Factor-price homogeneity restrictions imposed on all estimated equations. Cross-equations restrictions imposed on system estimation. N=160. DFS=306. Significance levels: \*, .01; †, .05; °, .10.
 <u>b</u>/ R<sup>2</sup> of labor-share equation: 0.2775, F-ratio = 14.79.

	( Single (0	1) Equation LS)	(2) System of Equations (GLS)		
Parameter (Variable)	Estimate	t-ratio	Estimate	t-ratio (asymptotic)	
α <sub>0</sub> (intercept)	5.3210	9.817*	4.7545	14.219*	
$\alpha_1$ (lnq <sub>1</sub> , value of loans)	0.1439	1.153	0.3434	4.890*	
$\alpha_2$ (lnq <sub>2</sub> , deposit balances)	0.0163	0.101	-0.1037	-1.188	
$\beta_1$ (lnp <sub>1</sub> , price of labor)	0.5439	3.399*	0.5776	14.385*	
$\beta_2$ (lnp <sub>2</sub> , price of capital)	0.4561	2.851*	0.4224	10.518*	
$\gamma_{11} (lnq_1)^2$	0.0931	3.388*	0.1351	10.502*	
$\gamma_{22} (lnq_2)^2$	-0.0429	-1,211	0.0967	4.756*	
$\gamma_{12}$ (lnq <sub>1</sub> lnq <sub>2</sub> )	-0.0063	-0.257	-0.0113	-0.723	
$\delta_{11} \; (lnp_1)^2$	0.0003	0.007	0.1022	13.729*	
$\delta_{22} (lnp_2)^2$	0.0003	0.007	0.1022	13.729*	
$\delta_{12}$ (lnp <sub>1</sub> lnp <sub>2</sub> )	-0.0003	-0.007	-0.1022	-13.729*	
$\eta_{11}$ (lnq <sub>1</sub> lnp <sub>1</sub> )	-0.0426	-1.255	-0.0954	-14.914*	
$\eta_{12}$ (lnq <sub>1</sub> lnp <sub>2</sub> )	0.0426	1.255	0.0954	14.914*	
$\eta_{21}$ (lnq <sub>2</sub> lnp <sub>1</sub> )	0.0922	2.786*	-0.0169	-2.029†	
η <sub>22</sub> (lnq <sub>2</sub> lnp <sub>2</sub> )	-0.0922	-2.786*	0.0169	2.029†	
R <sup>2</sup>	0.8491		0.8586 <u>b</u> /	/	
F-value	173.86*		117.51*		
Weighted R <sup>2</sup> (system)			0.7331		

Table 5. Development Bank: Estimated Parameters of the Cost Function. Output Defined as Value of Loans  $(q_1)$  and Deposit Balances  $(q_2)$ . Single Equation <u>versus</u> System Estimation.<u>a</u>/

 a/ Factor-price homogeneity restrictions imposed on all estimated equations. Cross-equations restrictions imposed on system estimation. N=288. DFS=562. Significance levels: \*, .01; †, .05.
 b/ R<sup>2</sup> of labor-share equation: 0.2886, F-ratio = 28.50.

The cost system that specifies loans and deposits in value terms (table 5) again presents a better fit (weighted R-square) than the cost-system with output defined in terms of number of loans and deposit accounts (table 4). In both cases the R-square obtained for the labor-share equation is rather low, a result that affects the overall weighted Rsquare obtained for the system. However, the gains in efficiency of the estimation attained through the use of a system approach are clear. Almost all individual coefficients are statistically significant and in general their magnitudes differ from those obtained under the singleequation approach. As a consequence, scale-economy measures and other related results obtained using the estimated parameters of the cost-system will differ from those derived from single-equation results. These differences will be discussed later in section 5.2. Before considering the expressions for the marginal costs of the two outputs, it is useful to take into account the effects of loan size and deposit size on the estimation of the cost function.

### Effects of Loan Size and Deposit Size

The results of the cost-system estimations including loan-size and deposit-size effects are presented in table  $6.2^{/}$  In this table, model 1 describes the estimated coefficients of the function that utilizes the number of loans  $2^{/}$  Results of the single-equation estimation are included for reference in table 35, appendix C.

	Model (output definition)							
	Number of Dep	(1) of Loans (q <sub>1</sub> ), posit Accounts (q <sub>2</sub> )	(2) Value of Loans (q,), Deposit Balances (q <sub>2</sub> )					
Parameter (Variable)	Estimate	t-ratio (asymptotic)	Estimate	t-ratio (asymptotic)				
α <sub>0</sub> (intercept)	1.6131	0.941	6.0005	11.489*				
$\alpha_1$ (Inq <sub>1</sub> , Ioans)	-0.3023	-1.117	0.5814	5.787*				
$\alpha_2$ (Inq <sub>2</sub> , deposits)	0-4123	0•769	-0.6449	-3.203*				
$\beta_1$ (inp <sub>1</sub> , price of labor)	0.6710	6•793	0.5585	10.192*				
$\beta_2$ (inp <sub>2</sub> , price of capital)	0.3290	3.331*	0.4415	8.056*				
$\gamma_{11} (lnq_1)^2$	0.1116	<b>4</b> •046*	0.1463	8.835*				
$\gamma_{22} (lnq_2)^2$	0.0027	0.023	0.2619	4.799*				
$\gamma_{12}$ (Inq <sub>1</sub> Inq <sub>2</sub> )	-0.0226	-0.365	-0.0646	<del>-</del> 2.693*				
$\delta_{11} (lnp_1)^2$	0.0479	5•564*	0.0756	8.041*				
$\delta_{22} (Inp_2)^2$	0.0479	5•564*	0.0756	8.041*				
$\delta_{12}$ (Inp <sub>1</sub> Inp <sub>2</sub> )	-0.0479	<del>~</del> 5•564 <b>*</b>	<del>-</del> 0•0756	-8.041*				
η <sub>li</sub> (Inq <sub>l</sub> Inp <sub>l</sub> )	-0.0445	-5.513*	-0.0766	-9.542*				
$\eta_{12}$ (lnq <sub>1</sub> lnp <sub>2</sub> )	0.0445	5.513*	0.0766	9.542*				
$\eta_{21}$ (lnq <sub>2</sub> lnp <sub>1</sub> )	-0.0356	-1.932†	-0.0103	-0.858				
$\eta_{22}$ (lnq <sub>2</sub> lnp <sub>2</sub> )	0.0356	1.932†	0.0103	0.858				
θ <sub>1</sub> (Inq <sub>1</sub> InLS, ioan-size interaction)	0.0858	19•505*	-0.0091	-2.332†				
$\theta_2$ (Inq <sub>2</sub> InDS, deposit-size	0.0527	9•786*	-0.0180	-1.950†				
R <sup>2</sup> Interaction)	0•8955 <u>b/</u>		0.8858 <mark>b/</mark>					
F-value	76.03		68.86					
Weighted R <sup>2</sup> (system)	0.7848		0.8168					

Table 6. Development Bank: Estimated Parameters of the Cost Function, Including Loan-Size and Deposit-Size Effects. System Estimation with Two Output Definitions

a/ Factor-price homogeneity and cross-equations restrictions imposed on estimation. DFS=304. Significance levels: \*, .01; †, .05.

•

<u>b/</u>  $R^2$  of labor-share equation: Model (1) = 0.2775 (F = 14.79) Model (2) = 0.3116 (F = 17.42)

-

and the number of deposit-accounts as the measure of the bank's outputs. The results that correspond to the definition of outputs in value terms (value of loans, deposit balances) are presented as model 2 in table 6. Both models show satisfactory results from a statistical point of view. In general, model 2 (with output in value terms) performs better than model 1, considering the values of the weighted R-square and the statistical significance of individual coefficients, therefore this model will be preferred later for prediction purposes. The most interesting contrast between the two models corresponds precisely to the estimated coefficients of the loan-size and deposit-size interactions. Both coefficients show positive signs in the function that specifies outputs in terms of numbers of loans and deposit-accounts, whereas these signs are negative when the cost function is specified using the value of loans and deposit balances as output definition. This contrast can be understood more clearly recalling the expressions for the marginal costs of each output, previously developed in section 4.1.

Marginal costs of lending:

 $MC_{1} = (C_{1}/q_{1})(B_{1} + \gamma_{11}\ln q_{1} + \gamma_{12}\ln q_{2} + \theta_{1}\ln LS) , (5.1)$ marginal costs of mobilizing deposits:  $MC_{2} = (C_{2}/q_{2})(B_{2} + \gamma_{22}\ln q_{1} + \gamma_{12}\ln q_{1} + \theta_{2}\ln DS) , (5.2)$ 

where, MC, is the marginal cost of the ith output, i=1,2,

C<sub>1</sub>/q<sub>1</sub> is the point-estimate of the average cost of i the ith output, at given levels of q<sub>1</sub> and all other variables in the model (see section 4.1.1),

$$B_{i} = \alpha_{i} + \Sigma_{j} \eta_{ij} \ln p_{j} , j=1,2$$

- q1: loans, q2: deposits,
- p1: price of labor, p2: price of capital,

LS : loan size, DS : deposit size.

Consider first the marginal cost of lending. Results of model 1 in table 6 show the following signs for the coefficients of the variable affecting the marginal cost per loan:  $\hat{\gamma}_{11} > 0$ ,  $\hat{\gamma}_{12} < 0$ ,  $\hat{\theta}_1 > 0$ . These signs indicate that the marginal cost per loan is an increasing function of the total number of loans  $(\hat{\gamma}_{11} > 0)$ , is reduced by increases in the total number of deposit accounts  $(\hat{\gamma}_{12} < 0), \frac{3}{2}$  and increases with increases in loan size ( $\hat{\gamma}_1$  > 0). In short, the marginal cost of a loan is an increasing function of loan size. Now, the signs of the parameter-estimates of model 2 reported in table 6, associated with the marginal costs of lending per lempira lent are:  $\hat{\gamma}_{11} > 0$ ,  $\hat{\gamma}_{12} < 0$ ,  $\hat{\theta}_1$  < 0. Therefore, the marginal cost of lending per lempira <u>lent</u> increases with the total amount lent  $(\hat{\gamma}_{11} > 0)$ , benefits from economies of joint production ( $\hat{\gamma}_{12}$  < 0), and decreases when the loan-size increases  $\hat{\theta}_1 < 0$ ), i.e.,

<sup>3/</sup> This coefficient is not significantly different from zero, i.e., the estimated economies of joint production are not too strong when output is defined as numbers of loans and deposit-accounts.

marginal costs of lending per lempira lent are a decreasing function of the loan size.

The results obtained for the costs of mobilizing deposits are qualitatively very similar to those discussed above for the marginal cost of lending. The marginal cost of mobilizing deposits per account (coefficients from model 1 in table 6) tends to increase with the number of accounts  $(\hat{\gamma}_{22} > 0)$ , decreases with increases in the number of loans being jointly produced ( $\hat{\gamma}_{12}$  < 0), and increases as the deposit-size increases ( $\hat{\theta}_2 > 0$ ). It has to be pointed out that the estimates for both the scale effect  $(\hat{\gamma}_{22})$  and the effect of joint production  $(\hat{\gamma}_{12})$  are not significantly different from zero. These results suggest that the main determinant of the marginal cost of deposits per account is the average deposit-size. However, in the case of the marginal cost of mobilizing deposits per lempira mobilized (model 2 in table 6) all relevant coefficients are statistically significant. These estimated parameters indicate a clear cost-increasing effect of scale ( $\hat{\gamma}_{22}$  > 0), a significant cost-saving effect due to joint production of loan services ( $\hat{\gamma}_{12}$  < 0), and a cost-decreasing effect of deposit size  $(\hat{\theta}_2 < 0)$ .

Summarizing, marginal costs of lending are positively associated with the total volume of lending operations, and benefit from the joint production of deposit services. The marginal cost <u>per loan</u> increases with increases in the loan size, whereas the marginal cost <u>per lempira lent</u> decreases as the loan-size increases. Marginal costs of mobilizing deposits are an increasing function of total deposit activity, and enjoy cost-economies due to joint production of loans. The marginal cost <u>per deposit-account</u> increases as the average deposit-balance increases, whereas the marginal cost <u>per lempira mobilized</u> decreases as the average deposit-size increases.

Point estimation of average costs and marginal costs on a <u>per-lempira</u> basis utilizes the results of model 2 in table 6. This point estimation is obtained evaluating model 2 at the geometric means of all variables involved. $\frac{4}{}$  The resulting estimates are the following: $\frac{5}{}$ 

Share of lending costs in total costs : 71.1% Share of deposit-mobilization costs in total costs: 28.9%

#### Costs of lending

Average cost per lempira lent: L.O.1002, or 10.02% Marginal cost per lempira lent: L.O.0764, or 7.64% Costs of mobilizing deposits

Average cost per lempira mobilized L.0.0878, or 8.78% Marginal cost per lempira mobilized: L.0.0272, or 2.72%

Total average costs of intermediation for the bank, per lempira

L.O.1880, or 18.80%

- 4/ The procedure to allocate total estimated costs among loans and deposits was described in section 4.1.
- 5/ Figures in lempiras are lempiras of 1966. 1 Lempira = 0.5 US\$.

### Total marginal costs of intermediation for the bank, per lempira

L.O.1036, or 10.36%

The estimated costs of lending reported above are consistent with those reported by Cuevas and Graham [26] in a cross-sectional study of this development bank undertaken in 1982. This study found average costs of lending of 8.4%, using a sample with larger average amounts of funds lent and mobilized, and a loan size almost twice as large as the sample means used here to evaluate the results of the cost function. Other estimates of loan-administration costs in agricultural lending institutions range between 3 and 14 percentage points [71,79,96]. The results presented here confirm the existence of significant costs associated with lending in development banks. These costs are well above the administrative margins allowed for in special credit projects usually channeled through these financial institutions. Projects sponsored by international donors usually contemplate a 4%-margin for loan-administration, whereas the average spread of central-bank rediscount lines lies between 4% and 7% (see chapter 2). Therefore, the development bank will assume operational losses when engaging in credit programs of this nature.

There are no comparable estimates for the costs of mobilizing deposits in agricultural development banks in the available literature. The results reported here indicate that these costs are far from being negligible. However,

the relationship between the estimated values of average cost and marginal cost of mobilizing deposits indicates that there are important unexploited economies of scale involved in deposit-mobilization for this bank. These economies of expansion are also observed for lending activities, but in this case they appear less important than in the case of deposit-mobilization. This contrast in the effect of output expansion will be discussed at greater length in section 5.2.

### 5.1.2. The Cost Function of the Private Bank

The lack of data on number of loans limited the estimation of the cost function in the case of the private bank to output specifications in value terms (value of loans, deposit balances). At the same time, the loan-size variable could not be computed thus limiting the analysis of size effects to those associated with the average size of deposit balances.

As was the case in the development bank, the translog form is in general a better specification for the private bank's cost function than the Cobb-Douglas form. However, the differences are not as striking as those found in the development-bank case. Table 7 summarizes the R-square values obtained in different specifications, and the corresponding tests contrasting the two functional forms. Detailed results of these equations are presented in table 36, appendix C. The R-square figures reported in table 7

	Model (Functio	nal form)
	$\frac{100001}{(1)}$	(2)
Deposit-size Interaction	Cobb-Douglas	Translog
(l) No Interaction in the Specification		
R-square	0.9699	0.9758
F-test of functional form <sup>a/</sup>		4.11*
(2) Interaction Included in the Specification		
R-square	0.9701	0.9769
F-test of functional form		5.09*
t-test of deposit-size interaction	-1.204	<del>-</del> 2.856*
Source: table 36, appendix C.		<del></del>
<u>a</u> / F = [(SSE <sub>C</sub> - SSE <sub>T</sub> )/10]/(SSE <sub>T</sub> /N-1 where SSE = C de T de	4), error sum of so enotes Cobb-Doug enotes Translog	luares, las form, form.

Table 7. Private Bank: Summary of R-Square Values Obtained in the Estimation of the Cost Function, Under Different Specifications

\* : significant at 0.01 level

.

indicate excellent goodness of fit in all specifications. The translog form shows a significantly better fit than the Cobb-Douglas, even though the actual magnitude of the difference in R-squares is very small. The variable that captures the effect of deposit size does not show a coefficient significantly different from zero in the Cobb-Douglas form. It is statistically significant under the translog specification.

The gains in efficiency due to the adoption of a system-estimation procedure for the translog functional form were not very important in the private bank. As can be seen in table 8, all individual coefficients maintained their signs and approximate magnitude when comparing the results of the single equation estimation (model 1) against the parameter estimates obtained with the cost-system approach (model 2). All but one of the t-ratios for individual coefficients showed a moderate increase in absolute value in the cost-system estimation as compared to the single-equation results. However, the coefficients of only two explanatory variables went from below significant to a significant level  $(\eta_{21} \text{ and } \eta_{22})$ . This is in sharp contrast to the effects of the estimation procedure on the significance of individual coefficients found in the development bank, where most individual coefficients showed drastic changes in their levels of statistical significance as a result of the adoption of the cost-system approach.

		() Single H (OI	l) Equation LS)	(2) System of Equations (GLS)		
Para	ameter (Variable)	Estimate	t-ratio	Estimate	t-ratio (asymptotic)	
α0	(intercept)	-4.0332	-1.665°	-4.0121	-1.728°	
α1	(lnq <sub>1</sub> , loans)	-0.7159	-2.029†	-0.8772	-2.595*	
α2	(lnq <sub>2</sub> , deposits)	1.4888	2.938*	1.5261	3.139*	
β1	(lnp <sub>1</sub> , price of labor)	1.5403	2.780*	1.6593	3.126*	
β <sub>2</sub>	(lnp2, price of capital)	-0.5403	-0.975	<del>-</del> 0.6593	-1.242	
γ11	(lnq <sub>1</sub> ) <sup>2</sup>	0.0097	0.236	0.0160	0.404	
γ <sub>22</sub>	(lnq <sub>2</sub> ) <sup>2</sup>	-0.0584	-0.785	-0.0232	-0.325	
γ <sub>12</sub>	(lnq <sub>1</sub> lnq <sub>2</sub> )	0.1154	1.262	0.0928	1.060	
δ <sub>11</sub>	(lnp <sub>1</sub> ) <sup>2</sup>	-0.0481	-0.633	-0.0572	-0.785	
δ <sub>22</sub>	(lnp <sub>2</sub> ) <sup>2</sup>	-0.0481	-0.633	-0.0572	-0.785	
δ <sub>12</sub>	(lnp <sub>1</sub> lnp <sub>2</sub> )	0.0481	0.633	0.0572	0.785	
η <sub>11</sub>	(lnq <sub>1</sub> lnp <sub>1</sub> )	0.0323	0.697	0.0596	1.343	
η <b>12</b>	(lnq <sub>1</sub> lnp <sub>2</sub> )	-0.0323	-0.697	<del>-</del> 0.0596	-1.343	
<sup>ŋ</sup> 21	(lnq <sub>2</sub> lnp <sub>1</sub> )	-0.0825	-1.331	-0.1146	-1.928†	
η <sub>22</sub>	(lnq <sub>2</sub> lnp <sub>2</sub> )	0.0825	1.331	0.1146	1.928†	
$R^2$		0.9734		0.9758 <u>b</u> /		
F−va	alue	733.04*		503.31*		
Weig	ghted R <sup>2</sup> (system)			0.9536		

Table 8. Private Bank: Estimated Parameters of the Cost Function, Single Equation versus System Estimation.<sup>a/</sup>

a/ Factor-price homogeneity restrictions imposed on estimation. N=190. DFS=365. Significance levels: \*, .01; †, .05; °, .10. b/  $R^2$  of labor-share equation: 0.3590, F-ratio = 25.90.

.

In summary, the statistical improvement observed in the estimation of the cost function for the development bank, from Cobb-Douglas to translog form, then from singleequation to system estimation is also present in the estimation of the private bank's cost function. In the latter case however, the degree of improvement is less important than in the former, and more importantly, very few individual coefficients show statistical significance. This may be a consequence of widespread multicollinearity between the variables in the cost function. In fact, the correlation coefficients between several of these variables were extremely high (see table 37, appendix C).

This result implies that, in the case of the private bank, all independent variables in the cost function tend to "move together" over time and across branches. Increases in lending are accompanied by proportional movements in deposit-mobilization, and branches with large loanportfolios also carry large deposit-portfolios. In addition, it is likely that in the private bank salaries and wages vary proportionally to overall activity. Also, with all variables in real terms, variations over time tend to be reduced, that makes possible that essentially non-linear relationships (e.g., between a variable and its square) approach linear relationships, thus creating the multicollinearity problem. In contrast, the development bank will typically experience unbalanced expansions in outputs.

Given its reliance on external sources of funds, lending can be expanded over time, or across branches following regionally-targeted programs, without proportional changes in deposit mobilization. Also, these changes will tend to be more abrupt in the development bank, thus even the deflated values of the variables will show more variation than in the private bank. These characteristics of the development bank's operations tend to break collinearity patterns in the data, and facilitate the obtention of statistically significant coefficients for individual variables, using the appropriate estimation procedures.

With multicollinearity in the data, sample variances of individual coefficients will tend to be very large, therefore t-ratios will be in general small and will thus appear not significantly different from zero [43,70]. On the other hand, the high values of the R-square and the F-ratio for the overall function reject the null hypothesis that all individual coefficients are zero. This typical pattern of multicollinearity - high R-square and overall significance, and poor significance of individual coefficients - implies that the set of explanatory variables explains very well the behavior of the dependent variable, but their individual effects cannot be distinguished. In the case of the private bank, the additional degrees of freedom obtained with the adoption of a system-estimation approach are not sufficient

to overcome this degree of multicollinearity. $6^{-/}$  This characteristic of the data poses a problem whenever the interpretation of results relies upon values of estimated coefficients with low levels of significance. Therefore, the discussion that follows should be taken with a word of caution.

As pointed out before, the inclusion of the depositsize effect in interactive form with total deposit balances gave significant estimates in the translog specification. Table 9 shows the results of both single-equation and system estimation of the cost function including depositsize interactions. In both cases the sign of the estimated parameter is negative, meaning that intermediation costs in general will decrease, <u>ceteris paribus</u>, with increases in the average size of deposit balances. Specifically, marginal costs of mobilizing deposits per lempira mobilized are a decreasing function of deposit size. The expressions for the marginal costs of lending and of mobilizing deposits in the private bank can be written as follows:

marginal cost of lending:

$$MC_{1} = (C_{1}/q_{1})(B_{1} + \gamma_{11}\ln q_{1} + \gamma_{12}\ln q_{2}) , \qquad (5.3)$$

<sup>6/</sup> Dropping the two largest branches from the data set did not improve the significance of individual coefficients. Simplifying the model to the Cobb-Douglas form reduces but does not eliminate the collinearity problem, and has the disadvantage of restricting the parallel analysis of the two banks.

		( Single (0	l) Equation LS)	(2) System of Equations (GLS)	
Para	ameter (Variable)	Estimate	t-ratio	Estimate	t-ratio (asymptotic)
α0	(intercept)	-3.1379	-1.307	-3.0242	-1.324
α1	(lnq <sub>1</sub> , loans)	-0.5526	-1.572	-0.7300	-2.184†
α2	(lnq <sub>2</sub> , deposits)	1.3148	2.621*	1.3405	2.805*
β <sub>1</sub>	(lnp <sub>1</sub> , price of labor)	1.4563	2.673*	1.5913	3.071*
β <b>2</b>	(lnp <sub>2</sub> , price of capital)	-0.4563	-0.837	-0.5913	-1.141
γ <sub>11</sub>	(lnq <sub>1</sub> ) <sup>2</sup>	0.0040	0.098	0.0110	0.284
γ <sub>22</sub>	(lnq <sub>2</sub> ) <sup>2</sup>	-0.0567	-0.777	-0.0142	-0.205
γ <sub>12</sub>	(1nq <sub>1</sub> 1nq <sub>2</sub> )	0.1384	1.535	0.1134	1.323
δ <sub>11</sub>	(lnp <sub>1</sub> ) <sup>2</sup>	-0.0404	-0.542	-0.0508	-0.715
δ <sub>22</sub>	(lnp <sub>2</sub> ) <sup>2</sup>	-0.0404	-0.542	-0.0508	-0.715
δ <sub>12</sub>	(lnp <sub>1</sub> lnp <sub>2</sub> )	0.0404	0.542	0.0508	0.715
η11	(lnq <sub>1</sub> lnp <sub>1</sub> )	0.0064	0.138	0.0366	0.829
η <sub>12</sub>	(lnq <sub>1</sub> lnp <sub>2</sub> )	-0.0064	-0.138	-0.0366	-0.829
<sup>ח</sup> 21	(lnq <sub>2</sub> lnp <sub>1</sub> )	-0.0538	-0.872	-0.0894	-1.522
η <sub>22</sub>	(lnq <sub>2</sub> lnp <sub>2</sub> )	0.0538	0.872	0.0894	1.522
θ2	(lnq <sub>2</sub> lnDS, deposit-size	-0.0227	-2.768*	-0.0247	-3.169*
R <sup>2</sup>	interaction)	0.9745	هند فحد ظنه	0.9769 <u>b</u> /	
F-va	lue	684.92*		489.52*	
Weighted R <sup>2</sup> (system)				0.9562	

Table 9. Private Bank: Estimated Parameters of the Cost Function, Including Deposit-Size Effects. Single Equation versus System Estimation.<sup>a</sup>/

a/ Factor-price homogeneity restrictions imposed on estimation. N=190.
 DFS = 365. Significance levels: \*, .01; †, .05.
 b/ R<sup>2</sup> of labor-share equation: 0.3590, F-ratio = 25.90.

•

marginal cost of mobilizing deposits:

 $MC_{2} = (C_{2}/q_{2})(B_{2} + \gamma_{22}\ln q_{2} + \gamma_{12}\ln q_{1} + \theta_{2}\ln DS) , (5.4)$ where,  $MC_{i}$  is the marginal cost of the ith output, i=1,2,  $C_{i}/q_{i}$  is the point-estimate of the average cost of the ith output, at given levels of  $q_{i}$  and all other variables in the model (see section 4.1.1 for details),  $B_{i} = \alpha_{i} + \sum_{j} \eta_{ij} \ln p_{j} , j=1,2$  $q_{1}$ : loans,  $q_{2}$ : deposits,  $p_{1}$ : price of labor,  $p_{2}$ : price of capital, DS : deposit size.

The estimated coefficients reported in table 9 show the following signs:  $\hat{\gamma}_{11} > 0$ ,  $\hat{\gamma}_{22} < 0$ ,  $\hat{\gamma}_{12} > 0$ ,  $\hat{\theta}_2 < 0$ . However, with the exception of  $\hat{\theta}_2$  none of these coefficients is statistically different from zero at the commonly accepted levels of significance. Therefore, the individual effects of scale ( $\hat{\gamma}_{11}$ ,  $\hat{\gamma}_{22}$ ) and of joint production ( $\hat{\gamma}_{12}$ ) will not be given particular emphasis in this analysis. The interpretation of the effect of deposit size on the marginal cost of deposit-mobilization (i.e., the sign of  $\hat{\theta}_2$ ) has been pointed out above.

It is interesting to highlight here the relationships between the average-cost and marginal-cost levels for each of the bank's outputs. In general, marginal costs are greater than average costs when the firm is operating on the upward-sloping portion of its average cost curve. This means that the output level is beyond the overall minimumcost level, and that further output expansions will be
always cost-increasing. The reverse is true if marginal costs lie below average costs at a specific level of output. Marginal costs will be greater than (above) average costs or less than (below) average costs depending on the magnitude of the second term in parentheses on the right-hand side of equations (5.3) and (5.4). The evaluation of this term at the geometric means of the variables involved for the marginal cost of lending gives a value less than one, i.e., the marginal cost of lending is lower than the corresponding average costs. In other words, the bank is operating on the downward-sloping section of a hypothetical average cost curve for lending activities. On the other hand, the evaluation of the corresponding expression in equation (5.4) for the marginal cost of mobilizing deposits gives a value greater than one meaning that the marginal cost of mobilizing deposits at the geometric means of the variables in the model is greater than the corresponding average costs. This result indicates that the bank is operating on the upward sloping section of a hypothetical average cost curve for deposit-mobilization. Further discussion of this contrast between the bank's cost position with respect to each output is presented in section 5.2.

Evaluating the results of the cost-system estimation in table 9 (model 2) at the geometric means of all the variables in the model, the following point-estimates were obtained for the private bank:

Share of	E lending costs in total costs:	28.3%
Share of in total	E deposit-mobilization costs L costs:	71.7%

#### Costs of lending:

Average costs per lempira lent: L.0.0339, or 3.39% Marginal cost per lempira lent: L.0.0169, or 1.69% Costs of mobilizing deposits: Average cost per lempira mobilized: L.0.0533, or 5.33%

Marginal cost per lempira mobilized: L.O.0671, or 6.71%

Total average costs of intermediation for the bank, per lempira: L.0.0872, or 8.72%

# Total marginal costs of intermediation for the bank, per lempira: L.0.0840, or 8.40%

These results show several similarities with those reported for this bank in Cuevas and Graham [26]. The shares of lending and deposit activities in total costs are quite similar to those obtained in this cross-sectional study. Average costs of lending reported above are somewhat higher than those found in Cuevas and Graham, but this difference may be reflecting differences in average loan-size between the two samples that are not accounted for in the cost-function model estimated here.

Some general conclusions and several interesting contrasts emerge from the analysis of the results obtained for the development bank and the private bank. A review of these general conclusions and contrasts is presented below,

notwithstanding more thorough discussions of some of these findings in subsequent sections.

5.1.3. General Findings and Contrasts, A Summary

A summary of some of the results presented in previous sections is shown in table 10 to facilitate the analysis and discussion of these results. Rows 1 and 2 of table 10 indicate the distribution of total intermediation costs in each bank between lending costs (row 1), and costs of mobilizing deposits (row 2). Rows 3 and 4 show the average and marginal costs of lending on a per-lempira basis, while rows 5 and 6 report the corresponding average and marginal costs figures for the costs of deposit mobilization. Finally, overall intermediation costs (lending costs plus depositmobilization costs) are reported in rows 7 and 8 of table 10.

The first important contrast between the two banks is shown in rows 1 and 2 of table 10. Over 70% of the development-bank's costs of intermediation correspond to lending activities, whereas only 29% of its costs are attributed to the administration of deposit accounts. The opposite is true for the private bank, where only 28% of the costs are associated with lending, while 72% of the bank's total intermediation costs are related to deposit mobilization. This acute contrast reflects the development-bank's greater reliance on foreign funds and special rediscount lines from the central bank, as compared to the private bank

Cos	t Concept	Development Bank (%)	Private Bank (%)
1.	Share of Lending Costs in Total Intermediation Costs	71.1	28.3
2.	Share of Deposit-Mobilization Costs in Total Intermediation Costs	28.9	71.7
	Costs of Lending		
3.	Average Costs	10.02	3.39
4.	Marginal Costs	7.64	1.69
	Costs of Mobilizing Deposits		
5.	Average Costs	8.78	5.33
6.	Marginal Costs	2.72	6.71
	Overall Lender's Intermediation Costs <sup>a</sup> /		
7.	Average Costs	18.80	8.72
8.	Marginal Costs	10.36	8.40

Table 10. Lender's Intermediation Costs: Lending Costs and Costs of Mobilizing Deposits. Summary of Findings for the Development Bank and the Private Bank.

Source: Results of cost-system estimations, table 6 (model 2) and table 9 (model 2), evaluated at geometric means of the variable in the models.

<u>a</u>/ Lending Costs + Costs of Deposit Mobilization

which relies more heavily upon financial resources mobilized from the general public.

Over the period under analysis (1971-1982), an average of 51% of the loan-portfolio of the development bank was funded through foreign funds or central-bank rediscount lines. Furthermore, these external (non-deposit) sources of funds have grown in relative importance with respect to the loan-portfolio from a 44%-average in the period 1971-1974 to a 57%-average in the period 1979-1982. Consequently, the proportion of the total value of new loans funded through deposit mobilization decreased from an average of 56% in the period 1971-1974, to a 43%-average in the last four years of the series. On the other hand, the private bank has relied primarily upon deposits mobilized from the general public to finance its loan portfolio. This bank's access to rediscount lines at the central bank has been limited, and only recently has it engaged in foreign-funded special credit projects. In 1981, a representative year according to bank officials, 91% of the loan portfolio was funded with own deposits, almost 7% came from central-bank rediscount funds, and a little over 1% from foreign funds (primarily World Bank projects). This sharp contrast in the composition of the banks' liabilities has a counterpart in the allocation of real resources in each bank, that is reflected in the participation of lending and deposit activities in total intermediation costs.

Costs of lending show a second striking contrast between the two banks. The estimated average costs of lending are 10% for the development bank, three times as high as those estimated for the private bank (3.39%). The marginal costs of lending are 4.5 times larger in the development bank (7.6%) than in the private bank (1.7%). This is again reflecting the differences in the sources of funds with which the banks operate. The greater reliance on external funds by the development bank implies the acceptance of loan-targets imposed by foreign donors, international lenders, and/or the government. These targets typically imply servicing a more risky and numerous clientele, and a high incidence of relatively small loans. Also, targeted funds are accompanied by monitoring, supervision, and reporting requirements that force the institution to maintain a more centralized operation, and a heavier incidence of supervisory and record-keeping resources, than would be the case in the absence of these targeting requirements.

It is important to note that the cost estimates reported in table 10 do not include provisions for bad debts, thus representing a lower-bound estimate for the operational spread that these institutions would require in order not to suffer operational losses. In this sense, the results obtained for the development bank are particularly striking, especially when comparing these results with the

margins contemplated in credit projects funded by external agencies or the central bank. These funding sources usually allow only 3 to 4 percentage points to cover the administrative costs associated with the on-lending of their funds. Thus, to operate with these special lines of credit the development bank experiences an operational loss of over 6%, assuming that all loans are fully repaid.

The foregoing results highlight the existence of a policy inconsistency, in the sense that external donors and/or the government impose on the development bank costly loan-targets without appropriate support to service these target-groups. The costs of servicing a more risky, more numerous, and more costly clientele, for which the institution is reimbursed only at a margin of 3 or 4 percentage points, seriously compromise the financial viability of the institution. It is interesting to note that the usual 3-4% margin is closer to the average lending costs observed in efficient private commercial banks like the one under study here, than to the average lending costs observed in the development bank. However, as it has been documented elsewhere [26], the average cost of lending for the private bank increases substantially when dealing with foreignfunded credit projects. The average cost of agricultural loans made by the private bank with World Bank funds has been estimated at 8.4% ignoring default risks [26], a figure

that exceeds by far the 4%-spread allowed in these credit projects for loan-administration costs.

The average costs of mobilizing deposits are also higher in the development bank as compared to the private bank (row 5 in table 10). However, marginal costs of deposit-mobilization show the opposite pattern, they are lower in the development bank than in the private bank (row 6 in table 10). Note also that the private bank has gone beyond the minimum average cost level in its deposit activity, since the marginal costs of deposit-mobilization appears higher than the corresponding average cost. These differences in costs between the two banks are explained by the contrasts in their scale of operations, and to some extent by differences in the composition of their depositclientele. A brief discussion of these differences follows.

Taking the geometric means of the relevant variables in the two banks for the overall period 1971-1982, the ratio between the administrative costs of the private bank and those of the development bank is 1.4 to 1, the ratio of their loan portfolios is 2 to 1, and the ratio between their deposit portfolios is 6.8 to 1. In short, the private bank has extended its deposit-mobilization activity relatively more than its lending activity. Moreover, this bank has reached a point of decreasing returns to further expansions of the deposit-mobilization activity, unless this expansion

relies upon increasingly large average deposit-balances (see discussion on the effects of deposit-size on the bank's costs in section 5.1.2 above). At the other extreme, the development bank is operating on the steep-downward-sloping section of a hypothetical average-cost curve for depositmobilization, considering the large difference between average costs and marginal costs observed in table 10. Another factor that contributes to the low marginal costs of mobilizing deposits in the development bank is the incidence of deposits from public-sector institutions that, in general, should imply lower handling costs on a per-lempira basis. As indicated in chapter 2, the incidence of these deposits is larger in the liability portfolio of the development bank than in the private bank, due to regulations that force public institutions to deposit a majority of their cash-flow or surplus funds in the development bank. Yet, the main explanation for the behavior of depositmobilization costs in the development bank is overcapacity, since the large difference between average costs and marginal costs should be attributed primarily to under-utilized fixed or quasi-fixed resources in the structure of the bank.

Overall, intermediation costs are higher in the development bank than in the private bank (rows 7 and 8 in table 10). However, this difference is more important in terms of the total average costs of operation than it is in terms

of the marginal costs of intermediation. The relationship between the levels of average costs and marginal costs in the development bank reflects under-utilization of existing resources, whereas the private bank appears very close to its minimum-cost level of activity (marginal cost almost equals average cost). Marginal costs of intermediation in the development bank are only two percentage points higher than in the private bank, according to the estimates reported in table 10. This result indicates that the differences in efficiency are not too substantial between the two banks. However, an important implication is that marginal-cost pricing would imply large operational losses for the development bank, whereas in the case of the private bank it would represent an almost break-even situation. From a policy-making point of view, if operational margins were administered so that the development bank could cover its marginal costs of intermediation, this bank would still experience substantial losses, since its average costs exceed by far its marginal costs. Under such a policy however, the private bank would earn a profit since its average costs are lower than the marginal costs of the development bank.

# 5.2. Economies of Scale, Cost Complementarities, and Factor Substitution

#### Economies of Scale

The importance of output definition, functional form, and the procedure utilized to estimate the parameters of the cost function, in terms of the resulting scale-economies indicators, is discussed based on the results presented in tables 11 and 12. Table 11 summarizes the scale-economies estimates obtained for the development bank, under different output definitions, functional forms, and estimation procedures. Table 12 presents a similar breakdown of results for the private bank, although using only one output definition (value of loans, deposit balances).

As pointed out in the previous section, the best estimates of the development bank's cost function are obtained using the translog functional form and a system-estimation procedure. It is therefore pertinent to judge the different estimates of economies of scale against those obtained with this specification using the value of loans and deposit balances as output definition (panel 4, columns 4 and 5 in table 11). These are considered the best overall estimates of the development-bank's cost function.

Table 11 shows that the results of all single-equation estimations (columns 1 through 3) tend to underestimate the value of the scale-economies measure of the bank, as compared to the estimates obtained under a cost-system approach

	1	Estimation Procedure, Functional			
	Single	-Equation	(OLS)	System Es	timation (GLS)
Output Definition	Cobb-Douglas (1)	Translog (2)	Translog with Loan-Size and Deposit-Size Effects (3)	Translog (4)	Translog with Loan-Size and Deposit-Size Effects (5)
1. Single Output, Number of Loans + Numb	er of Deposit Accou	ints			
Economies of Scale (ES)	0.48	0.47	0.84	_	
F-test for Ho: ES=1	45.65*	41.70*	3.97*		
<b>R-square</b> of equation	0.73	0.77	0.85		
2. Single Output, Value of Loans + Depos	it Balances				
Economies of Scale (ES)	0.64	0.58	0.75	-	
F-test for Ho: ES=1	58 • 67 *	65.14*	12-17*		
R-square of equation	0.79	0.82	0.86		
3. Two-Output, Number of Loans, Number of	of Deposit Accounts				
Economies of Scale (ES)	0.61	0.53	0.80	1.34	1.11
F-test for Ho: ES=1	20.14*	21.55*	7.04*	17.28*	4.62†
<b>R-s</b> quare of equation	0.76	0.79	0.89	0.79 (0.42)	0.89 (0.78)
4. Two-Output, Value of Loans, Deposit H	<b>Salances</b>				
Economies of Scale (ES)	0.65	0.71	0.79	1.08	1.07
F-test for Ho: ES=1	47.30*	34.95*	7.83*	8.67*	2.88
R-square of equation	0.82	0.86	0.88	0.86 (0.73)	0.89 (0.82)

.

#### Table 11. Development Bank: Scale-Economies Estimates (ES), Under Different Output Definitions, Functional Forms, and Estimation Procedures. Tests of Null Hypothesis ES=1, and R-squares of Underlying Estimated Equations.

.

Source: Tables 4, 5, and 6 in the text, tables 33 and 34 in appendix C.

•

.

a/ Weighted R-squares of the system indicated in parenthesis.
F : Null hypothesis (ES=1) rejected, at .01 level of significance
t : Null hypothesis (ES=1) rejected, at .05 level of significance

(columns 4 and 5). Among the estimated values obtained with a single-equation procedure, the models that do not include loan-size and deposit-size effects (columns 1 and 2) generate lower values of the economies-of-scale parameters than models incorporating these variables (column 3). There are no substantial differences in the resulting estimates of scale-economies due to "pure" differences in functional form (column 1 <u>versus</u> column 2), or across different output definitions. An exception is the case of the single-output definition in terms of number of loans plus deposit accounts (panel 1, columns 1 and 2), that generates very low estimates of scale economies.

In general, all single-equation estimations give values of economies of scale significantly less than one, whereas the most preferred estimates do not differ significantly from one (panel 4, column 5), or tend to be significantly greater than one (panel 4, column 4). Confidence intervals with 95% confidence limits calculated with these more reliable estimates give the following boundaries for the scale-economies (ES) parameter of the development bank:

1.027  $\leq$  ES  $\leq$  1.135, panel 4, column 4 in table 11, point-estimate: ES = 1.08. 0.989  $\leq$  ES  $\leq$  1.156, panel 4, column 5 in table 11, point-estimate: ES = 1.07.

In contrast with the foregoing results, the functional form appears to be very important in determining the

estimated level of scale-economies in the private bank (table 12). Values of this parameter estimated with a translog form are twice as large as those obtained with a Cobb-Douglas specification, with little or no effect attributable to the inclusion of the deposit-size variable. Despite these large variations in the results due to the choice of functional form, all but one of the estimated scale-economies measures shown in table 12 are not significantly different from one. This is due to the widespread multicollinearity existing in the data utilized to estimate the private bank's cost function, whose effects were discussed in the previous section. One of the consequences of this multicollinearity problem is the presence of large variances of the individual coefficients. The economies-ofscale estimate is a linear combination of these coefficients, therefore its associated standard error will tend to be large, since this is obtained as a combination of the variances of the individual coefficients. As a consequence, a 95% confidence interval for the scale-economies parameter of the private bank results in a very large interval  $(0.31 \leq ES \leq 2.88)$ . Therefore, these results obtained for the private bank should be considered with caution.

Summarizing, in the two banks the results obtained with the preferred estimation approaches generate estimates of economies of scale that are not significantly different from

		Functiona	1 Form	
	C	obb-Douglas wi Deposit-Size	th	Translog with Deposit-Size
	Cobb-Douglas	Effect	Translog	Effect
Estimation Procedure	(1)	(2)	(3)	(4)
1. Single-Equation				
Economies of Scale (ES)	0.95	0.96	1.73	1.86
F-test for Ho: ES=1	6.17*	3.85	1.12	1.61
R-square of equation	0.97	0.97	0.98	0.98
2. Cost-System				
Economies of Scale (ES)			1.59	1.76
F-test for Ho: ES=1			0.82	1.39
R-square of equation			0.98	0.98
(weighted R-square of sys	stem)		(0.95)	(0.96)

## Table 12. Private Bank: Scale-Economies Estimates (ES), Under Different Functional Forms, and Estimation Procedures. Tests of Null Hypothesis ES=1, and R-squares of Underlying Estimated Equations.

Source: Tables 8 and 9 in the text, table 36 in appendix C.

\* : significant at 0.01 level

one. The important contrast however, is that in the development bank the point estimate of economies of scale is close to one (1.07), whereas in the private bank this point estimate is considerably greater than one (1.59). In the former case, a 10%-increase in the production of both outputs (loans and deposits) will generate a 10.7%-increase in total administrative costs of the development bank. Τn the private bank, the same 10%-increase in both outputs will create an increase in total administrative costs of 15.9%. It is important to note however, that the values of scaleeconomies reported in tables 11 and 12 are not fixed. Instead, these values are a function of the scale of operations and of the output mix. In this respect, it is useful to recall equation (4.4) for the value of the economies of scale (ES):

 $ES = A + \gamma_{11} \ln q_1 + \gamma_{22} \ln q_2 + \gamma_{12} (\ln q_1 + \ln q_2)$ (5.5) where, A summarizes all parameters and variables in (4.4) that do not involve output quantities,

q<sub>1</sub>: loans, q<sub>2</sub>: deposits.

According to results presented in the previous section for the development bank (table 5), the scale-economies measure for this bank tends to increase (i.e., to turn into diseconomies) as output increases, since both  $\hat{\gamma}_{11}$  and  $\hat{\gamma}_{22}$ are positive. There is, however, an offsetting effect due to joint production of loans and deposits, since  $\hat{\gamma}_{12}$  is negative. A similar analysis cannot be done for the private bank, since none of the coefficients involved is significantly different from zero in this case (see tables 8 or 9 in the previous section). A more direct way of portraying the effects of scale and output-mix on the values of economies-of-scale measures is computing these values for branches of different sizes in each bank. This exercise is summarized in table 13, that shows the values of partial economies of scale (percentage change in costs with a 1% change in one output), and overall economies of scale (change in costs as a result of a change in the production of both outputs), for the two banks. Different (real-life) branch-size cases have been selected in both banks, in order to assess the main similarities and contrasts between banks and branch-sizes. Partial and overall economies of scale computed at the sample means of each bank are also reported in table 13.

A first contrast between the two banks is found in the values of the overall scale-economies indicators (rows 3 and 6 in table 13). The magnitudes of the estimated indicators for the two branch-sizes of the development bank denote the existence of a U-shaped overall average-cost "curve" (indeed a cost-surface), with the small branch lying on the downward sloping portion of this "curve", and the large branch located on the upward sloping section of this surface. The results obtained for the private bank however, indicate that

		L	evel of Evaluation	on	
Bank	, Economies-of-Scale Measure (ES)	Sample Mean (1)	"Small" Branch Case (2)	"Large" Branch Case (3)	
Deve	lopment Bank <sup>a</sup> /				
	Partial ES (ðlnC/ðlnq <sub>i</sub> )				
(1)	ES <sub>1</sub> , Loans	0.77	0.66	1.14	
(2)	ES <sub>2</sub> , Deposits	0.31	0.30	0.40	
(3)	Overall ES (Σ <sub>i</sub> ðlnC/ðlnq <sub>i</sub> )	1.08	0.96	1.54	
<u>Priv</u>	ate Bank <sup>b/</sup>				
	Partial ES (ðlnC/ðlnq <sub>i</sub> )				
(4)	ES <sub>1</sub> , Loans	0.39	0.22	0.67	
(5)	ES <sub>2</sub> , Deposits	1.20	1.05	1.42	
(6)	Overall ES ( <code>∑i</code> ðlnC/ðlnq <sub>i</sub> )	1.59	1.27	2.09	

Table 13. Estimated Values of Partial and Overall Economies of Scale (ES), at the Sample Means and at Different Branch-Sizes. Development Bank and Private Bank

<u>a</u>/ Computed from table 5 (model 2). Branch-size cases selected on the basis of loan activity, 1982.

b/ Computed from table 8 (model 2). Branch-size cases selected on the basis of loan and deposit activity, 1982. even the small branch would be experiencing "diseconomies" associated with overall output expansion. A representation of the private bank's average cost would be a surface with positive slopes for all movements that imply proportional expansions of both outputs.

The expression "overall expansion" needs to be underlined in the foregoing discussion since, as is evident from table 13, there exist substantial differences in the separate cost-effects of the expansion of different outputs. Furthermore, these differential effects of output expansions vary across banks. For the development bank, there exist substantial economies of scale to the expansion of depositmobilization activities. The partial scale-economies value of 0.31 computed at the sample mean (row 2, column 1 in table 13) indicates that a 10%-increase in deposit balances mobilized by the bank generates only a 3.1%-increase in administrative costs. On the other hand, the lending activities of this development bank are approaching constant returns-to-scale for the average-branch case  $\frac{7}{2}$ , and display diseconomies of scale in branches with large amounts of funds lent (row 1, column 3). The opposite pattern is observed for the private bank in table 13. This bank's lending is the activity that shows cost-advantages as compared to deposit mobilization. In all cases, an expansion in the private bank's lending of 10% would generate

<sup>7/</sup> i.e., a hypothetical branch that could be described by the sample means of all variables.

small-to-moderate increases in administrative costs, depending on the branch size (2.2% in the small branch, to 6.7% in the large branch). In contrast, the same expansion in the private bank's deposit-mobilization activity would create a cost increase between 10.5% (small branch) and 14.2% (large branch).

The general conclusion of the foregoing analysis is that the two banks could benefit from "economies of scale" by engaging in unbalanced output-expansion. In brief, each bank should expand relatively more the activity with the lowest value of the partial scale-economies measure, i.e., the output whose expansion generates the smallest cost increase. Expansion strategies for the development bank should emphasize deposit mobilization over lending activities. On the other hand, the private bank's expansion strategies should be biased towards lending operations. With deposit-mobilization expanding at a slower pace than lending, the overall (weighted) economies-of-scale indicator for this bank should approach unity.

## Cost Complementarities (Economies of Scope), and Elasticities of Factor Substitution and Factor Demand

Several parameters associated with the underlying technology of production are derived from the estimated parameters of the cost function. The relationships that allow the derivation of these results were discussed in section 4.1. Table 14 reports these results for the two banks under

study. The necessary condition for the existence of costcomplementarities indicated by Murray and White [69] is met in the two banks (row 1 in the table). In addition to the satisfaction of this necessary condition for cost complementarity, no further conclusions can be derived from the numerical values reported in table 14, row 1, since there are no specific units associated with these estimated parameters. Notwithstanding this limitation, these results indicate that, for the two banks, joint production of loans and deposits offers cost-advantages as compared to specialized single-output activity. This argues against the widespread "development" strategy of creating specialized lending institutions with no deposit-mobilization functions. Without necessarily altering their initial goals and objectives, these institutions would be better off in terms of costs and financial liability with the provision of multiple services to their clientele.

The elasticity of substitution between labor and capital,  $\sigma_{12}$ , computed using the estimated parameters of the cost function and reported in table 14 (row 2) appear relatively low for the development bank, even though there are no appropriate points of reference in the literature revised. The result obtained for the private bank ( $\hat{\sigma}_{12}$ =1.24) is almost twice as high as the value reported for the development bank ( $\hat{\sigma}_{12}$ =0.63). For comparison, the values reported by Murray and White [69] for British Columbia

Table 14.	Cost Complementarities in Production (Economies of Scope), Elasticities of Factor Substitution, and Price-Elasticities
	of Demand for Factors of Production, Derived from Cost-Function Estimates. Development Bank and Private Bank.

	Development	Private
Estimated Parameters	Bank <sup></sup>	Bank <sup>b</sup>
1. Cost Complementarities, $(\hat{\gamma}_{12} + \hat{\alpha}_1 \hat{\alpha}_2)$	-0.4393	-0.8651
2. Elasticity of Factor-Substitution between Labor and Capital, $\hat{\sigma}_{12}$	0.6328	1.2419
3. Price-Elasticities of Demand for Factors of Production, e <sub>ij</sub>		
$\hat{e}_{11}$ (demand for labor, price of labor)	-0.4493	-0.8693
$\hat{e}_{12}$ (demand for labor, price of capital)	0.4493	0.8693
ê <sub>22</sub> (demand for capital, price of capital)	-0.1835	-0.3726
$\hat{e}_{21}$ (demand for capital, price of labor)	0.1835	0.3726

a/ Computed from cost-system estimates, table 6, model 2.

b/ Computed from cost-system estimates, table 9, model 2.

credit unions ( $\hat{\sigma}_{12}$ =1.74) are even higher than the elasticity of substitution found here for the private bank. However, their results imply price-elasticities of demand for factors of production of a magnitude similar to those calculated here in the case of the private bank.

In general, the private bank shows a higher value of the elasticity of substitution between factors of production, and factor demands more price-elastic than the development bank. These results denote a greater flexibility of the private bank in the allocation of productive resources, and a larger response to factor-market signals, as compared to the development bank. As expected, in both cases the demand for labor services shows a higher price-elasticity than the demand for services of capital goods. Overall, the results presented in table 14 reflect a more rigid structure of operations in the development bank, and a less important role of market signals in this bank's resource allocation, as compared to the private bank. As will be discussed later at greater length, the management of a development bank operates in an environment subject to pressures and constraints that at times induces decisions independent of market considerations.

# 5.3. Interest-Rate Regulations and Lenders' Intermediation Costs

In this section, the effects of interest-rate regulations on the banks' administrative costs are analyzed, including in each bank's function the real deposit-rate and the real lending-rate as proxies for the interest-rate restrictions imposed on financial intermediaries. It is useful to recall here the expression that includes these two proxy-variables:

 $\lambda_1 (d - \dot{p}) + \lambda_2 (l - \dot{p})$  (5.6)

where, d is the nominal deposit-rate,

1 is the nominal ceiling on the lending rate,

p is the inflation rate.

As indicated before, the nominal deposit-rate is calculated in two alternative ways. First, the deposit rate is calculated as the arithmetic mean of all deposit-rate ceilings established by the central bank. This average ceiling is denoted as d<sub>c</sub>. Second, the implicit rate of interest paid on deposits by each bank is computed dividing total interest payments over total deposit balances. This implicit nominal deposit-rate is denoted dp and is considered a proxy for the institution's cost of funds under the existing conditions of regulation. In both cases, the real deposit-rate is obtained by substracting the rate of inflation p from the nominal deposit-rate. The real lending-rate ceiling is computed as the difference between the overall maximum ceiling on the nominal lending-rate, 1, and the inflation rate p. The rate of inflation is calculated as the 12-month variation in the country's implicit GDP deflator.

Selected statistics for the foregoing nominal and real rates are presented in appendix C (table 38), along with correlation coefficients for different pairs of these variables. An average nominal deposit-rate ceiling for the entire period 1971-1982 of 6.6% is associated with an average nominal implicit deposit-rate (cost of funds) of 3.6% and 3.7% respectively for the development bank and the private bank. The average nominal rates paid by these banks on interest-bearing accounts over the same period were 5.6% in the development bank, and 5.4% in the private bank. These rates are still below the average overall ceiling, reflecting the large proportion of the deposit portfolios accounted for by savings accounts in both banks. As it has been documented before (appendix A, table 30), savings accounts have been subject to lower interest-rate ceilings than time-deposits and other instruments. In real terms, the average deposit-rate ceiling in the period under study was -2.51%, while the average real implicit-rates paid on interest-bearing deposits were -2.52% in the development bank, and -2.71% in the private bank. The average real cost of funds for the two banks was -4.73% and -4.13% respectively. Finally, the average nominal lending-rate ceiling over this period was 17.2%, implying an average real lending-rate ceiling of 9.08%.

A brief exercise with the average figures reported above, and the average intermediation costs discussed in the

. .

preceding section is undertaken here. The average implicit cost of funds adjusted by reserve requirements is 5.1% and 5.3% for the development bank and the private bank respectively. Assuming that all lending was done over the period at the maximum ceiling (17.2%), those cost of funds imply gross-spreads of 12.1% for the development bank, and 11.9% for the private bank. Therefore, under this rather optimistic assumption for the effective lending rate, the private bank would result with a (positive) net spread of 3.2%, whereas the development bank would have suffered net losses of 6.7%. Note that for the latter, the alternative of engaging in foreign-funded projects with a 4% allowance for administrative costs is not a profitable option, since the average lending cost reported in section 5.1 implies average losses of 6% in these operations. The main difference however, is that the economies of scale to the expansion of deposit-mobilization in the development bank can sharply reduce the operational losses indicated here. In contrast, cost-saving possibilities under external financing are far more limited, since these depend only on the reduced scaleeconomies associated with lending.

It is important to note in table 38 (appendix C) the estimated correlations between implicit deposit-rates paid by the banks and the deposit-rate ceilings established by the monetary authorities. Nominal implicit rates (cost of

funds) showed near zero correlation with the average deposit-rate ceiling in the development bank. For the private bank this correlation was also very low, although statistically different from zero. Despite this weak relationships between the nominal rates, real implicit deposit-rates were highly correlated (in both banks) with real deposit-rate ceilings. This result is a consequence of the fact that the correlation between two real rates includes as an element the perfect correlation of the inflation rate with itself. This same reason underlies the high values of the correlation coefficients between all real rates (see table 38). This correlation has to be considered when interpreting the results of the cost-function estimations that include real deposit and real lending rates.

Table 15 summarizes the results obtained in the estimation of the effects of interest-rate regulations on bank's non-financial costs. For the two banks (panels A and B), rows 1, 2, and 3 indicate the estimated coefficients obtained including in the cost function the real depositrate (rows 1 and 2), <u>or</u> the real lending-rate ceiling (row 3). The two definitions of real deposit-rate have been included in different estimated equations. The estimated parameters for the real deposit-rate ceiling are reported in column 1, while those associated with the real implicit deposit-rate are presented in column 2. Rows 4 and 5 of the

Table 15. Effects of Interest-Rate Regulations on Lenders' Intermediation Costs: Estimated Parameters of the Real-Deposit Rate, and the Real Lending-Rate in Different Equations. Cost-system Estimates for the Development Bank and the Private Bank.<sup>a/</sup>

•

	Parameter, Variable					
	λ <sub>l</sub> , (d <sub>c</sub> - p): real deposit-rate ceiling	, (d <sub>p</sub> - p): real implicit rate paid on deposits	λ <sub>2</sub> , (l - p): real lending-rate ceiling	Weighted	F-test of Joint Null	t-test of Null
Bank (model)	Estimate (asymptotic t-ratio)	Estimate (asymptotic t-ratio)	Estimate (asymptotic t-ratio)	R-square of the cost-system	Hypothesis: $\lambda_1 = 0, \lambda_2 = 0$	Hypothesis: $\lambda_1 + \lambda_2 = 0$
	(1)	(2)	(3)	(4)	(5)	(6)
A. Development	t bank					
(1)	-0.0383 (-8.48)*			0.78		
(2)		-0.0195 (-6.06)*		0.70		
(3)			-0.0263 (-9.71)*	0.78		
(4)		0.0143 (2.46)†	-0.0403 (-6.91)*	0.73	31.75*	-6.85*
(5)	0.0345 (4.33)*		-0.0584 (-10.32)*	0.83	68.93*	-4.84*

#### Table 15, continued

******	Pa	rameter, Variable	e		,	t-test of Null Hypothesis: $\lambda_1 + \lambda_2 = 0$
Bank (model)	$\lambda_1$ , $(d_c - \dot{p})$ : real deposit-rate <u>ceiling</u> Estimate (asymptotic t-ratio)	$\lambda_1$ , $(d_p - \dot{p})$ : real implicit rate paid on <u>deposits</u> Estimate (asymptotic t-ratio)	<pre>λ<sub>2</sub>, (l - p): real lending-rate ceiling Estimate (asymptotic t-ratio)</pre>	F-test of Weighted Joint Null R-square Hypothesis: of the $\lambda_1 = 0, \lambda_2 = 0$	F-test of Joint Null Hypothesis: $\lambda_1 = 0, \lambda_2 = 0$	
	(1)	(2)	(3)	(4)	(5)	(6)
B. Private bank						
(1)	-0.0214 (-5.58)*			0.97		
(2)		0.0041 (1.20)		0.95		
(3)			-0.0008 (-0.26)	0.95		
(4)		0.0571 (5.53)*	-0.0495 (-5.33)*	0.96	15.36*	2.34†
(5)	-0.0153 (-1.80)		-0.0052 (-0.81)	0.97	15.16*	-4.97*

a/ Other parameters of the cost function not reported. See basic specifications in table 5 (development bank) and table 8 (private bank).

\* : significant at 0.01 level

† : significant at 0.05 level

obtained when the real deposit-rate <u>and</u> the real lending-

In general, the results presented in table 15 indicate that there is an inverse relationship between the levels of real interest-rates and the costs of financial intermediation. Increases in the levels of the real deposit-rate or the real lending-rate will generate reductions in total intermediation costs of the banks. On the other hand, interest-rate restrictions that reduce the levels of the real deposit or lending rates will have cost-increasing effects on the financial intermediaries.

For the development bank, panel A in table 15, the real deposit-rate under its two definitions, and the real lending-rate ceiling show negative and statistically significant coefficients when included in separate regressions (rows 1 through 3). The sign of the real deposit-rate becomes positive when this variable is included together with the real lending rate (rows 4 and 5), a result that is probably a consequence of the high correlation between these real rates. Despite this difficulty, it is pertinent and revealing to test for the effect of a simultaneous change in both the real deposit-rate and the real lending-rate ceiling, i.e., a change that policy-makers could claim is "spread-neutral". These tests for the combined effect of changes in the two rates  $(\lambda_1 + \lambda_2)$  indicate that the net

.affagt.on.thoudownlepment bank's costs will be of opposite sign and significantly different from zero (see column 6). For example, using the results reported in row 5 of panel A, the net effect of a one-point increase in both the real deposit-rate ceiling and the real lending-rate ceiling would be a reduction of 2.4% in total intermediation costs of the  $bank^{8/}(\lambda_1 + \lambda_2 = -0.024, t-ratio = -4.84)$ . Note that the sign and magnitude of the combined effects of simultaneous changes in deposit and lending rates  $(\lambda_1 + \lambda_2)$  is not too different from the individual effects of each of these variables when included in separate equations (rows 1, 2, and 3). Therefore, it is possible to conclude that changes in the real deposit-rate ceiling and the real lending-rate ceiling will bring about changes in total intermediation costs of the development bank in the opposite direction of 2 to 3% per unit of change in the real interest-rate ceilings.

The results obtained for the private bank (panel B in table 15) are similar to the findings discussed above for the development bank. In the private bank, the estimated parameters for the real deposit-rate ceilings show negative signs and are statistically significant in the two equations including this variable (rows 1 and 5, column 1). The variable that represents the implicit cost of funds in real terms (column 2) shows an unstable pattern of effects. This estimated coefficient is not significantly different from  $\frac{3}{1.e.}$ , approximately 0.35 million lempiras in 1982.

zero when this real implicit deposit-rate is the only proxy for interest-rate regulations in the model (row 2). However, this coefficient appears positive and significant when this variable is specified together with the real lending-rate ceiling (row 4). Again, the instability of this coefficient may be reflecting the collinearity between all rates in real terms. The test performed for the combined effect of simultaneous changes in the real deposit-rate ceiling and the real lending-rate ceiling (row 5, column 6) indicates that this combined effect has a negative sign and is significantly different from zero ( $\lambda_1 + \lambda_2 = -0.0205$ , t-ratio = -4.97). This result is similar to that obtained in the same test performed for the development bank. A simultaneous increase in both the real deposit-rate ceiling and the real lending-rate ceiling of one percentage point will generate a 2.05%-decrease in total intermediation costs of the private bank. $\frac{9}{}$ 

The general conclusion that derives from the foregoing analysis is that interest-rate regulations are an important determinant of non-financial intermediation costs in the financial institutions under study. According to these results, interest-rate restrictions that translate into reductions in the real ceilings imposed on deposit and lending rates generate significant cost increases to the

9/ i.e., about 0.8 million lempiras in 1982.

financial intermediaries. It is important to note that the cost increases borne by lenders estimated in this section are only a lower-bound estimate of the total cost-effects of interest-rate restrictions. Part of these total costeffects are passed-on by the financial intermediaries primarily to borrowers, under the form of transaction costs associated with loan procedures established by lenders. Transaction costs of borrowing are the subject of chapter 6.

#### 5.4. Loan-Targeting, Portfolio Composition and Lenders' Intermediation Costs

This section discusses the effects of loan-targeting on the intermediation costs of the development bank. The mechanisms through which targeted funds create additional costs for the financial intermediaries, as well as the methodological approach adopted for this analysis, have been described in previous chapters. Targeted funds are identified as funds obtained from central-bank rediscount lines, or from foreign donors. The term "external funds" is utilized here to refer to both central-bank and foreign funds combined. Non-targeted funds are demand, savings, and time deposits captured from public institutions and from the general public. First, a brief analysis of the relationships betweeen the sources of funds and portfolio composition is presented. Second, the results of the estimated cost function including variables that capture the effects of targeting are analyzed.

### Sources of Funds and Portfolio Composition

The shares of different sources of funds in the development-bank's portfolio of new loans for the period 1971-1982 are presented in appendix C, table 39. The proportion of the total value of new loans funded through deposit mobilization decreased from an average of 56% in the period 1971-1974 to an average of 43% in the last four Consequently, external funds (central-bank and years. foreign funds combined) increased their share in loan amounts from a 44%-average in the first four years to a 57%-average for the period 1979-1982. Foreign funds were predominant among external sources during the first half of the period under analysis, while gradually decreasing in importance within this group after 1975. Central-bank funds became the most important component among external funds in the second half of the period.

The share of deposits in the number of loans decreased drastically from an average of over 40% in the first four years of the period to a remarkably small 5%-average in the last four years, reflecting the re-allocation of these funds to increasingly larger-sized loans. The relationship between the average size of loans granted out of deposits and the average size of loans funded by external sources grew from about 3:1 in the early 70's to over 7:1 in the early 80's (see table 39).

The increased share of external sources of funds both in the number of loans and in the value of loans would suggest that agriculture as a whole, and specially small farmers, have increased their share in the developmentbank's portfolio of new loans. The loan-size figures presented in table 39 and the shares of agricultural loans in the bank's portfolio shown in table 40 allow an examination of the degree to which loan targeting has been effective in modifying the composition of the loan portfolio.

Data on loans by farm size are not available for the period under discussion here, therefore the average loan sizes are used as a reasonable proxy. These figures indicate that the average loan size serviced by external funds has not changed substantially over the period 1971-1982. This, in turn, suggests that the share of small loans in the portfolio has not experienced significant variations in this period. On the other hand, the share of loans to agriculture (crops + livestock) in the development bank's portfolio by the early 30's was at the same level of the early 70's, about 70% of the total value of new loans. The highest shares are observed in 1977-1978, when agricultural loans accounted for over 90% of the portfolio. Within agriculture, crop loans have in general increased their share, while the share of loans to livestock has decreased steadily since 1972.

In short, the increasing share of external funds, through the period under analysis, has not been associated with an increased share of agricultural loans in the portfolio. The peak years of 1977-1978 may be better explained by real-sector phenomena such as the "coffee boom" rather than by an increased proportion of the bank's loan funds being supplied by external sources under targeting arrangements. These findings are reinforced by the results of the correlation analysis between the shares of sources of funds in the portfolio and the share of agricultural loans.

<u>None</u> of the sources of funds showed a statistically significant correlation with total agricultural loans. The correlation coefficients between external funds (combined or separate) and agricultural loans were not statistically different from zero, either taking the shares in the number of loans or the shares in the value of new loans. Significant coefficients were found only when correlating the shares of different sources of funds with the shares of the components of total agricultural loans, i.e., crops and livestock. Table 41 in appendix C summarizes these results, showing the specialization of external funds in terms of the activities financed, central-bank funds primarily financing crop enterprises and foreign funds focusing on livestock activities.

Table 42 in appendix C presents the correlation coefficients between the shares of the different sources of
funds in the portfolio of new loans. There is a high negative correlation between central-bank funds and foreign funds, showing that these external sources have been substituting for each other during the period under analysis. Central-bank funds have been also compensating for the decline in importance of deposits as a source of funds, as denoted by the negative correlation between these two sources.

These findings may be summarized as follows: (a) the growing share of external sources of funds (largely directed towards agriculture) has not been reflected in a significant change in the relative role of agricultural loans in the portfolio. The fungibility of finance is at work here, with external funds substituting for own-deposit funds that have been transferred from agricultural to non-agricultural loans. (b) The increased share of external funds may have induced the re-allocation of non-targeted funds to increasingly larger-sized loans in the non-agricultural sector. This cost-saving adjustment compensates for the increasing costs of handling a growing proportion of external funds in the "targeted" portion of the loan portfolio.

## Effects of Loan-Targeting on the Costs of the Development Bank

The cost-function estimates including expression (4.17) to capture the effects of different sources of funds on intermediation costs are presented in table 16. Model 1 in

this table includes the indicator variable for non-targeted funds (own-deposits,  $S_1$ ), and the lagged combined effect of external funds (central bank and foreign funds). Model 2 also includes the non-targeted-funds variable, but separates the effect of (current-year) central bank funds from those of foreign funds with a one-year lag.

The estimated coefficients for the variable that captures the effects of increases in the amount of non-targeted funds (S $_1$ , deposits) are not statistically different from zero, with very low asymptotic t-ratios in both models. Targeted funds show significant cost-increasing effects, whether they are included as a combined variable (column 1), or as separate effects (column 2). Given the typical features of central-bank and foreign-funded projects, model 2 is a more appropriate representation of these targeting schemes than model 1. The results of model 2 indicate that increases in central-bank funds will have a contemporaneous cost-increasing effect on the development bank ( $\omega_2 > 0$ ), given the short-term nature of the targeted programs funded through rediscount lines of the central bank. On the other hand, additional funds originated in foreign sources, usually targeted to medium-to-long-term activities with extended periods of disbursement, will exercise a costincreasing effect with a one-year lag  $(\hat{\omega}_3 > 0 \text{ for } S_{3(t-1)})$ .

The definition of the variables that account for the effects of different sources of funds does not allow a

Table 16. Effects of Loan-T Intermediation Co Development Bank: Coefficients for of Funds.a	argeting on t sts of the Estimated Different Sou	he rces
Source of Funds, Parameter	Estimated Co in Different (1)	efficients Models <sup>b</sup> (2)
Non-targeted funds		
$\omega_1$ , deposits (S <sub>1</sub> ) Targeted funds	0.0056 (0.31)	0.0145 (0.76)
$\omega_2$ , central bank (S <sub>2</sub> )		0.1129 (4.20)*
ω <sub>23</sub> , lagged central bank and foreign (S <sub>23(t-1)</sub> )	0.0756 (2.81)*	
ω <sub>3</sub> , lagged foreign funds (S <sub>3(t-1)</sub> )		0.0525 (1.94)†
Weighted R-square of the cost system	0.73	0.74
F-test of Joint Null Hypothesis: $\omega_i = 0$ , all i	3.13†	5.30*

<u>a</u>/ Cost-system estimation, other parameters of the cost function not reported. See basic specification of the cost function in table 5.

b/ Asymptotic t-ratios in parenthesis.

\* : significant at 0.01 level
t : significant at 0.05 level

direct interpretation of the magnitudes of the estimated coefficients. However, comparisons of effects across variables are valid. The cost-increasing effects of central bank funds appear twice as high as those associated with lagged foreign funds. However, the null hypothesis that these two coefficients are equal to each other can be rejected only at the 0.10 level (i.e., accepting a 10%-probability of error).

In summary, these results support the hypothesis that there is a lagged, "ratchet"-type effect of targeted funds on the intermediation costs of the development bank. Overall intermediation costs are increased as a result of additional funding received from external sources. This effect is more significant in the case of rediscount lines of credit coming from the central bank than in the case of foreign-funded projects. On the other hand, greater reliance on deposits as a source of loan-funds will not affect overall intermediation costs of the development bank.

# 5.5. Loan Delinguency and Intermediation Costs in the Development Bank

Delinquency is a common characteristic of the loan portfolio of most development banks. Therefore it is pertinent to investigate the degree to which the delinquent portfolio in the development bank has contributed to the lending costs of the bank, above and beyond the costs considered up to this point.

This section presents a brief discussion of the delinquency record in the development bank, emphasizing the biases in the measurement of delinquency rates. At the same time, the results of the estimated cost function are analyzed including different indicators of delinquency. The estimated effects that delinquency has on bank's costs are interpreted as an indication of managerial response to this problem. The mechanisms of response were discussed in chapter 4.

#### Delinquency Records and Measurement Problems

---- -

Composition of the loan portfolio according to loan delinquency status is summarized in table 17. In this table, loans that have been rescheduled or refinanced are classified separate from non-delinquent balances, in order to provide a more complete description of the "health" of the loan portfolio. Table 17 shows that the loan portfolio has (apparently) deteriorated over time. Delinquent balances alone accounted for an average of 21% of the value of the loan portfolio during the period 1970-1974. Their relative importance grew to an average of 32% in the period 1975-1978, and subsequently to almost 39%, on average, in the period 1979-1982. If rescheduled and refinanced loans are added to delinquent balances, this broader definition of the delinquent loan portfolio goes from 34% of total loans. outstanding in the first period indicated above, to 45% in

	Percent	of Total Loans Ou	tstanding
	(1)	(2)	(3)
Year	Delinquent Balancesª/ %	and Refinanced Loans	Non-Delinquent Balances %%
1970	21.4	11.2	67.4
1971	24.9	11.5	63.6
1972	18.0	13.8	68.2
1973	18.1	15.5	66.4
1974	22.7	14.2	63.1
1975	25.9	13.4	60.7
1976	32.3	12.3	55.4
1977	35.5	11.6	52.9
1978	33.7	13.1	53.2
1979	31.9	12.6	55.5
1980	41.3	8.9	49.8
1981	36.9	11.5	51.6
1982	43.8	5.9	50.3

Table 17. Delinquent Balances and Refinanced Loans as a Proportion of Total Loans Outstanding in the Development Bank, 1970-1982

Source: 1970-1980, Graham et al. [36], Chapter V. 1981, 1982, computed from BANADESA records

a/ Loans one day or more overdue

the second period, reaching an average of 48% of the loan portfolio in the last 4-year period (1979-1982).

The foregoing descriptive record of delinquency for the bank requires some qualifications that arise from the definition of delinquency utilized in the bank, and the method for the calculation of delinquency rates. As indicated in chapter 4, delinquency rates are computed as the ratio of delinquent balances over total loans outstanding (i.e., column 1 in table 17). The bank defines a loan as delinquent or overdue only when all installments have become past-due. Therefore, overdue installments on long-term loans are not included as delinquent balances until the last repayment becomes overdue. However, the total amount of the same loan is included in the figure of total loans outstanding, i.e., in the denominator of the ratio that defines the delinquency rate.

This procedure to measure delinquency implies that these delinquency rates are underestimated as compared to a ratio of overdue balances to balances due. At the same time, the magnitude of this bias depends on the share of long term loans in the portfolio. The higher the share of these loans and the longer the term structure, the more underestimated is the delinquency rate. During the period under analysis (1970-1982), the share of long-term loans in the overall portfolio has decreased consistently, from an average of 42% for the years 1970-1974, to 32% on average

during the period 1975-1978, and even further to about 20% in the last four years of the series. 10/ This decreasing share of long-term loans is clearly related to the decreasing proportion of foreign funds among the bank's sources of funds, reported in the preceding section. Foreign-funded programs, typically targeted to long-term projects, have the property of "disguising" delinquency when the corresponding indicators are computed as described above.

These variations in the term structure of the loan portfolio and their implications for the measurement of delinquency have to be taken into account in the analysis of the figures reported in table 17. First, all delinquency rates reported in the table are underestimating the "true" delinquency rates that would result from calculating a ratio of overdue balances to balances due. Second, the pattern of growing delinquency may be misleading, since the degree of underestimation is more serious in the earlier years of the series, due to the higher share of long-term loans in the portfolio at that time. It can be asserted that "true" delinquency has not grown as the rates reported in table 17 suggest. Whether the actual delinquency rate has grown at a different rate, has remained stable, or has decreased, cannot be ascertained with the information available.

An alternative way of looking at the delinquency problem considers the "age" of delinquent balances. If old

<sup>10/</sup> See Graham et al. [36], chapter V. This report discusses other features of the delinquent portfolio of the development bank, that are omitted here.

delinquent balances become more important in the delinquent portfolio, then the overall probability of repayment is decreasing. This situation indicates that loan recovery efforts are non-existent or ineffective, and loan-quarantees are nonexistent or not enforced. Table 18 sets forth the age composition of the delinguent portfolio of the development bank for the period under analysis. The shares of delinquent balances of different ages reported in this table indicate that the delinquent portfolio has become increasingly older throughout the period. Even though the proportion of delinquent loans more than 90 days overdue has remained fairly stable, around 80 or 90%, the proportion of balances with more than one year overdue shows a growing trend. On average, loans more than one year overdue accounted for 40% of the delinguent portfolio during the period 1970-1974, decreasing to 32% in the subsequent 4-year period (1975-1978), and then increasing to an average of 52% in the period 1979-1982. The results reported below will indicate that loan-recovery initiatives have existed in the bank, as a response to growing delinguency, but the figures discussed above suggest that these efforts have been in general ineffective.

## Effects of Delinquency on Intermediation Costs

The estimated cost function of the bank reported in table 19 includes as explanatory variables the delinquency

	Percent of Total Delinquent Balances					
	verdue					
Year	Less than 90-days overdue (1) %	Total over 90 days overdue (2) %	More than 90 days but less than l year (3) %	More than l year overdue (4) %		
1972	12.6	87.4	33.7	53.7		
1973	7.7	92.3	53.3	39.0		
1974	25.0	75.0	45.9	29.1		
1975	15.0	85.0	59.8	25.2		
1976	14.5	85.5	72.8	12.7		
1977	9.5	90.5	63.5	27.0		
1978	8.5	91.5	29.5	62.0		
1979	7.4	92.6	46.1	46.5		
1980	22.4	77.6	28.1	49.4		
1981	17.5	82.5	31.5	51.0		
1982	15.4	84.6	26.0	62.6		
1902	13.4	0-1-0	20.0	02.0		

Table 18. Composition of Delinquent Balances by Age in the Development Bank, 1972-1982

Source: BANADESA, Economic Studies

rate (DR), and the proportion of delinquent balances more than 90 days overdue (DA). The hypotheses formulated in chapter 4 stated that increases in delinquency will have cost-increasing effects due to the allocation of additional resources to loan recovery activities, and to the tightening of loan procedures. This latter effect is likely to operate with a lag, whereas the intensification of loan recovery is expected to be the first response of the bank's management to upward trends in delinquency.

The delinquency rates and proportion of "old" loans in the delinguent portfolio were calculated for all branches over the 1971-1982 period and included in the cost-function. Model (1) in table 19 includes these two variables, model (2) considers the current and lagged effect of the delinquency rate, and model (3) specifies the lagged values of the delinquency rate, and the lagged share of loans more than 90 days overdue. In general, the results presented in table 19 indicate that there is a cost-increasing effect of delinquency on the operations of the development bank. In all three models, the joint F-test for the inclusion of the delinquency-related variables is significant, indicating that these variables can be considered relevant in the specification of the cost-function. The current-year level of the delinquency rate shows a positive and significant coefficient in the models in which the variable is included

		in Di	ifferent Mod	els <sup>b</sup> /
Parame	ter, variable	(1)	(2)	(3)
Curren	t Effects			
<sup>π</sup> 10′	Overall delinquency rate (DR)	0.3567 (3.61)*	0.4419 (3.28)*	
<sup>π</sup> 20′	Proportion of delinquency over 90 days-overdue (DA)	0.0798 (0.81)		
Lagged	Effects			
<sup>π</sup> 11′	DR (t-1)		-0.0642 (-0.45)	0.1995 (1.82)°
<sup>π</sup> 21′	DA (t-1)			0.1642 (1.67)°
Weight cost	ed R-square of the system	0.75	0.75	0.77
F-test hypo	of joint null thesis:	5.59*	6.33*	2.92†
πi	= 0, all i			

Table 19. Effects of Delinquency on the Intermediation Costs of the Development Bank: Estimated Coefficients for Current and Lagged Effects<sup>a</sup>/

a/ Cost-system estimation, other parameters of the cost function not reported. See basic specification of the cost function in table 5.

b/ Asymptotic t-ratio in parenthesis

\* : significant at 0.01 level
† : significant at 0.05 level
° : significant at 0.10 level

included (models 1 and 2). However, the estimated coefficient for the current level of the variable that accounts for the "age" of delinquency is not significantly different from zero (see model 1 in table 5.17). The lagged value of the delinquency rate does not show a significant coefficient when it is included together with its current-year value (model 2), whereas its estimated coefficient is positive and significant in model 3. In this model, the lagged values of the two variables show positive and significant coefficients. Furthermore, the weighted R-square obtained for this model is the highest among the models specified in table 19.

In the three models reported in table 19, the sum of the coefficients of the two variables included fluctuates between 0.36 and 0.43, indicating that an increase in loandelinquency of one percentage point would generate an increase in costs of about 0.4%. The results of model 3 suggest that aging plays a role in this cost-increasing effect, and that the response to increased delinquency occurs with a lag. In short, additional resources are devoted to loan-recovery and/or to loan-evaluation and monitoring, once an increase in the level of delinquency has been detected. However, there appears to be a delay in this response and, more importantly, these efforts have not been succesful in improving the "health" of the loan-portfolio of

the bank. High levels of delinquency, and increasingly older delinquent balances continue to characterize the loan portfolio of the development bank.

-

•

#### CHAPTER VI

Transaction Costs and Borrowing Costs

This chapter presents first a brief overview of the results obtained from the field-survey on borrowing costs described in the methodology chapter. Second, the results of the econometric analysis of the trade-off model for borrower's transaction costs and the interest rate are presented and discussed. This section emphasizes the main factors affecting both the level of transaction costs and the magnitude of the trade-off between these costs and the explicit interest rate charged on loans.

## 6.1. Overview of Survey Results

A total of 198 interviews of farmer-clients of financial institutions were conducted in July-August 1981, throughout four main agricultural regions of Honduras. The general characteristics of the sample are summarized in appendix D, tables 43 through 46. Approximately one-half of the total sample of farm borrowers had loans less than 5,000 lempiras (i.e. \$2,500 at the current exchange rate of two lempiras equal to one dollar). The average loan size however was close to 23,000 lempiras indicating a clear asymmetry or skewness in the overall distribution of loans.

Although the distribution of the clientele for each loan source overlaps to some extent, each can be clearly identified with respect to the predominant scope of their operations in terms of loan and farm size. Rural credit unions in Honduras are the classic small-farmer loan source with most loans below 2,000 lempiras on farms typically less than 20 hectares. At the other extreme are the farmerclients of the private commercial banks with the larger proportion of their loans over 25,000 lempiras on farms generally above 100 hectares. The National Agricultural Development Bank (BANADESA) activity falls in between with a majority of its loan operations between 1,000 and 10,000 lempiras on farms largely between 10 and 100 hectares.

The pattern of association between loan size and farm size is fairly similar across different loan sources. Small loans tend to be directed to smaller farms and large loans appear concentrated in larger farm sizes. However, this relationship is not as strong as might have been expected, a correlation coefficient between loan size and farm size estimated for the overall sample was close to 0.5. Tables 44 through 46 in appendix D explain this finding by the presence of relatively large loans made to small farms by private banks, and relatively small loans being granted to some large farms by credit unions. BANADESA appears to be the source of loans for which the observed association between farm size and loan size more closely fits the

expected pattern. It must be recalled that this expected positive relationship is based on the assumption that farm size is a good indicator of farmer's wealth and therefore creditworthiness. However, this may not necessarily be the case in Honduras, given the heterogeneity of land quality observed across regions and within regions.

#### 6.1.1. Overall Sample Results

Borrowing transaction costs were defined (section 4.2) as all those non-interest explicit and implicit expenses incurred by the <u>borrower</u> in the process of obtaining a loan. Explicit expenses refer basically to the following: (a) Cost of transportation, lodging and meals when travelling to the office of the institution granting the loan, or to other places with the purpose of obtaining related documents. (b) Fees, taxes or other charges associated with the issuing of documents, registration of guarantees or collateral, contracts and the like. (c) Explicit charges imposed by the lending institutions in the process of handling the application.

Implicit transaction costs directly related to borrowing correspond to the value of the time forgone by the farmer attributable to negotiating and securing a loan. The minimum wage rate (5 lempiras per day) was imputed as a lower-bound estimate of the opportunity cost of time diverted from farming operations. In many cases a higher

value could have been imputed to farmer's time, but using a lower limit has the advantage of obtaining a measure of transaction costs that represents a lower boundary as well. Total borrowing costs are computed as the sum of the explicit interest rate charged on the loan <u>plus</u> the transaction costs expressed on a per lempira basis.

A summary of the results obtained for the total sample of individual farmers is presented in Table 20. As pointed out before, the distribution of loan amounts approved is asymmetric, concentrated around small loan sizes. Likewise, the amounts actually disbursed as well as the different measures of transaction costs show a similar skewness. This common characteristic of the distribution of the variables involved in the analyses makes more appropriate the use of the median values as a basis for the presentation of descriptive results, since these should reflect more accurately (than the arithmetic mean) the central or typical values of the different variables.

It is interesting to note in table 20 the relationship between the amount actually disbursed in credit operations and the amount originally approved. The proportion disbursed (76 percent, taking the aggregate figures, or 66 percent considering the ratio of the two median values) is rather low considering the period of the year in which the survey was carried on. As of August, the first growing season ("primera") is precisely coming to its end, i.e,

		Mean Value	Median Value
1.	Amount of Loan Approved, L.	22,930.1	5,289.5
2.	Amount Disbursed, L.	17,414.8	3,500
з.	Ratio Disbursed/Approved	0.76	0.66 <u>a</u> /
4.	Transaction Costs		
	(i) per loan (Lps.)	226.88	57.75
	(ii) per Lempira (%)		
	Approved Disbursed	2.50 3.54	1.26 1.82
5.	Interest Rate (%)	13.54	13.00
6.	Total Borrowing Costs, per Lempira (%)		
	(i) Approved (ii) Disbursed	15.95 16.98	15.25 16.11

Table 20. Mean and Median Values of Transaction Costs and Total Borrowing Costs (Per Loan and Per Lempira) for Total Sample of Borrowers

a/ Ratio of the two median values.

Source: Survey results.

loans granted for that reason should be completely disbursed. The low proportion of long-term loans in the sample corresponded mainly to operations in 1980, implying that all loans were also disbursed. In sum, the observed ratio disbursed/approved should not be attributed to biases in the data collection. Rather, it may be an indication of either or both of the following phenomena:

-- <u>First</u>, the influence (guidance, counseling or outright dominance) of the bank official or agent in the determination of the amount of the loan to be applied for (an element of "supply-led finance"). The client then follows the official's "advice" when filling out the loan application, but will request disbursements according to his own idea of the level of indebtedness he is willing to undertake. The reasons for this being lower (on average) than the amount approved leads to the second explanation.

-- <u>Secondly</u>, even though agricultural credit in the country may appear as "cheap money" for farmers, a less than 100 percent disbursement rate would be counter-evidence for that assertion. This apparent contradiction suggests that credit conditions may not be as "cheap" when compared with the investment alternatives available to farmers, and that they do not have (in general) efficient ways to divert agricultural loans to other more profitable uses. Either the financial system is not well developed at the rural level,

or loan monitoring and supervision is active enough to prevent credit diversion. It is likely that both arguments are valid in the case under analysis.

The tables included in this section present transaction costs per lempira both as a percent of the amount approved and as a percent of the amount disbursed. The former is interpreted as the implicit interest that the lending institution "intended" to charge whereas the latter corresponds to the actual cost incurred by the borrower.

The aggregate results for the sample as a whole reported in table 20, indicate that the various elements of borrower transaction costs added roughly three percentage points to the average explicit interest rate of 13 percent. This represents almost one quarter of the interest rate. The incidence of these borrowing costs by loan and farm size in the sample, and the differences observed between different institutional lenders are briefly discussed in the following sections.

## 6.1.2. Different Loan Sizes

The behavior of transaction costs and total borrowing costs, by loan size category is shown in table 21. Several consistent patterns can be observed in this table. The cost of obtaining a loan increases systematically with the size of loan (excepting for only one "drop" in the 15,000 to 25,000 lempira category). Notwithstanding this behavior, transaction costs per lempira borrowed decrease consistently

	Trai	Transaction Costs			Total Borrowing Costs	
Loan Size	Per Loan	Per Loan Per Lempira (%)		Rate	Per Lem	pira (%)
Category (Lps.)	(Lps.)	Approved	Disbursed	(%)	Approved	Disbursed
Less than 1,000	30.75	5.87	6.0	13	18.92	19.23
1,001 - 2,000	42.0	2.86	4.33	13	16.07	17.73
2,001 - 5,000	44.88	1.18	1.52	13	14.88	15.77
5,001 - 10,000	53.0	0.77	1.13	13	14.03	14.94
10,001 - 15,000	86.75	0.81	1.12	13	14.56	14.87
15,001 - 25,000	42.75	0.20	0.44	13.5	13.89	14.35
25,001 - 50,000	131.50	0.40	0.95	14	14.40	15.71
50,001 - 100,000	322.50	0.42	0.63	13	13.17	13.63
More than 100,000	1,414.50	0.83	1.01	11	12.09	12.36

Table 21. Components of Total Borrowing Costs, by Loan Size $\frac{a}{}$ 

a/ All values are median values. Therefore, the median values of total borrowing costs are not necessarily the sum of the median values of the separate transaction costs per lempira <u>plus</u> the median value of the interest rate, as they would be if mean values had been used.

Source: Survey results.

as the loan size increases. In other words, the increase in transaction costs per loan occurs at a slower rate than that of the loan size itself. Note that there appears to be a trade-off between the explicit interest rate and implicit charges (transaction costs) such that the behavior of total borrowing costs follows an almost perfectly decreasing order, from smaller loans to larger loans.

What appears as a very uniform treatment of all loans, if only the levels of explicit interest rates are considered, turns out to be a consistently negative relationship between borrowing costs and loan size. This result indicates the presence of rationing mechanisms exercised by lending institutions against small loans, due to their relatively high costs of handling and monitoring as well as the eventual higher default risk associated with these loans.

## 6.1.3. Different Farm Sizes

As pointed out before, the association between loan size and farm size was found significant but relatively low. However, table 22 shows that the behavior of transaction costs and total borrowing costs observed with respect to loan size are also apparent and consistent when contrasting the measures obtained for different farm-size categories. Transaction costs per loan increase as farm size increases, the reverse pattern is observed for transaction costs per lempira and for overall borrowing costs. These results suggest that the farm size of the borrowing unit is being

	Transaction Costs			Interest	Total Borrowing Costs	
Farm Size	Per Loan	Per Lem	npira (%)	Rate	Per Lem	pira (%)
Category (Has.)	(Lps.)	Approved	Disbursed	(%)	Approved	Disbursed
Less than 5	31.75	2.97	4.31	13	16.0	17.33
5.1 - 10	40.0	2.39	4.68	13	15.07	17.14
10.1 - 20	53.5	1.65	2.68	13	16.20	17.67
20.1 - 50	56.25	1.00	1.74	13	14.64	15.52
50.1 - 100	75.0	0.84	1.97	13	14.84	15.64
100.1 - 200	133.75	1.23	1.68	13.5	16.52	17.52
More than 200	149.25	0.41	0.60	13	13.82	14.02

Table 22. Components of Total Borrowing Costs, by Farm Size $\frac{a}{a}$ 

All values are median values. Therefore, the median values of total borrowing costs are not necessarily the sum of the median values of the separate transaction costs per lempira <u>plus</u> the median value of the interest rate, as they would be if mean values had been used.

Source: Survey results.

considered by banks as an indicator (although imperfect) of creditworthiness, tending to charge implicit risk-premia through transaction costs to smaller-farm operations.

## 6.1.4. Different Lenders

The borrowing costs associated with different institutional sources of loans are presented in table 23, broken down by the two basic components of borrowing costs: transaction (non-interest) costs, and the interest rate.

The lowest transaction costs per loan correspond to those granted by credit unions, where obtaining a loan would typically cost a farmer less than 18 lempiras. This figure rises to 60 lempiras in the case of BANADESA, and even higher for private banks, 136 lempiras. Private banks become the least expensive source of credit when transaction costs are expressed on a per lempira basis, as well as when total borrowing costs are considered. This is a direct result of the differences in loan size distribution between the sources set forth before. Overall borrowing costs in BANADESA and credit unions amount to a similar figure, although transaction costs per lempira appear higher in the case of BANADESA.

Summarizing the results of this and preceding sections, it has been found that:

(a) transaction costs per loan

-- are positively related to loan size,

	Trai	Transaction Costs			Total Borrowing Costs	
	Per Loan	Per Loan Per Lempira (%)		Rate	Per Lempira (%)	
Source	(Lps.)	Approved	Disbursed	(१)	Approved	Disbursed
BANADESA	60	1.71	2.69	13	15.30	16.57
Private Banks	136	0.70	1.01	13	13.54	14.11
Credit Unions	17.75	1.43	1.62	12	16.11	16.52

## Table 23. Borrowing Costs Associated with Different Institutional Sources of Credita

a/ All values are median values. Therefore, the median values of total borrowing costs are not necessarily the sum of the median values of the separate transaction costs per lempira <u>plus</u> the median value of the interest rate, as they would be if mean values had been used.

.

Scurce: Survey results.

-- are positively related to farm size,

-- differ sharply between sources of credit.

- (b) transaction costs <u>per lempira</u>
   -- are negatively related to loan size,
   -- are negatively related to farm size,
   -- vary considerably between lending institutions.
- (c) lending institutions tend to substitute implicit charges (that translate into transaction costs) for explicit interest, due to the limited discretionary power they have to determine these rates.

The econometric analysis presented in the following section will assess the statistical significance of some of the foregoing results.

### 6.2. The Transaction-Costs Function

This section presents the results obtained in the estimation of the transaction-cost function specified in chapter 4. Throughout this section, the functional form specified as equation (4.23) is referred to as the "basic model". The function that allows for interactions in the trade-off between transaction costs and interest rate is labeled "extended model". After the overall results of these two versions are presented and discussed, section 6.2.1 will focus on the specific findings for the relationships between transaction costs and loan amount. Section 6.2.2 concentrates on the results obtained for the trade-off

between transaction costs and interest rate, and the factors that affect its magnitude.

The extended version of the transaction-costs function can be written as:

$$lnT = a_0 + a_1 lnA + a_2 lnL + a_{30} ln(i) + a_{31} sln(i) + a_{32} D_1 ln(i) + a_{33} D_2 ln(i) + a_{34} Fln(i) + b_1 D_1 + b_2 D_2 + c_1 U_1 + c_2 U_2 + c_3 U_3$$
(6.1)

where,

- T is the borrowing (non-interest) transaction costs per loan,
- A is the area of the farm,
- L is the loan amount,
- i is the explicit interest rate that can be charged on the loan by the lender,
- S is a dummy variable for loan-size category,
  - S=l if the loan amount is less than or equal to
    L. 2,000,
  - S=0 otherwise,
- D1 and D2 are dummy variables that account for deviations of T in private banks with respect to the development bank, that is used as the base or level of reference,
  - $D_1 = 1$  if the lender is a private bank,
  - $D_1=0$  otherwise,
  - $D_2=1$  if the lender is a credit union,

 $D_2=0$  otherwise,

- F is a dummy variable for farm-size category,
  - F=1 if the area of the farm is less than or equal to 20 hectares,
  - F=0 otherwise,

 $U_1$ ,  $U_2$ , and  $U_3$  are dummy variables defined to capture the effects on transaction costs of different loanuses: basic grains, export crops, and livestock, as deviations with respect to a miscellaneous end-use category conformed by all other end-uses in agriculture (land purchases, trade, vegetable crops, and others),

 $U_1=1$  if the stated end-use of the loan is basic grains,  $U_1=0$  otherwise,

 $U_2$ =1 if the stated end-use of the loan is export crops,  $U_2$ =0 otherwise,

 $U_3$ =1 if the stated end-use of the loan is livestock,  $U_3$ =0 otherwise.

Therefore, the basic model (4.23) is equivalent to the extended model (6.1) with zero restrictions on the parameters  $a_{31}$ ,  $a_{32}$ ,  $a_{33}$ , and  $a_{34}$ .

The results of the OLS estimation of the basic model and the extended version are presented in table 24. The extended model (column 2 in table 24) has been simplified dropping the interactions of farm size and credit unions with the interest-rate variable, which did not add statistical significance to the overall function. The F-statistic computed to contrast the extended model including these two interactions and the simplified one presented in table 24 has a value of 0.14, therefore the hypothesis that the two additional coefficients are equal to zero is not rejected. Moreover, neither one of the coefficients for these two interaction variables was significantly different from zero.

		(1)		(2)	
		Basic	model	Extende	d model
Para	ameter (Independent Variable)	Estimate	t-ratio	Estimate	t-ratio
a <sub>0</sub>	(intercept)	3.4225	3.397*	1.9183	1 <b>.7</b> 45°
a <sub>l</sub>	(lnA, area of the farm)	0.0001	0.002	0.0448	0.605
a <sub>2</sub>	(lnL, loan amount)	0.3388	3.844*	0.4461	4.027*
a <sub>30</sub>	(lni, interest rate)	-0.9237	-3.782*	-0.8425	-3.390*
a <sub>31</sub>	(Slni, small loans x interest rate)			0.2874	2.833*
a32	(D <sub>l</sub> lni, private banks x interest rate	)		-2.1141	-2.536*
b1	(D <sub>1</sub> , private banks)	0.5879	2.201†	6.1259	2.690*
b2	(D <sub>2</sub> , credit unions)	-0.8329	-3.113*	-0.7997	-3.126*
c <sub>1</sub>	(U <sub>1</sub> , basic-grains loans)	0.3607	1.357	0.2980	1.176
c <sub>2</sub>	(U <sub>2</sub> , export-crops loans)	-0.3394	-1.085	-0.2943	-0.987
c3	(U <sub>3</sub> , livestock loans)	0.4925	0.329	0.3844	1.204
R-so	quare	0.4652		0.5207	
F-va	alue	18.81*		18.58*	

Table 24. Estimated Coefficients of the Transaction-Costs Function, Under Different Model Specifications. Dependent Variable : lnT.

OLS estimation, N=182. S,  $D_1$ ,  $D_2$ ,  $U_1$ ,  $U_2$ , and  $U_3$  are dummy variables. Levels of significance: \*, 0.01; †, 0.05; °, 0.10.

In general, the results shown in table 24 are satisfactory from the statistical point of view. R-square values are fairly high considering that the sample consists of a set of cross-sectional observations. Most individual coefficients are significantly different from zero and show the expected signs, according to the discussion presented in section 4.2. The results corresponding to the different factors entering the estimated equations are outlined below. (a) Area of the farm, A

An exception to the significance of individual coefficients is the farm-area variable (A). The parameterestimate is not significantly different from zero in any of the estimated models. This result may be reflecting the fact that farm size does not constitute a good proxy for farmer's wealth in the Honduran case, given the heterogeneity of land quality across and within different areas of the country. Two other proxies for farmer's wealth or firm size were substituted for farm area in different regressions with the results outlined below.

First, the ratio of the value of the loan guarantee to the loan amount was used instead of area of the farm. The regressions estimated using this ratio showed lower R-square values than those obtained in the equations including farm area (see table 47, column 1, in appendix D). The individual coefficient of the ratio loan-guarantee/loan-amount was also non-significant. The values and significance of the

other coefficients in the model were similar to those obtained with the model including farm area, presented in table 24. Another proxy for wealth was total labor costs. The results obtained using this variable were essentially the same as those reported for the models using farm area as proxy for firm size (see table 47, columns 2 and 3). Overall statistical significance and values of the estimated coefficients were very similar between these two alternative models. In view of these similarities, it was decided to keep farm area as the proxy for firm size, on the basis that this variable is more easily available for simulation purposes using the estimated equations.

(b) Loan amount, L

The results obtained for the relationship between transaction costs and loan amount were as expected in terms of sign and magnitude of the estimated coefficients. In all equations the estimated parameter for this variable was significantly greater than zero and less than one. The t-ratios for the null hypothesis that the parameter is zero ranged between 3.8 and 5.4 across different equations (see tables 24, and 47 in appendix D). On the other hand, t-ratios for the null hypothesis that the parameter is equal to one ranged between -4.7 and -7.6. In all cases the null hypothesis can be rejected at the 0.01 level of significance. The 99%-confidence intervals for the parameter

a<sub>2</sub> (loan amount) computed with the results presented in table 24 are the following:

basic model (column 1 in table 24),  $0.1082 \leq a_2 \leq 0.5694$ , extended model (column 2 in table 24),  $0.1564 \leq a_2 \leq 0.7358$ . These results indicate that transaction costs per loan are an increasing function of the amount of the loan. However, the elasticity of transaction costs per loan with respect to changes in the loan amount is less than one. Therefore, transaction costs per loan increase as loan size increases, at a decreasing rate.

The implicit values for the elasticity of transaction costs per lempira  $(\tau)$  with respect to the loan amount are -0.6612 in the basic model, and -0.5539 in the extended version. Thus transaction costs per lempira are a decreasing function of the loan amount. As discussed in section 4.2, this result implies that the absolute value of the slope of the transaction-costs per lempira curve with respect to loan amount will decrease as the loan amount increases. These relationships between transaction costs and loan amount will be discussed with more detail in section 6.2.1. In this section these results will be laid out in graphs that emphasize the role of the interest rate and the loan source as shift parameters in the relationship between transaction costs and loan amount.

(c) Interest rate, i

The results obtained for the effect of this variable on transaction costs are also significant and stable across different specifications. The coefficient associated with the direct effect of the interest rate,  $a_{30}$ , is significantly less than zero and not significantly different from -1 in all estimated equations.<sup>1</sup>/ These results indicate that there exists a trade-off between transaction costs of borrowing and the explicit interest rate charged on loans, such that a one-percent increase in the interest rate will bring about a one-percent decrease in borrowing transaction costs.

The foregoing results must be qualified when the interactions with the dummy variables for small loans (S) and private banks  $(D_1)$  are considered. These two interactions showed significant coefficients when included in the estimated equations (see table 24, column 2, also table 47, column 3). As indicated before, the interactions of farm size and credit unions with the interest-rate variable were not significant and were dropped from the extended model in order to simplify the equation and its analysis.

The positive sign shown by the interaction between interest rate and small loans  $(a_{31})$  indicates that in these cases the magnitude of the trade-off becomes smaller in

<sup>1/</sup> t-ratios for the null hypothesis  $a_{30} = 0$  range between -3.3 and -3.8 (Ho rejected); t=ratios for the null hypothesis  $a_{30} = -1$  show values between 0.22 and 0.74 (Ho not rejected).

absolute terms, than in the case of medium-size or larger loans. $\frac{2}{}$  In other words, there is less substitution between implicit interest (transaction costs) and explicit interest for loans of very small amounts (less than 2,000 lempiras). The coefficient associated with the interaction between the interest rate and the dummy variable for private banks (a32) is negative and significant. Thus the magnitude of the trade-off between transaction costs and the interest rate is larger when the lender is a private bank. This is a result of greater flexibility of loan procedures in these banks, and more prompt response of their management to changes in the economic environment, than is the case in the development bank and in the credit unions. This trade-off between transaction costs and interest rate is analyzed with further detail in section 6.2.2. In this section, the values and statistical significance of the estimated elasticities between transaction costs and interest rate for different combinations of loan sources and loan sizes will be presented and discussed.

(d) Loan sources, D<sub>1</sub>, D<sub>2</sub>

The results presented in table 24 and the tests of significance shown in table 25 indicate that loan source is a relevant factor in determining the level of transaction costs. The values of the F-statistics reported in table 25 reject the joint null hypothesis  $b_1=0$ ,  $b_2=0$ , in both the  $\frac{2}{2}$  Recall that  $\frac{\partial \ln \tau}{\partial \ln i} = a_{30} + a_{31}S + a_{32}D_1$ , < 0.

	Basic	Model <u>a</u> /	Extended	Model <u>a</u> /
Factor, Null Hypothesis	F-value	Signif- icance	<b>F-value</b>	Signif- icance
Loan Source, H: $b_1 = b_2 = 0$	8.51	0.01	6.35	0.01
End-Use H: $c_1 = c_2 = c_3 = 0$	3.70	0.01	2.69	0.05

Table 25. Tests of Significance of the Effects of Loan Source and Loan-Use on Borrowers' Transaction Costs. Values of the F-Statistics for Joint Null Hypotheses, under Different Model Specifications.

a/ See Table 24 for detailed results of these models.

basic model and the extended model. The coefficient associated with the dummy variable for private banks  $(b_1)$  shows a positive sign and is significantly different from zero (see table 24). This result reflects the greater ability of private banks to pass-on transaction costs to borrowers. Therefore, for any given loan amount, borrowing from private banks entails higher transaction costs than borrowing from other institutional sources.

The results presented in table 24 show that credit unions are the least expensive source of funding for farm borrowers, at given loan sizes and interest rates. The coefficient of  $D_2$ , the dummy variable for credit-union loans, is negative and significantly different from zero, denoting lower transaction costs associated with borrowing from these financial institutions. This result is explained
by the existence of less complicated loan procedures in credit unions, where loans are usually approved on the basis of personal knowledge of the applicant, thus requiring little time and documentation.

(e) Loan uses,  $U_1$ ,  $U_2$ ,  $U_3$ 

The results obtained for end-use of loans as a factor determining the level of transaction costs are less clear than the findings discussed above for the effects of loan sources. In the case of loan use, the joint null hypothesis that the effects of different loan uses are non-significant  $(c_1=c_2=c_3=0)$  is also rejected by the values of the Fstatistics reported in table 25. However, individual coefficients are in general not significantly different from zero, though their signs and magnitudes are very stable across different specifications (see table 24, and table 47 in appendix D). The sign of the coefficient associated with basic grains (c1) is positive as expected, indicating that these loans bear higher transaction costs than the miscellaneous end-uses that play the role of reference level in the estimated equations. As discussed in section 4.2, this may be a result of a higher level of loan-risk that can be associated with basic-grains farming.

Export crops appear to be seen by lenders as low-risk ventures, thus requiring less information, documents and monitoring. It was pointed out in section 4.2 that these crops are usually grown under marketing-repayment agreements between producers, banks, and processing/marketing firms. These schemes reduce production and marketing risks, and minimize the risk of default from the lender's viewpoint. The negative sign observed for  $c_2$ , the coefficient associated with export-crop loans, is consistent with the foregoing discussion. Finally, livestock loans show a positive though non-significant coefficient associated with the corresponding dummy variable for this end-use (U<sub>3</sub>). This result suggests that for lenders the perceived riskiness of this activity prevails over the fact that livestock can be used as built-in collateral in these loan operations.

### 6.2.1. Transaction Costs and Loan Size

This section draws upon the results of the basic model reported in table 24 for a presentation of the estimated relationship between transaction costs and the loan amount. The use of the basic-model results in this section, instead of the results of the extended version, makes the graphic presentation less complicated without affecting the essence of the relationship under analysis. The results obtained with the extended version are utilized extensively in the following section, to discuss the nature of the trade-off between transaction costs and the explicit interest rate.

To analyze the relationship between transaction costs and loan amount it is convenient to summarize the results of the basic model as follows:

 $T = 30.6459 \ A^{0.0001} L^{0.3388} i^{-0.9237} e^{\Sigma} j^{b} j^{D} j^{+} \Sigma k^{c} k^{U} k \quad (6.2)$ where, T is transaction costs per loan,

- A is area of the farm,
- L is the loan amount,
- i is the explicit interest rate,
- $D_{j}$  are dummy variables for loan source, j=1,2,
- $U_k$  are dummy variables for end-uses, k=1,2,3.

Equation (6.2) is evaluated at the geometric mean of area of the farm (35 hectares), and written for each loan source disregarding the effects of different loan uses to simplify the exposition (i.e., setting all  $c_k=0$ ):

(a) Development bank  $(D_1=0, D_2=0)$ ,

$$T = 30.6568 L^{0.3388} i^{-0.9237}$$
 (6.3)

(b) Private banks 
$$(D_1=1, D_2=0),$$

T = 55.1895 
$$L^{0.3388}i^{-0.9237}$$
 (6.4)  
(c) Credit unions (D<sub>1</sub>=0, D<sub>2</sub>=1),

$$T = 13.3292 L^{0.3388} i^{-0.9237}$$
 (6.5)

Note that in these equations the interest-rate variable plays the role of a shift parameter for the relationship between transaction costs and loan size. Evaluating equations (6.3) through (6.5) at the geometric mean of interest rate in the sample (13%), the corresponding functions become:

$$T = 2.3680 L^{0.3338}$$
, for the development bank, (6.6)

 $T = 5.1631 L^{0.3388}$ , for private banks, and (6.7)

 $T = 1.2470 L^{0.3388}$ , for credit unions. (6.8)These equations are portrayed in figure 1 to show the effect of different loan sources on the relationship between transaction costs and loan size. As pointed out in the previous section, private banks impose the highest transaction costs per loan for any given loan amount. Credit unions are the least expensive source of funding at given loan sizes and interest rates, and the development bank falls in between the two extremes. The typical (median) value of the loan amount for each lender is indicated in figure 1 by points C (credit unions), D (development bank), and P (private banks), along the loan-size axis. The predicted values of transaction costs per loan corresponding to each typical value of loan size can be read on the curve corresponding to each loan source.

It is interesting to analyze the importance of loan source in the transaction-costs/loan-amount relationship looking at the corresponding value of transaction costs per lempira ( $\tau$ ) in each case. Equations (6.6) through (6.8) can be easily transformed into the following:

 $\tau$  = 2.8680 L<sup>-0.6612</sup>, for the development bank,  $\tau$  = 5.1631 L<sup>-0.6612</sup>, for private banks, and  $\tau$  = 1.2470 L<sup>-0.6612</sup>, for credit unions.

These curves are presented in figure 2. In general, transaction costs per lempira borrowed are a decreasing





function of loan size. Curves are very steep for small loan amounts, becoming increasingly flatter as the loan size increases. For any given loan size, transaction costs per lempira are higher in private banks than in the development bank, where these costs are in turn higher than in credit unions. However, it is important to note again points C, D, and P in figure 2, and read the corresponding level of transaction costs per lempira on the vertical axis. For these most typical loan amounts (median values) private banks are the least-costly source of credit (0.42%) followed by credit unions (0.93%) and the development bank (1.19%). In general, given the domain of loan amounts with which different lenders operate (see table 43 in appendix D) the relevant range of transaction costs for each will lie on different sections of their corresponding curves. For credit unions the typical range is close to the vertical axis, on the steepest part of the curve, whereas for private banks the relevant range falls on the rather flat portion of the corresponding curve. A good proportion of the relevant range of transaction costs in the case of the development bank is on the steep section of the curve, although this bank's range of loan amounts implies a wider variety of slopes, as compared to the other two cases.

An important feature in the relationship between transaction costs and loan amount is the role of the explicit interest rate charged on loans. As pointed out before, the



interest-rate variable works as a shift parameter in the relationship between transaction costs and loan amount. Figure 3 shows the curves corresponding to the development bank at two different levels of the interest rate, 10% and 20%. The curves corresponding to private banks and credit unions are omitted to avoid unnecessary complication of the presentation. It can be seen in figure 3 that for any given loan size, transaction costs are reduced by an increase in the interest rate. In other words, the curve that relates transaction costs per lempira and loan amount shifts downward with an increase in the explicit interest rate. Furthermore, it is evident from figure 3 that the vertical magnitude of the shift is larger for small loans than for large loans. This implies that the decline in transaction costs due to an increase in the interest rate will be more important for small loans than for large loan amounts. For example, the shift portrayed in figure 3 associated with an increase in the interest rate from 10% to 20% represents a decrease in transaction costs per lempira of 1.2 percentage points for a loan of 2,000 lempiras. This reduction amounts to only 0.24 percentage points in the case of a loan of 20,000 lempiras.

In the following section, the foregoing results will be broken down and qualified to consider the interactions of loan size and loan source with the elasticity of transaction



Figure 3. Transaction Costs per Lempira by Loan Size, for Different Levels of the Interest Rate (i)

costs with respect to the interest rate. These modifications will affect the magnitude of the trade-off between transaction costs and the interest rate, rather than the general direction of the effects under discussion here. Therefore, it is still valid and useful to represent the trade-off suggested by the analysis of figure 3 as a relationship between transaction costs per lempira and the interest rate. In this case, shown in figure 4, the loan amount is the variable that acts as a shift parameter. The curves drawn in figure 4 correspond to two different levels of loan amount, 2,000 and 20,000 lempiras.

Figure 4 shows that for any given interest rate, transaction costs are higher for small loans than for large loans. However, figure 4 also shows that a change in the interest rate will induce a larger compensating change in the opposite direction in the level of transaction costs per lempira for small loans than for larger-sized loans. This differential effect can be seen in figure 4 by looking at the different slopes of the curves, at a point such as  $i_0$ . The steepest slope corresponds to the curve associated with the smallest loan-size. It follows from this result that an increase in the explicit interest rate will have a progressive result, since it will reduce transaction costs more for small loans than for large loans. On the other hand, a reduction in the interest rate will increase transaction costs in a larger magnitude for small loans than for



Figure 4. Trade-off Between Transaction Costs per Lempira and the Interest Rate, for Different Loan Sizes (L)

in the case of large loans. In this sense, policies that reduce interest rates will have regressive distributive consequences. Under more strict interest-rate restrictions, lenders substitute implicit charges (transaction costs) for explicit interest in such a way that transaction costs of borrowing increase more for small-loan operations than for large loans.

# 6.2.2. Trade-Off between Transaction Costs and the Interest Rate

This section utilizes the results of the transactioncosts function that includes interaction variables, to account for the effects of loan size and loan source on the elasticity of transaction costs with respect to the interest rate. The result of this extended version were presented in table 24, column 2. It is useful to recall here the expression for the elasticity between transaction costs and interest rate, including the effects of small-loan operations and of private banks as the loan source with a significantly different behavior:3/

 $\partial \ln \tau / \partial \ln i = a_{30} + a_{31}S + a_{32}D_1$  (6.9)

where,

 $\partial \ln \tau / \partial \ln i$  is the elasticity between transaction costs per lempira ( $\tau$ ) and interest rate (i)

 $<sup>\</sup>frac{3}{2}$  Overall results and statistical significance of individual interactions were discussed in the first part of section 6.2. Interactive effects of credit unions on the trade-off between  $\tau$  and i were not significant, therefore these institutions are excluded from the analysis presented here.

S is a dummy variable for loan-size category

- S = 1 if the loan amount is less than or equal to  $L_{2,000}$ ,
- S = 0 otherwise,
- $D_1$  is a dummy variable for loan source,
  - $D_1 = 1$  if the lender is a private bank,
  - $D_1 = 0$  otherwise.

Expression (6.9) and the definitions of the variables involved imply that the elasticity of  $\tau$  with respect to the interest rate will vary according to the loan source and to the loan-size category. These elasticities can be written for the different combinations of loan source and loan size as follows:

(a) development bank (D<sub>1</sub>=0),

small loans (S=1),  $\partial \ln \tau / \partial \ln i = a_{30} + a_{31}$ ,

large loans (S=0),  $\partial \ln \tau / \partial \ln i = a_{30}$ ,

(b) private banks  $(D_1=1)$ ,

small loans (S=1),  $\partial \ln \tau / \partial \ln i = a_{30} + a_{31} + a_{32}$ ,

large loans (S=O),  $\partial \ln \tau / \partial \ln i = a_{30} + a_{32}$ . The values of these different elasticities are reported in table 26, along with the values of the F-statistics for different null hypotheses about these elasticities.

These results indicate that there are significant differences between the elasticities of transaction costs with respect to the interest rate associated with different combinations of loan source and loan size. The response of

Table 26.	Transaction Costs per Lempira (7): Estimated Elasticities
	of $\tau$ with respect to Changes in the Explicit Interest Rate
	(i). Differences Between Lenders and Loan-Size Categories.

Lender/Loan-Size Category	Estimated Value of Elasticity	Null Hypotheses, <sup>H</sup> o	F-Value (conclusion)					
Development Bank								
Small Loans <mark>a</mark> / (ðlnī/ðlni = a <sub>30</sub> + a <sub>31</sub> )	-0.5551	$a_{30} + a_{31} = 0$ $a_{30} + a_{31} = -1$	5.05 (H <sub>o</sub> rejected)† 3.24 (H <sub>o</sub> rejected)°					
Large Loans <u>b</u> / (ðlnτ/ðlni = a <sub>30</sub> )	-0.8425	a <sub>30</sub> = -1	0.63 (H <sub>o</sub> not rejected)					
Private Banks								
Small Loans <mark>a</mark> / (ðlnτ/ðlni = a <sub>30</sub> + a <sub>31</sub> + a <sub>32</sub> )	-2.6692	a <sub>30</sub> + a <sub>31</sub> + a <sub>32</sub> = -1	4.18 (H <sub>o</sub> rejected)†					
Large Loans <mark>b/</mark> (ðlnτ/ðlni = a <sub>30</sub> + a <sub>32</sub> )	-2.9566	a <sub>30</sub> + a <sub>32</sub> = -1	6.16 (H <sub>o</sub> rejected)*					

Source: Results presented in table 6.5, column 2 (extended version of transaction-costs function). Significance levels: \* .01 t .05 ° .10 a/ Less than or equal to L.2,000 b/ Greater than L. 2,000

borrowing transaction costs to changes in the interest rate is very elastic in the case of private banks. For these loans, the absolute value of the elasticity is between three and five times as large as the values obtained for the development bank. In the latter, large loans show a unitary elasticity whereas for small loans the magnitude of the response of transaction costs to changes in the interest rate is inelastic (less than zero, greater than -1).

For small-loan operations (less than 2,000 lempiras), the absolute value of the elasticity is lower than the values obtained for loans of medium-to-large size (over 2,000 lempiras), denoting a less flexible response to changes in the interest rate in this class of loan operations. This result could be expected, on the basis that usually these smaller loans are associated with more rigid and cumbersome targeting schemes. These schemes prevent the modification of loan procedures when a more flexible interest rate environment would allow the substitution of explicit interest for implicit charges (transaction costs), thus reducing the level of borrower's transaction costs.

At this point, it is useful to summarize the results obtained with the transaction-costs function presented so far: --

 (a) loan amount, interest rate, and loan source are significant determinants of the level of transaction costs.
 The results obtained for loan use are ambiguous, and farm area is not a significant factor in the transaction-costs function;

- (b) transaction costs per loan:
  - increase with the loan amount, at a decreasing rate,
  - decline with increases in the interest rate
    (i.e., the trade-off relationship),
  - are higher for private-bank loans than for development-bank loans, at given loan-sizes and interest rates,
  - are lower for credit unions than for the development bank, at given loan sizes and interest rates;
- (c) transaction costs per lempira are a decreasing function of loan amount. All other relationships follow the same pattern indicated above for transaction costs per loan;
- (d) the trade-off (negative elasticity) between transaction costs and interest rate exists and is significant in all cases. Its magnitude is affected by small-loan operations, and by the source of the loan.

A simulation exercise in which all the foregoing effects can be observed is presented in table 27. This table shows the estimated values of transaction costs per lempira, and total borrowing costs (interest rate plus transaction costs) for different combinations of loan sources and hypothetic loan-sizes. These estimates are obtained using the results of the extended version of the

				-	
Lend	er/Loan Size	Estimated Value of Transaction Costs per Lempira î, %	Estimated Total Borrowing Costs (i+î), %	Change in τ with a one-point increase in the interest rate δτ/δi, pct. points	Change in Total Borrowing Costs with a one-point increase in the interest rate ð(i+τ)/ði, pct. points
Deve	lopment Bank				
	Lps. 2,000	2.85	15.85	-0.123	0.877
	Lps. 5,300 <u>a</u> /	0.80	13.80	-0.052	0.948
	Lps. 20,000	0.38	13.38	-0.025	0.975
Priva	ate Banks				
	Lps. 2,000	5.77	18.77	-1.184	-0.184
	Lps. 5,300 <u>a</u> /	1.61	14.61	-0.366	0.634
	Lps. 20,000	0.77	13.77	-0.175	0.825

Table 27. Transaction Costs per Lempira  $(\tau)$ : Estimated Values of  $\tau$ , Changes in  $\tau$  and in Total Borrowing Costs  $(i+\tau)$  with Increases in the Explicit Interest Rate (i), by Lender and Loan Size.

Source: Extended-model results presented in table 24. Estimates evaluated at geometric means of farm area (35 Has.) and interest rate (13%). Differences by end-use are not considered.

a/ Median value of loan size in the overall sample.

٠

transaction-costs function (see table 24, column 2). Also, the change in transaction costs per lempira, and in total borrowing costs predicted as a result of a one-point increase in the interest rate are reported in table 27. These changes are computed using the expression:

 $\partial \tau / \partial i = (\partial \ln \tau / \partial \ln i) (\hat{\tau} / i)$ 

where,  $\hat{\tau}$  denotes the estimated value of transaction costs per lempira, and the elasticity ( $\partial \ln \tau / \partial \ln i$ ) has already been defined for the different combinations of loan source and loan size (see table 26).

The potential effects of loan use on the behavior of transaction costs are not considered in the results presented in table 27. However, these results are a valid representation of the average of end-uses. Tables 48 and 49 in appendix D present a similar exercise that takes into account different loan-uses, even though these results should be taken with some caution since the individual coefficients for different end-uses were not significantly different from zero in the estimated equations (see table 24).

The figures presented in table 27 show that, for any given loan size, transaction costs of borrowing from private banks are about twice as high as those associated with development-bank loans (see column 1 in table 27). For each type of lender, transaction costs per lempira borrowed vary substantially by loan size. The larger the loan size, the lower the level of transaction costs per lempira. At a

given level of interest rate (13% in table 27), total borrowing costs are also higher in private-bank loans than in the case of loans granted by the development bank (see column 2 in table 27). However, this difference across institutions becomes less important as the loan size increases. These results indicate that under interest-rate restrictions, private banks tend to be more efficient in passing-on intermediation costs to the borrower, than the development-bank.

At the same time, the response of transaction costs per lempira to changes in the interest rate is considerably larger in the case of loans from private banks than in development-bank loans (column 3 in table 27). A one-point increase in the explicit interest rate will lead to a larger compensatory decline in borrowing-transaction costs in private-bank loans than in the case of loans from the development bank. The simulation with a loan size of 2,000 lempiras shows that this compensating decline could morethan-offset the increase in the interest rate in the case of the private bank, thus resulting in a decrease in total borrowing costs. In general, the compensatory change in transaction costs that occurs as a result of a change in the explicit interest rate is larger for small loans than for large loans.

The differences in the level of transaction costs and the response of these to changes in the interest rate, for

different loan-sizes and loan sources, are represented in figure 5. In this figure, the functional relationships underlying the results presented in table 27 are graphed to allow more general comparisons between loan sources and loan sizes. It is clear from figure 5 that, for any given level of the interest rate, transaction costs are higher for small loans than for large loans. Also, for any given loan size, transaction costs decrease as the interest rate increases. Finally, for given levels of interest rate and loan size, loans from private banks entail higher transaction costs than those granted by the development bank. However, transaction costs of private-banks loans are more responsive to changes in the interest rate than are these costs associated with the development-bank loans.

Summarizing the foregoing results, changes in the explicit interest rate will have a differential effect on transaction costs, depending on the source of the loan and on the loan size. Column 4 in table 27 shows that a onepoint increase in the interest rate will be almost fully translated into a corresponding increase in total borrowing costs in the case of large loans. This effect will be reduced for smaller loans due to the offsetting response of transaction costs to changes in the interest rate. This compensating effect is more important in the case of private-bank loans. Since this offsetting effect occurs for increases as well as decreases in the level of the interest



Figure 5. Trade-off Between Transaction Costs per Lempira and the Interest Rate, for The Development Bank and Private Banks, by Different Loan Sizes (L)

rate, it follows that further reductions in the interest rate on loans will benefit primarily borrowers of large amounts, instead of farmers borrowing small loan-amounts. In this sense, cheap-credit policies will not attain their intended distributional goals in the rural sector.

The main implication of the results discussed above is that interest-rate regulations will not be effective. Lenders will use the loan-application procedure, with its associated set of requirements and documentation to impose transaction costs on borrowers, in direct proportion to the perceived risk of the loan operations. It has been estimated that these costs range from negligible amounts (0.4%) to almost 6%, depending on the size of the loan and the loan source, and given the sample mean of explicit interest rate (13%). It has also been shown that this range will shift upward under further restrictions in the explicit interest rate. Therefore, interest-rate restrictions generate implicit-price adjustments in the credit market rather than the quantity adjustments predicted in earlier studies [31]. Transaction costs (i.e., implicit pricing), instead of quantity restrictions, play the role of rationing mechanism under the conditions prevailing in the Honduran financial system.

## CHAPTER VII Summary and Conclusions

This study has investigated the costs involved in financial intermediation in a small lesser developed economy. Aside from the explicit costs of finance, interest rates, all participants in financial markets use real resources when performing their roles in financial intermediation: savings mobilization, lending, and borrowing. The non-interest transaction costs borne by financial intermediaries and borrowers, have been the focus of this study. Financial regulation, and regulation-avoidance were additional major themes in this work. The effects of financial regulation on transaction costs and the mechanisms of regulatory circumvention by financial intermediaries were also addressed. The study was carried out in Honduras, where about 60% of the population live in the rural areas, and agriculture contributes about 30% of GDP and accounts for about two-thirds of export earnings during the period (1970-82). At the same time, agriculture receives about one-fourth of the total credit of the banking system.

Financial-intermediation issues dealt with in this study were classified into two areas. First, the magnitude

of intermediation costs accruing to lenders and borrowers were measured and various hypotheses were tested concerning the response of these costs to different variables. Second, important policy issues associated with financial regulation were addressed, particularly the effects that regulatory policies have had on the costs of financial intermediation. Among these, interest-rate regulations and selective credit policies emphasizing agricultural credit received major attention.

Financial-intermediaries' cost functions were the basis for measuring and analyzing the costs incurred by these institutions. Several characteristics of the underlying technology of banks were assessed through these costfunction estimates: scale economies, economies of scope (cost complementarities), elasticities of factor substitution and factor demand. In the analysis, a number of methodological issues surrounding cost studies in banking were addressed. Among these were the definition of bank output, the choice of functional forms for the cost function, and the selection of an estimation procedure for the specified functions. Evaluating the effects of financial regulations posed an additional challenge of defining appropriate proxyvariables for these regulations, which are usually difficult to measure directly (e.g., loan-targeting strategies).

On the borrowers' side, a transaction-costs function was conceptualized within the framework of credit rationing

under loan-pricing constraints. Transaction costs for borrowers arise from the use of the loan-application procedure established by lenders as a discriminatory device that substitutes for explicit pricing, when explicit interest rates are restricted. The limited range of administered loan-rates, and several risk related characteristics of loan operations entered the transaction-costs function as explanatory variables for loan rationing behavior.

Throughout the study, the contrasting performance was analyzed between the National Agricultural Development Bank of Honduras (BANADESA), and the largest private commercial bank in the country. The estimation and analysis of the cost functions of these financial institutions were based on branch-level data of the two banks. Finally, the borrowers' transaction-costs function was estimated using data from a field-survey of agricultural borrowers undertaken in several regions of the country. These borrowers held active loans from different financial institutions including the development bank, several private banks, and rural credit unions.

#### Results and Implications

#### a) Lenders' Costs

Estimation of lenders' intermediation costs indicated that these costs are considerably higher than is usually assumed. At the same time, there were important differences between the costs of deposit-mobilization and those associated with lending. Also, the findings revealed notorious

contrasts across institutions (i.e. the development bank versus the private bank). All results summarized below are based on translogarithmic cost functions, estimated by seemingly unrelated regressions (GLS), with bank outputs defined as the value of loans and deposit balances.

For the development bank, over 70 percent of intermediation costs corresponded to lending, whereas less than 30 percent were attributed to the administration of deposit accounts. The opposite is true for the private bank, where only 28 percent of the costs were associated with lending, while 72 percent were related to deposit mobilization. This acute contrast reflects the development-bank's greater reliance on foreign funds and special rediscount lines from the central bank, as compared to the private bank which relied more heavily upon financial resources mobilized from the general public. This contrast in the composition of the banks' liabilities is reflected in the allocation of resources in each bank, and therefore in the share of lending and deposit-mobilization activities in total intermediation costs.

The average costs of lending in the development bank (10%) were almost three times as high as those estimated for the private bank (3.4%). The marginal costs of lending were 4.5 times larger in the development bank than in the private bank (7.6% <u>versus</u> 1.7%). This again reflects the differences in the sources of funds with which each bank operates.

The greater reliance on external loan sources by the development bank created additional costs of compliance with loan-targeting requirements imposed by foreign donors or the government. These in turn forced the institution to maintain a more centralized operation, and a heavier incidence of supervisory and record-keeping resources than would have been the case in the absence of these targeting requirements. The results obtained for the development bank indicate that the usual administrative margins contemplated by foreign donors for special credit projects (3 or 4%) are unrealistically low, and compromise the financial viability of lending institutions participating in these "targeted" programs.

The average costs of mobilizing deposits were also higher in the development bank compared to the private bank (8.8% <u>versus</u> 5.3%). On the other hand the marginal costs of deposit-mobilization showed the opposite pattern. These were lower in the development bank (2.7%) than in the private bank (6.7%). Furthermore, given the fact that the marginal cost of deposit mobilization was higher than the average costs in the private bank (6.7% <u>versus</u> 5.3%), deposit mobilization in this institution had reached a point of decreasing returns. Further expansion of the depositmobilization activity in the private bank is an unattractive option, unless this expansion relies upon larger-sized average deposit balances. In sharp contrast, the results

for the development bank highlight the existence of excess capacity unexploited for deposit mobilization, i.e. the marginal costs of deposit mobilization (2.7%) were well below the average costs (8.8%).

Overall, intermediation costs were higher in the development bank than in the private bank. However, the difference between the average costs was larger than the difference between the marginal costs of intermediation. The latter were only two percentage points higher in the development bank than in the private bank, thus indicating a similar degree of economic efficiency in terms of marginal cost criteria. Marginal cost criteria are frequently used as efficient pricing strategies. However, in this case any policy emphasizing marginal cost pricing would represent large operational losses for the development bank (given its much higher average costs), whereas in the case of the private bank this policy would imply an almost break-even situation.

#### Evidence on Scale and Scope Economies

Scale-economies estimates for both banks were not significantly different from one, even though the estimated levels (i.e. the elasticity of costs to increases in output) were consistently lower in the case of the development bank. These results of non-significant scale economies were not surprising, considering the small size of financial markets in low-income countries. An important finding was the substantial difference in the separate effects of the expansion of different outputs on intermediation costs. For the development bank, there were important economies of scale to the expansion of deposit-mobilization activities, whereas lending activities were approaching constant returns-toscale for the average-branch case. The opposite is found for the private bank, where the expansion of lending activities showed cost advantages as compared to deposit mobilization. These different cost effects of different outputs indicate that both banks could benefit from "economies of scale" by engaging in unbalanced output expansion. Expansion strategies for the development bank should emphasize deposit mobilization over lending activities, whereas the private bank's expansion should be biased towards lending operations.

Both banks show cost complementarities (economies of scope) associated with the joint production of loans and deposits. This finding argues against the strategy of creating a specialized lending institution with no depositmobilization functions. The joint provision of deposit services will not only improve the financial viability of the institution and promote financial savings, but also will reduce the marginal costs of lending through costcomplementarity effects.

#### Interest-Rate Restrictions

Interest-rate restrictions that reduced the level of real deposit rates or real lending rates had cost-increasing effects on financial intermediaries. This strongly suggests that increases in the level of real rates of interest would generate reductions in total intermediation costs. This trade-off between real rates and costs reflects the costs of regulatory-avoidance. Restrictions on deposit-rates force financial intermediaries to offer non-interest rewards to depositors in order to at least maintain their deposit balances. This is particularly important in the Honduran case, where real deposit-rates have been low (usually negative) and unstable, thus discouraging the holding of financial savings. The provision of free banking services or preferential treatment in loan contracts to selected clients generate additional costs of deposit-mobilization for the institution, that could be avoided if explicit interest compensation could be paid to depositors.

On the other hand, multiple ceilings on lending rates constrained the ability of financial intermediaries to discriminate between potential borrowers in Honduras. In response to these constraints, financial institutions created rationing mechanisms of more complicated loan procedures that substituted for a more flexible interest-rate environment. This, in effect, passed on to borrowers a substantial part of the costs of intermediation. However, a proportion of these additional costs had to be borne by the lender, and this was reflected in the inverse relationship estimated between real lending-rate ceilings and lenders' intermediation costs.

Loan targeting was found to be a cost-increasing factor affecting the development bank. Both foreign-funded projects and central-bank rediscount lines created additional intermediation costs, due to the increased resources that the bank devoted to accounting, monitoring, record-keeping and reporting, in order to comply with the requirements of targeted programs.

An additional finding in the case of the development bank was a direct relationship between the delinquency rate and intermediation costs. Increases in delinquency created a lagged cost-increasing effect due to additional resources allocated to loan-recovery, loan-evaluation and monitoring. Despite these efforts, high delinquency and increasingly older delinquent balances continued to characterize the loan portfolio of the development bank.

#### Borrowing Costs: Magnitudes and Incidence

The effectiveness of regulatory circumvention by financial intermediaries was evident from the results obtained in the analysis of borrowing transaction costs. Borrower's transaction costs were important and amounted, on average, to 3 percentage points above and beyond the explicit interest rate charged on loans. At the sample mean of this

explicit rate (13%) transaction costs of borrowing ranged between very small amounts (0.4%) to almost 6%, depending primarily on the size of the loan, and the loan source. Transaction costs per lempira borrowed were a decreasing function of the loan amount, and varied depending on the loan source. Thus a rather homogeneous set of explicit interest rates resulted in a wide range of total borrowing costs, that penalized those loans perceived by lenders as riskier or costlier than average, and favored preferred customers borrowing larger sized loans.

The most interesting finding of the borrowers' cost analysis was the existence of a trade-off (i.e. an inverse relationship) between transaction costs and the explicit interest-rate charged on loans. This negative elasticity implied that increases in the explicit interest rate would be partially offset by a compensating change in transaction costs to the borrower. In short, under a more flexible interest-rate environment, lenders would substitute explicit interest charges for implicit pricing through transaction costs. This trade-off between transaction costs and the interest rate is more important for small loans than large loans. A change in the explicit interest rate induced a larger compensating change in the opposite direction in the level of transaction costs for small loans than for largersized loans. It follows from this result that, contrary to conventional wisdom, an increase in the explicit interest

rate on loans would have a progressive impact, since it would reduce transaction costs more for small loans than for large loans.

Borrowing from private banks entailed, in general, higher transaction costs than borrowing from the development bank. On the other hand, the elasticity of transaction costs with respect to changes in the explicit interest rate was larger for private banks than for the development bank. These results indicate that private banks were more effective in passing on intermediation costs to borrowers. At the same time, these banks were more responsive to changes in the economic and regulatory environment, and were more flexible in adjusting their loan procedures and requirements.

In summary, total intermediation costs including costs borne by lenders and borrowers were estimated, on average, to be about 20 percent in the case of the development bank, and approximately 11 percent in the case of the private bank in Honduras. These costs, however, had a wide range of variation depending upon the average size of depositbalances, the average size of loans and, more importantly, on the nature and level of existing regulations affecting financial intermediaries.

#### Conclusions

Financial intermediation costs in Honduras are substantial and vary widely depending on the conditions under which financial intermediaries and borrowers operate. These transaction costs associated with deposit-mobilization, lending, and borrowing were two or three times the level of the deposit rates of interest received by savers. Part of the intermediaries' costs were explained by various forms of non-interest compensation paid for financial savings by intermediaries. In general, however, a good proportion of total intermediation costs have been introduced into the financial system through the impact of financial regulations.

An important implication of the results discussed here is that financial market regulations will not be effective, or their effects will be distorted by the regulatorycircumvention response of financial intermediaries. The only certain effect of regulation was the increase of intermediation costs. Financial intermediaries will respond to deposit-rate ceilings through non-interest rewards to depositors. At the same time, they will counteract lending-rate ceilings through implicit pricing. In fact, this study confirms that transaction costs (implicit pricing), rather than quantity restrictions, become the principal rationing mechanism under interest-rate restrictions in Honduras.

Only when the overall intensity of financial regulations restrict the ability of financial intermediaries to implicitly compensate depositors, will the total price received by savers decrease. This may have been one of the

reasons underlying the decline in financial activities observed in Honduras after 1978.

Policy-makers should consider both the real effectiveness and the costs of financial regulations when evaluating policy measures. The Honduran experience strongly suggests that for many of these measures costs will offset benefits, due to the effects of regulatory circumvention. The development bank analyzed here provides a good example of the cost-increasing effects of creating a specialized institution to deal with agriculture. The usual social benefits that may be argued in support of this institution should be weighed against the less widely recognized social costs of maintaining and subsidizing these costly operations. In the end, taxpayers are providing the resources to cover the bank's operational losses, thus distributional effects attributed to the institution should be adjusted accordingly.

Transaction costs are a measure of the "friction" existing in the functioning of financial markets. The higher the costs of intermediation, the less efficient the performance of the financial sector in resource allocation and distribution. This study has shown that there are several ways in which transaction costs can be reduced, thus reducing the friction and improving financial-markets performance. Financial reforms that provide a more flexible

interest-rate environment and reduce the cost-increasing burden of targeting schemes should greatly benefit the overall performance of the Honduran financial system. Maintaining the present set of financial regulations and targeting requirements will reduce potential resource mobilization within the Honduran financial sector and only add to the real costs of financial intermediation.
### BIBLIOGRAPHY

- Adams, Dale W, "Effects of Finance on Rural Development", <u>Colloquium on Rural Finance</u>, World Bank, Washington, D.C., 1981.
- Adams, Dale W, and Douglas H. Graham, "A Critique of Traditional Agricultural Credit Projects and Policies", Journal of Development Economics, 8 (June 1981), 347-366.
- 3. Adams, Dale W, and Gerald J. Nehman, "Borrowing Costs and the Demand for Rural Credit", <u>The Journal of</u> Development Studies, 15 (January 1979), 165-176.
- 4. Adams, Dale W, and Alfredo A. Pablo Romero, "Group Lending to the Rural Poor in the Dominican Republic: A Stunted Innovation", <u>Canadian Journal of Agricultural</u> <u>Economics</u>, 29 (July 1981), 217-244.
- 5. Adams, Dale W, Douglas H. Graham, and J. D. Von Pischke, editors, Why Cheap Credit Undermines Rural Development, (forthcoming), World Bank: Johns Hopkins University Press.
- Azzi, Corry F., and James C. Cox, "A Theory and Test of Credit Rationing: Comment", <u>American Economic Review</u>, 66 (December 1976), 911-917.
- Baltensperger, Ernst, "Credit Rationing, Issues and Questions", Journal of Money, Credit and Banking, 10 (May 1978), 170-183.
- Baltensperger, Ernst, "Alternative Approaches to the Theory of the Banking Firm", Journal of Monetary Economics, 6 (January 1980), 1-37.
- Banco Central de Honduras, <u>Boletin Estadistico</u> (Monthly Publication). Tegucigalpa, Honduras, several issues, 1970-1983.
- Banco Central de Honduras, <u>Cuentas Nacionales de</u> <u>Honduras, 1970-1980</u>. Tegucigalpa, Honduras, November 1982.

- 11. Banco Central de Honduras, <u>Indice General de Precios al</u> <u>Consumidor</u>. Tegucigalpa, Honduras, November, 1982.
- 12. Benston, George J., "Commercial Banking Price Discrimination Against Small Loans: An Empirical Study", Journal of Finance, 19 (December 1964), 631-643.
- Benston, George J., "Economies of Scale of Financial Institutions", Journal of Money, Credit and Banking, 4 (May 1972), 321-341.
- 14. Benston, George J., and Clifford W. Smith, Jr., "A Transactions Cost Approach to the Theory of Financial Intermediation", <u>Journal of Finance</u>, 31 (May 1976), 215-231.
- 15. Benston, George J., Gerald Hanweck, and David Humphrey, "Scale Economies in Banking: A Re-Examination", Board of Governors of the Federal Reserve System (mimeographed, October 1980).
- 16. Benston, George J., Gerald A. Hanweck, and David B. Humphrey, "Operating Costs in Commercial Banking", <u>Economic Review</u>, Federal Research Bank of Atlanta, 67 (November 1982).
- 17. Benston, George J., Gerald A. Hanweck, and David B. Humphrey, "Scale Economies in Banking. A Restructuring and Reassessment", Journal of Money, Credit and Banking, 14 (November 1982), 435-456.
- 18. Benston, George J., Allen N. Berger, Gerald A. Hanweck, and David B. Humphrey, "Economies of Scale and Scope in Banking", Board of Governors of the Federal Reserve System (June, 1983).
- 19. Berndt, Ernst R., and Laurits R. Christensen, "The Translog Function and the Substitution of Equipment, Structures, and Labor in U.S. Manufacturing 1929-68", Journal of Econometrics, 1 (March 1973), 81-114.
- 20. Bhatt, V.V., "Development Banks in the Financial System", Economic Development Institute, <u>International</u> <u>Bank for Reconstruction and Development</u>, Seminar Paper No. 15, 1979.
- 21. Binswanger, Hans P., "A Cost Function Approach to the Measurement of Factor Demand and Elasticities of Substitution", American Journal of Agricultural <u>Economics</u>, 56 (May 1974), 377-386.

- 22. Bottomley, Anthony, "Interest-Rate Determination in Underdeveloped Rural Areas", <u>American Journal of</u> Agricultural Economics, 57 (May 1975), 279-291.
- 23. Bourne, Compton, and Douglas H. Graham, "Problems with Specialized Agricultural Lenders", <u>Colloquium on Rural</u> Finance, World Bank, Washington, D.C., 1981.
- 24. Brito, Ney O., and Ricardo Fernandes Franco, "Retornos de Escala em Bancos Comerciais: A Experiencia Brasileira de 1978/1979", <u>Revista Brasileira de Mercado</u> <u>de Capitais</u>, 7 (Sept./Dec. 1981).
- 25. Christensen, L. R., D. W. Jorgenson, and L. J. Lau, "Trascendental Logarithmic Production Frontiers", <u>Review</u> of Economics and Statistics, 55 (February 1973), 28-45.
- 26. Cuevas, Carlos E., and Douglas H. Graham, "Agricultural Lending Costs in Honduras", in Adams, D., Graham, D., and Von Pischke, J. D., editors, <u>Undermining Rural</u> <u>Development with Cheap Credit</u>, Westview Press, Boulder, Colorado, forthcoming.
- 27. Cuevas, Carlos E., and Robert C. Vogel, "Desarrollo del Sistema Financiero y Movilización de Ahorros: La Experiencia de Honduras en el Período 1970-1982", ESO 1020, Ohio State University, April 1983.
- 28. DeJanvry, Alain, "The Generalized Power Production Function", American Journal of Agricultural Economics, 54 (May 1972), 234-237.
- 29. Diamond, William, and V.S. Raghavan, editors, <u>Aspects</u> of <u>Development Bank Management</u>. Baltimore: Johns Hopkins, 1982.
- 30. Gheen, William E., "Development Banks: Empirical Derivation of the Loan Administration Cost Curve". Unpublished Ph.D. Dissertation, University of Maryland, Maryland, 1976.
- 31. Gonzalez-Vega, Claudio, "On the Iron Law of Interest Rates Restrictions, Agricultural Credit Policies in Costa Rica and in Other Less Developed Countries". Unpublished Ph.D. Dissertation, Stanford University, California, 1976.
- 32. Gonzalez-Vega, Claudio, "The Honduran Economy", and "The Financial System of Honduras", in Graham, Douglas et al., An Assessment of Rural Financial Markets in Honduras. A Report to USAID-Honduras, Ohio State University, December 1981.

- 33. Gonzalez-Vega, Claudio, "Estabilidad Macroeconómica y El Sistema Financiero: El Caso de Honduras", ESO 1021, Ohio State University, April 1983.
- 34. Gordon, David L., "Development Finance Companies, State and Privately Owned", World Bank Staff Working Papers No. 578, 1983.
- 35. Graddy, Duane B., and Reuben Kyle, III, "The Simultaneity of Bank Decision-Making, Market Structure, and Bank Performance", <u>Journal of Finance</u>, 34 (March 1979), 1-18.
- 36. Graham, Douglas H., <u>et al</u>, "An Assessment of Rural Financial Markets in Honduras", A Report to USAID-Honduras, Ohio State University, December 1981.
- 37. Graham, Douglas H., and Carlos E. Cuevas, "Lending Costs and Rural Development: Is Cheap Credit Really Cheap?", <u>Savings and Development</u>, 8 (forthcoming, 1984).
- 38. Henderson, James M., and Richard E. Quandt, <u>Microeconomic Theory, A Mathematical Approach</u>. New York: McGraw-Hill, 1980.
- 39. Hodgman, Donald R., "Selective Credit Controls", Journal of Money, Credit and Banking, 4 (May 1972), 342-359.
- 40. Hoyos, Miguel, "Borrowing Costs and Credit Rationing for Three Credit Unions in Eastern Bolivia". M.S. Thesis, Arizona State University, 1982.
- 41. Inter-American Development Bank, "The Role of National Development Finance Institutions in the Mobilization of Domestic Resources in Latin America", XIII Meeting of Technicians of Central Banks of the American Continent, Ottawa, Canada, 1976.
- 42. International Monetary Fund, <u>International Financial</u> <u>Statistics</u>, several issues.
- 43. Intriligator, Michael D., <u>Econometric Models</u>, <u>Techniques</u>, and <u>Applications</u>. New Jersey: Prentice Hall, 1978.
- 44. Iqbal, Faroukh, "The Demand and Supply of Funds Among Agricultural Households", manuscript, The Rand Corporation, 1982.

- 45. Jaffee, Dwight M., <u>Credit Rationing and the Commercial</u> Loan Market. New York: John Wiley and Sons, 1971.
- 46. Jaffee, Dwight M., "A Theory and Test of Credit Rationing: Further Notes", <u>American Economic Review</u>, 62 (June 1972), 484-488.
- 47. Jaffee, Dwight M., and Franco Modigliani, "A Theory and Test of Credit Rationing", <u>American Economic Review</u>, 59 (December 1969), 850-872.
- 48. Jaffee, Dwight M., and Franco Modigliani, "A Theory and Test of Credit Rationing: Reply", <u>American Economic</u> <u>Review</u>, 66 (December 1976), 918-920.
- 49. Jaffee, Dwight M., and T. Russell, "Imperfect Information, Uncertainty, and Credit Rationing", <u>Quarterly Journal of Economics</u>, 90 (November 1976), 651-666.
- 50. Johnson, O.E.G., "Credit Controls as Instruments of Development Policy in the Light of Economic Theory", Journal of Money, Credit and Banking, 6 (February 1974), 85-99.
- 51. Johnston, J., <u>Econometric Methods</u>, McGraw-Hill, New York, 1963.
- 52. Kane, Edward J., "Good Intentions and Unintended Evil: The Case Against Selective Credit Allocation", Journal of Money, Credit and Banking, 9 (February 1977), 55-69.
- 53. Kane, Edward J., "Accelerating Inflation, Technological Innovation, and the Decreasing Effectiveness of Banking Regulation", <u>Journal of Finance</u>, 36 (May 1981), 355-367.
- 54. Kane, Edward J., "Political Economy of Subsidizing Agricultural Credit in Developing Countries", <u>Colloquium on Rural Finance</u>, World Bank, Washington, D.C., September 1981.
- 55. Kane, Edward J., and Burton G. Malkiel, "Bank Portfolio Allocation, Deposit Variability, and the Availability Doctrine", <u>Quarterly Journal of Economics</u>, 79 (February 1965), 113-134.
- 56. Khatkathe, Deena, "National and International Aspects of Financial Policies in LDCs: A Prologue", <u>World</u> <u>Development</u>, 10 (September 1982), 689-694.

- 57. Khatkathe, Deena, and Delano P. Villanueva, "Operation of Selective Credit Policies in Less Developed Countries: Certain Critical Issues", <u>World</u> Development, 6 (October 1978), 979-990.
- 58. Klein, Michael, "A Theory of the Banking Firm", Journal of Money, Credit and Banking, 3 (May 1971), 204-218.
- 59. Ladman, Jerry R., "Loan Transaction Costs, Credit Rationing and Market Structure: The Case of Bolivia", <u>Colloquium on Rural Finance</u>, World Bank, Washington, D.C. (September 1981).
- 60. Ladman, Jerry R., and Ronald L. Tinnermeier, "The Political Economy of Agricultural Credit: The Case of Bolivia", in Von Pischke, J. D., Dale W Adams, and Gordon Donald, editors, <u>Rural Financial Markets in</u> <u>Developing Countries</u>. John Hopkins University Press, Baltimore, 1983.
- 61. Laitinen, Kenneth, <u>A Theory of the Multiproduct Firm</u>. North-Holland, New York, 1980.
- 62. Lamberte, Mario B., "An Analysis of the Behavior of Commercial Banks", Staff Paper Series, Philippine . Institute for Development Studies, 1982.
- 63. Leamer, Edward E., <u>Specification Searches</u>. New York: John Wiley and Sons, 1978.
- 64. Lee, Warren F., and Chester B. Baker, "Market Risks and Lender Behavior", in Peter Barry, editor, <u>Risk and</u> <u>Management in Agriculture</u>. Ames, Iowa: Iowa State University Press (forthcoming).
- 65. Long, Millard, "A Note on Financial Theory and Economic Development", in Von Pischke, J. D., Dale W Adams, and Gordon Donald, editors, <u>Rural Financial Markets in</u> <u>Developing Countries</u>. Johns Hopkins University Press, Baltimore, 1983.
- 66. McKinnon, Ronald I., <u>Money and Capital in Economic</u> <u>Development</u>. Washington, D.C.: The Brookings Institution, 1973.

-

67. Mittlehammer, Ron C., Scott C. Matulich, and D. Bushaw, "On Implicit Forms of Multiproduct-Multifactor Production Functions", <u>American Journal of Agricultural</u> <u>Economics</u>, 63 (February 1981), 164-168.

- 68. Mullineaux, Donald J., "Economies of Scale and Organizational Efficiency in Banking: A Profit-Function Approach", <u>Journal of Finance</u>, 33 (March 1978), 259-280.
- 69. Murray, John D., and Robert W. White, "Economies of Scale and Economies of Scope in Multiproduct Financial Institutions: A Study of British Columbia Credit Unions", Journal of Finance, 38 (June 1983), 887-902.
- 70. Neter, John, and William Wasserman, Applied Linear Statistical Models. Richard D. Irwin, Homewood, Illinois, 1974.
- 71. Nyanin, Ohene O., "Costs of Agricultural Lending, Institutional Viability, and Lender Behavior in Jamaica". Unpublished Ph.D. Dissertation, the Ohio State University, Ohio, 1982.
- 72. Panzar, John C., and Robert D. Willig, "Economies of Scale in Multi-Output Production", <u>Quarterly Journal of</u> <u>Economics</u>, (August 1977), 481-493.
- 73. Panzar, John C., and Robert D. Willig, "Economies of Scope", <u>American Economic Review</u>, 71 (May 1981), 268-272.
- 74. Patrick, Hugh T., "Financial Development and Economic Growth in Underdeveloped Countries", <u>Economic</u> <u>Development and Cultural Change</u>, 14 (January 1966), 174-177.
- 75. Pollard, Stephen, Douglas Graham, and Carlos Cuevas, "Coffee and Basic Grains: Review of Sectoral Performance, Pricing and Marketing Margins, and Recent Policy Changes". A Report to USAID-Honduras, Ohio State University, January 1984.
- 76. Pyndick, Robert S., and Daniel L. Rubinfeld, <u>Econometric Models and Economic Forecasts</u>. New York: McGraw-Hill, 1981.
- 77. Rao, D.C., "Selective Credit Policy: Is It Justified and Can It Work?", <u>Journal of Finance</u>, 27 (May 1972), 473-479.
- 78. Ray, Subhash C., "A Translog Cost Function Analysis of U.S. Agriculture, 1939-77", <u>American Journal of</u> <u>Agricultural Economics</u>, 64 (August 1982), 490-498.

- 79. Saito, Katrine A., and Delano P. Villanueva, "Transaction Costs of Credit to the Small-Scale Sector in the Philippines", Economic Development and Cultural Change, 29 (April 1981), 631-640.
- 80. Sealey, C. W. Jr., "Credit Rationing in the Commercial Loan Market: Estimates of a Structural Model Under Conditions of Disequilibrium", <u>Journal of Finance</u>, 34 (June 1979), 689-701.
- 81. Sealey, C. W. Jr., "Deposit Rate-Setting, Risk Aversion, and the Theory of Depository Financial Intermediaries", <u>Journal of Finance</u>, 35 (December 1980), 1139-1154.
- 82. Sealey, C. W. Jr., and J. T. Lindley, "Inputs, Outputs, and a Theory of Production and Cost at Depository Financial Institutions", Journal of Finance, 32 (September 1977), 1251-1266.
- Shaw, Edward S., <u>Financial Deepening in Economic</u> Development. New York: Oxford University Press, 1973.
- 84. Shephard, Ronald W., <u>Theory of Cost and Production</u> <u>Functions</u>. Princeton University Press, Princeton, N.J., 1970.
- 85. Shumway, C. Richard, Rubin D. Pope, and Elizabeth K. Nash, "Allocatable Fixed Inputs and Jointness in Agricultural Production: Implications for Economic Modeling", <u>American Journal of Agricultural Economics</u>, 66 (February 1934), 72-78.
- 86. Silber, William L., "Selective Credit Policies: A Survey", <u>Banca Nazionale del Lavoro Quarterly Review</u>, 104-107 (December 1973), 328-351.
- 87. Smith, Vernon L., "A Theory and Test of Credit Rationing: Some Generalizations", <u>American Economic</u> Review, 62 (June 1972), 477-483.
- 88. Stiglitz, Joseph E., and Andrew Weiss, "Credit Rationing in Markets with Imperfect Information", <u>American Economic Review</u>, 71 (June 1981), 393-410.
- 89. Uzawa, Hirofumi, "Production Functions with Constant Elasticities of Substitution", <u>Review of Economic</u> <u>Studies</u>, 29 (October 1962), 291-299.
- 90. Uzawa, Hirofumi, "Duality Principles in the Theory of Cost and Production", <u>International Economic Review</u>, 5 (May 1964), 216-220.

- 91. Varian, Hal R., <u>Microeconomic Analysis</u>, W. W. Norton and Co., New York, 1978.
- 92. Von Pischke, J. D., "The Pitfalls of Specialized Farm Credit Institutions in Low-Income Countries", in Von Pischke, J. D., Dale W Adams, and Gordon Donald, editors, <u>Rural Financial Markets in Developing</u> <u>Countries</u>. Johns Hopkins University Press, Baltimore, 1983.
- 93. Von Pischke, J.D., Peter J. Heffernan, and Dale W Adams, "The Political Economy of Specialized Farm Credit Institutions in Low-Income Countries", World Bank, Staff Working Paper No. 446, 1981.
- 94. Wallace, T. D., and A. Hussain, "The Use of Error Components Models in Combining Cross Section with Time Series Data", Econometrica, 37 (January 1969), 55-72.
- 95. Wette, Hildegard C., "Collateral in Credit Rationing in Markets with Imperfect Information: Note", <u>American</u> Economic Review, 73 (June 1983), 442-445.
- 96. World Bank, Agricultural Credit. Sector Policy Paper, Washington, D.C., (May 1975).
- 97. Yu-Shan, Adriana, "Historia y Desarrollo de la Banca Privada en Honduras", Thesis, Facultad de Ciencias Económicas, Universidad Nacional Autónoma de Honduras. Tegucigalpa, Honduras, 1968.

APPENDIX A

Appendix to Chapter II

Honduras: Country Profile, Selected Indicators, 1982.

Area: 112,100 square kilometers (43,281 square miles)
Population: 3.96 million
Population Growth: 3.5% per year
Percent of Population in Rural Areas: over 60%

Gross Domestic Product (GDP): US \$2.8 billion Gross National Product (GNP): US \$2.6 billion Growth Rate of Real GNP (average 1978-1982): 3.3% per year GNP Per Capita: U.S. \$660.1

Share of Agriculture in Total GDP: 27.5% Share of Agriculture in Total Employment: 60% Share of Agriculture in Total Export Earnings: 67% Main Exports: Bananas, coffee, lumber, meat.

a/ Sources: Banco Central de Honduras [9], International Monetary Fund [42], and Wheeler, Richard O. et al., Report of the U.S. Presidential Agricultural Mission to Honduras, November 1982.

				Financial	Instituti	ons			
	Private		Pu	blic		Other Financia			Ratio of Total
	Commer	cial Banks	Develop	ment Banks	Tota			tutions	Population per Office
		Number of	•	Number of		Number of		Number of	(Thousand Inhabitants
Year	Number	Offices	Number	Offices	Number	Offices	Number	Offices	per Office)
1950	2	7	0	0	2	7	2	2	160.0
1968	8	52	2	23	11	75	4	7	28.2
1976	13	175	3	29	16	204	5	8	15.1
1977	13	183	3	30	16	213	6	9	15.0
1978	13	191	3	31	16	222	8	12	14.7
1979	14	200	3	31	17	231	8	21	14.1
1980	15	218	3	31	18	249	8	29	13.3
1981	15	212	3	31	18	243	8	31	13.9
1982	15	212	3	31	18	243	8	31	14.5

Table 28. Honduras: Size and Composition of the Financial System. Number of Institutions and Number of Offices of Main Financial Institutions.<sup>a</sup>/ Selected Years, 1950-82

Sources: Financial institutions in 1950 and 1968, Yu-Shan [97]. Financial institutions in 1976-1982, Banco Atlantida, Economic Studies Division. Population, International Monetary Fund [42].

a/ Not including Central Bank offices

ŧ

	Proporti	on of New Loa	ans Granted	Shares in Overall Deposit Mobilization <sup>a</sup> /				
	Private Commercial Banks	Public Development Banks	Other Financial Institutions	Private Commercial Banks	Public Development Banks	Other Financial Institutions		
Year	8	8	8	ક	8	\$		
1960	77.6	16.9	5.5	n.a.	n.a.	n.a.		
1965	70.5	21.2	8.3	n.a.	n.a.	n.a.		
1970	82.3	10.4	7.3	91.2	6.3	2.5		
1975	86.8	9.2	4.0	91.4	6.6	2.0		
1976	86.7	8.0	5.3	91.0	7.2	1.8		
1977	83.9	14.0	2.1	92.1	5.0	2.9		
1978	86.9	11.0	2.1	91.1	4.8	4.1		
1979	82.3	14.7	3.0	88.7	5.8	5.5		
1980	78.8	16.6	4.6	86.4	6.0	7.6		
1981	80.4	11.4	8.2	86.2	5.2	8.6		
1982	82.5	11.7	5.8	86.5	5.4	8.1		

Table 29.	Relative Importance of Commercial Banks, Development Banks,
	and Other Financial Institutions in the Banking System.
	Proportion of Total New Loans Granted, and Shares in
	Overall Deposit Mobilization. Selected Years, 1960-1982

Sources: Proportion of New Loans, 1960-80 from Gonzalez-Vega [33], 1981-82 computed from Central Bank Statistical Bulletin [9]. Shares in Overall Deposit Mobilization, computed from Central Bank Statistical Bulletin [9].

a/ Based on end-of-year balances of private-sector and public-sector deposits.

n.a.: not available

	Tatoma	+		Interest	-rate C	eiling		
	Inceres	c-race ce	erring	on rime Deposits				
	on Sav	ings Depo	sits,	Average	2/			
	Nominal Real rate"			Nominal	Real rate <u>a</u> /			
	rate	(CPI)	(ID)	rate <u>b</u> /	(CPI)	(ID)		
Year	8	ે જે	<u> </u>	8	ે સ્ટં	<u></u> 8		
						_		
1970	4.0	0.1	1.1	7.0	3.0	4.0		
1971	4.0	1.9	3.7	7.0	4.8	6.7		
1972	4.0	0.8	-0.4	7.0	3.7	2.4		
1973	4.0	-0.6	-1.2	7.0	2.3	1.6		
1974	5.0	-6.9	-4.0	7.8	-4.4	-1.4		
1975	6.0	-1.9	0.7	8.5	0.4	3.0		
1976	6.0	0.9	-8.7	8.5	3.2	-6.6		
1977	6.0	-2.2	-2.5	8.5	0.1	-0.2		
1978	6.0	0.3	-3.5	8.5	2.6	-1.3		
1979	7.0	-4.5	-4.1	<u>c</u> /	-	-		
1980	7 .8	-8.7	-6.2	-	-	-		
1981	<u>a</u> /	-	-	-	-	-		
1982	-	-		-	-			

Table 30. Interest-rate Ceilings: Nominal and Real Rates on Savings and Time Deposits, 1970-1982. Real Rates Computed for Two Deflators: the Consumer-Price Index (CPI) and the Implicit GDP Deflator (ID)

Sources: Central Bank publications, Statistical Bulletin [9], National Accounts [10], Consumer Price Index [11].

<u>a</u>/ Computed as r = [(i - p) / (l + p)], where: r is the real rate of interest, i is the nominal interest-rate, p is the rate of inflation, according to the CPI or to the Implicit GDP Deflator (ID).

b/ Annual average of rates for time-deposits of less than L.100,000, different terms. Rates for time deposits over L.100,000 were free most of the period (excepting those of less than 6 months-term between 1974 and 1978, fixed at 8%).

d/ Free since May 1981.

c/ Free after 1978.

		Maximum Lend.	ing-rate	<u>Ceiling</u>	Average Len	ding-rat	e Ceiling
	Different	Nominal	Real	rate <u>c</u> /	Nominal	Real	rate
	Lending-rațe	rate	(CPI)	(ID)	rate	(CPI)	(ID)
Year	Ceilings <sup>a</sup> /	8	<u> </u>	8	&	<u> </u>	<u>ę</u>
1970	n.a.	18	13.6	14.7	-	-	
1971	n.a.	18	15.6	17.7	-		
1972	n.a.	18	14.3	13.0	-	-	
1973	7	18	12.8	12.1	12.0	7.1	6.4
1974	8	18	4.6	7.9	12.8	0.0	3.1
1975	9	16	7.3	10.2	12.1	3.7	6.5
1976	9	16	10.4	-0.1	12.1	6.7	-3.4
1977	9	16	7.0	6.6	12.1	3.4	3.1
1978	9	16	9.7	5.5	12.1	6.1	2.0
1979	8	16	3.5	3.9	12.0	-0.1	0.3
1980	9	19	0.8	3.6	13.3	-4.1	-1.4
1981	7	19	8.8	17.3	14.3	4.5	12.6
1982	7	19	8.8	11.5	15.0	5.1	7.8

Table 31. Interest-rate Ceilings: Nominal and Real Lending-rate Ceilings, 1970-1982. Real Rates Computed for Two Deflators: the Consumer Price Index (CPI) and the Implicit GDP Deflator (ID)

Sources: Central Bank publications: Statistical Bulletin [9], National Accounts [10], Consumer Price Index [11]. Also Banco Atlantida, Economic Studies Division (internal records).

<u>a</u>/ Lending-rate ceilings established by the Central Bank for different end-uses and/or loan-sizes, with or without use of rediscount funds.

b/ Usually allowed for operations with resources mobilized from the general public.

c/Real rate computed as r = [(i - p) / (1 + p)],

where r is the real rate of interest,

i is the nominal interest-rate

 $\dot{p}$  is the rate of inflation, according to the CPI, or to the Implicit GDP Deflator (ID)

n.a.: not available

		General	Production <sup>D/</sup>	Basic-Gra	ing Production	Mar	keting <sup>C/</sup>
Year	Number of Rediscount Lines <sup>a</sup>	Discount rate	Lending-rate ceiling	Discount rate	Lending-rate ceiling %	Discount rate %	Lending-rate ceiling %
1970	n.a.	4	n.a.	n/e	-	n/e	-
1971	n.a.	4	n.a.	n/e	-	n/e	-
1972	n.a.	4	n.a.	n/e	-	n/e	-
1973	3	4	9	n/e	-	6	n.a.
1974	3	4	9	n/e	-	6	13
1975	4	5	10	3	7	7	13
1976	4	5	10	3	7	7	13
1977	7	5	10	3	7	7	13
1978	6	6	11	5	11	8	13
1979	8	8	13	6	13	4	13
1980	5	8	13	7	13	9	13
1981	7	11	16	7	13	12	16
1982	9	14	19	10	16	15	19

Table 32. Central Bank Rediscount Lines: Number of Rediscount Lines, DiscountRates and Associated Lending-rates of Main Rediscount Lines, 1970-1982

Sources: Central Bank Statistical Bulletin [9]. Banco Atlantida, Economic Studies Division (internal records).

a/ Counts lines with different discount-rates, i.e., it is a lower bound for the total number of existing rediscount lines. These are defined according to end-use of loans and/or to loan-size.

b/ Agriculture (includes Livestock), and Manufacture.

 $\overline{c}$  / Agriculture and Non-agricultural products. Includes export financing in some years.

n.a.: not available
n/e: nonexistent

APPENDIX B

Appendix to Chapter IV

•

.

B.l. Shares of Each Output in Total Marginal Cost, in Multi-Output Production.

The share of the ith output in total marginal cost, is defined as in Laitinen [61]:

$$g_{i} = \frac{q_{i}}{\rho} \frac{\partial c}{\partial q_{i}}$$
(1)

where,  $g_i$  is the share of output i in total marginal cost,

 $q_i$  is the quantity of output i,

 $\rho$  is Laitinen's definition of total or overall marginal cost,

$$\rho = \Sigma_{i} q_{i} \frac{\partial C}{\partial q_{i}} , \qquad (2)$$

 $\frac{\partial C}{\partial q_i}$  is the individual marginal cost of output i.

Recalling that:  $\frac{\partial \ln C}{\partial \ln q_i} = \frac{\partial C}{\partial q_i} \frac{q_i}{C}$ ,

and dividing (2) by C,

.•

$$\frac{\rho}{C} = \Sigma_{i} \frac{q_{i}}{C} \frac{\partial C}{\partial q_{i}} = \Sigma_{i} \frac{\partial \ln C}{\partial \ln q_{i}}$$
(3)

or  $\frac{\rho}{C}$  = ES (see equation 4.4 in the text)

Substituting  $\rho$  in (1):

$$g_{i} = \frac{q_{i}}{C\Sigma_{i}(\partial \ln C/\partial \ln q_{i})} \frac{\partial C}{\partial q_{i}} = \frac{1}{\Sigma_{i}(\partial \ln C/\partial \ln q_{i})} \frac{\partial \ln C}{\partial \ln q_{1}} ,$$
  
$$g_{i} = ES_{i}/ES , i.e., expression (4.8) in the text.$$

### B.2. Allen Partial Elasticities of Substitution in the Translog Cost Function

From Uzawa [89]:

$$\sigma_{ij} = \frac{c \frac{\partial^2 c}{\partial p_i \partial p_j}}{\frac{\partial c}{\partial p_i} \frac{\partial c}{\partial p_j}}$$
(1)

and

:

$$\sigma_{ij} - 1 = \frac{\frac{\partial^2 \ln C}{\partial p_i \partial p_j}}{\frac{\partial \ln C}{\partial p_i} \frac{\partial \ln C}{\partial p_j}}, \quad i = j \quad (2)$$

thus

$$\sigma_{ij} = \frac{\frac{\partial^2 \ln C}{\partial p_i \partial p_j}}{\frac{\partial \ln C}{\partial p_i} \frac{\partial \ln C}{\partial p_j}} + 1$$
(3)

$$\frac{\partial^2 \ln c}{\partial p_i \partial p_j} = \frac{\partial \left(\frac{1}{p_i} \frac{\partial \ln c}{\partial \ln p_i}\right)}{\partial p_j}$$
(4)

$$= \frac{1}{p_i} \frac{\partial^2 \ln c}{\partial \ln p_i \partial \ln p_j} \frac{1}{p_j} = \frac{1}{p_i p_j} \frac{\partial^2 \ln c}{\partial \ln p_i \partial \ln p_j}$$

$$\sigma_{ij} = \frac{\frac{1}{p_i p_j} \frac{\partial^2 \ln C}{\partial \ln p_i \partial \ln p_j}}{\frac{1}{p_i p_j} \frac{\partial \ln C}{\partial \ln p_i} \frac{\partial \ln C}{\partial \ln p_j}} + 1$$
(5)

$$\sigma_{ij} = \frac{\frac{\partial^2 \ln C}{\partial \ln p_i \partial \ln p_j} + \frac{\partial \ln C}{\partial \ln p_i} \frac{\partial \ln C}{\partial \ln p_j}}{\frac{\partial \ln C}{\partial \ln p_i} \frac{\partial \ln C}{\partial \ln p_j}}$$

recall: 
$$\frac{\partial \ln C}{\partial \ln p_i} = \frac{\partial C}{\partial p_i} \frac{p_i}{C}$$
  
and,  $\frac{\partial C}{\partial p_i} = x_i$ , (Shephard's lemma, [91])

therefore  $\frac{\partial \ln C}{\partial \ln p_i} = \frac{p_i x_i}{C} = S_i$  the cost share of factor i

then 
$$\sigma_{ij} = \frac{\frac{\partial^2 \ln C}{\partial \ln p_i \partial \ln p_j} + S_1 S_2}{S_1 S_2}$$
,  $i \neq j$ 

for i = j, (4) becomes:

$$\frac{\partial^{2} \ln c}{\partial p_{i}^{2}} = \frac{\partial \left(\frac{\partial \ln c}{\partial p_{i}}\right)}{\partial p_{i}} = \frac{\partial \left(\frac{1}{p_{i}}, \frac{\partial \ln c}{\partial \ln p_{i}}\right)}{\partial p_{i}}$$

$$= \frac{1}{p_{i}}, \frac{\partial^{2} \ln c}{\partial (\ln p_{i})^{2}}, \frac{1}{p_{i}} + \frac{\partial \ln c}{\partial \ln p_{i}}, \left(-\frac{1}{p_{i}^{2}}\right)$$

$$\frac{\partial^{2} \ln c}{\partial p_{i}^{2}} = \frac{1}{p_{i}^{2}}, \left(\frac{\partial^{2} \ln c}{\partial (\ln p_{i})^{2}}, -\frac{\partial \ln c}{\partial \ln p_{i}}\right)$$

$$\sigma_{ii} = \frac{\frac{1}{p_{i}^{2}}, \left(\frac{\partial^{2} \ln c}{\partial (\ln p_{i})^{2}}, -\frac{\partial \ln c}{\partial \ln p_{i}}\right)}{\frac{1}{p_{i}^{2}}, \left(\frac{\partial \ln c}{\partial \ln p_{i}}\right)^{2}} + 1$$

$$\sigma_{ii} = \frac{\frac{\partial^{2} \ln c}{\partial (\ln p_{i})^{2}} - \frac{\partial \ln c}{\partial \ln p_{i}} + \left(\frac{\partial \ln c}{\partial \ln p_{i}}\right)^{2}}{\left(\frac{\partial \ln c}{\partial \ln p_{i}}\right)^{2}}$$

$$\sigma_{ii} = \frac{\frac{\partial^{2} \ln c}{\partial (\ln p_{i})^{2}} + s_{i}(s_{i} - 1)}{s_{i}^{2}}$$

In terms of the parameters of the translog function:

$$\sigma_{ij} = \frac{(\delta_{ij} + s_i s_j) / s_i s_j}{(\delta_{ii} + s_i (s_i - 1)) / s_i^2}, \quad i \neq j$$

## B.3. Simultaneity and Estimation Bias in the Transaction-Costs Function

The issue of simultaneity in the transaction-costs function was discussed in section 4.2. It was pointed out in that section that, if the "true" model involved a loandemand function that depends on the magnitude of transaction costs, then the trade-off equation (4.20) includes a variable (loan amount) that is correlated with the error term. Ιn this case, the single-equation estimation of (4.20) would yield biased and inconsistent estimates of the parameters in the function. This appendix shows that the expected signs of the biases in the estimated coefficients of the variables loan-amount and interest-rate are negative and positive, respectively. For simplicity, other variables are omitted and the equations are written in linear form without the logarithm notation. The exposition draws upon the treatment given to this simultaneity problem in Johnston [51].

Let the simultaneous-equation system be:

 $T_{j} = \alpha + \beta L + \delta i_{j} + u_{j}, \text{ transaction-costs function, (1)}$   $L_{j} = a + bT_{j} + v_{j}, \text{ loan-demand function, (2)}$   $j = 1, \dots, n,$ 

where, T is transaction costs per loan,

L is loan amount;

i is the interest rate,

u, v are stochastic error terms.

The following assumptions apply:

E(u) = 0, E(v) = 0,

E 
$$(u^2) = \sigma_u^2$$
, E  $(v^2) = \sigma_v^2$ ,  
Cov  $(u,v) = 0$ ,  
Cov  $(i,u) = 0$ , Cov  $(i,v) = 0$   
 $\beta > 0$ ,  $\delta < 0$ ,  $b < 0$ .

Using Johnston's notation for second-order moments, for any two variables X and Y:

$$m_{YX} = \frac{1}{n} \Sigma_{j} (Y_{j} - \bar{Y})(X_{j} - \bar{X})$$
(3)

$$m_{XX} = \frac{1}{n} \Sigma_{j} (X_{j} - \overline{X})^{2}$$
(4)

The OLS estimate for the coefficients in equation (4.21) can be written as:

$$\hat{\beta} = \frac{m_{\rm TL}}{m_{\rm LL}}$$
(5)

$$\hat{\delta} = \frac{m_{Ti}}{m_{ii}}$$
(6)

The reduced-form system of equations that corresponds to the structural form (1), (2) is the following (j-subscripts omitted):

$$T = \frac{\delta}{\lambda} i + \frac{\alpha + a\beta}{\lambda} + \frac{1}{\lambda} u + \frac{\beta}{\lambda} v$$
 (7)

$$L = \frac{b\delta}{\lambda}i + \frac{a + b\alpha}{\lambda} + \frac{b}{\lambda}u + \frac{1}{\lambda}v$$
(8)

where,  $\lambda = (1 - b\beta)$ 

In deviations from the sample means:

$$\mathbf{T} - \overline{\mathbf{T}} = \frac{\delta}{\lambda} (\mathbf{i} - \overline{\mathbf{i}}) + \frac{1}{\lambda} (\mathbf{u} - \overline{\mathbf{u}}) + \frac{\beta}{\lambda} (\mathbf{v} - \overline{\mathbf{v}})$$
(9)

$$L - \overline{L} = \frac{b\delta}{\lambda} (i-\overline{i}) + \frac{b}{\lambda} (u-\overline{u}) + \frac{1}{\lambda} (v-\overline{v})$$
(10)

therefore,

$$m_{TL} = \frac{1}{\lambda^2} \left[ b\delta^2 m_{ii} + 2b\delta m_{iu} + \delta(1 + b\beta)m_{iv} + bm_{uu} + \beta m_{vv} + (1 + b\beta)m_{uv} \right]$$
(11)

$$m_{LL} = \frac{1}{\lambda^2} \left[ b^2 \delta^2 m_{ii} + 2b^2 \delta m_{iu} + 2b \delta m_{iv} + b^2 m_{uu} + ... + m_{vv} + 2b m_{uv} \right]$$
(12)

$$m_{Ti} = \frac{1}{\lambda} \left( \delta m_{ii} + m_{iu} + \beta m_{iv} \right)$$
(13)

Now, the OLS estimate for the coefficient of the loan amount variable in (A.1) is:

$$\hat{\beta} = \frac{m_{TL}}{m_{LL}}$$
,

$$\hat{\beta} = \frac{b\delta^{2}m_{ii} + 2b\delta m_{iu} + \delta(1+b\beta)m_{iv} + bm_{uu} + \beta m_{vv} + (1+b\beta)m_{uv}}{b^{2}\delta^{2}m_{ii} + 2b^{2}\delta m_{iu} + 2b\delta m_{iv} + b^{2}m_{uu} + m_{vv} + 2bm_{uv}}$$

Under the assumptions laid out above, as  $n \rightarrow \infty$ :

$$m_{uu} \rightarrow \sigma_n^2$$
,  $m_{vv} \rightarrow \sigma_v^2$ ,  
 $m_{iu} \rightarrow 0$ ,  $m_{iv} \rightarrow 0$ ,  $m_{uv} \rightarrow 0$ ,  
and  $m_{ii}$  tends to some value  $\overline{m}_{ii} > 0$ .

Therefore,

$$plim\hat{\beta} = \frac{b\delta^2 \,\overline{m}_{ii} + b\sigma_n^2 + \beta\sigma_v^2}{b^2\delta^2 \,\overline{m}_{ii} + b^2\sigma_n^2 + \sigma_v^2},$$

that can be transformed into:

$$plim\hat{\beta} = \beta + \frac{(1-b\beta)(b\delta^2 \ \overline{m}_{ii} + b\sigma_n^2)}{b^2 \delta^2 \ \overline{m}_{ii} + b^2 \sigma_u^2 + \sigma_v^2}$$
(14)  
$$n \neq \infty$$

Given the signs assumed for  $\beta$  and b in the structural form, the second term on the right-hand side of (14) is negative, therefore :

plim
$$\hat{\beta} < \beta$$
 ,

n → ∞

i.e., the estimate  $\hat{\beta}$  is biased downward, and this bias does not vanish as  $n \rightarrow \infty$ . Thus, the estimate is also inconsistent. In turn, the estimated coefficient of the interest-rate variable is:

$$\delta = \frac{m_{Ti}}{m_{ii}} ,$$

$$\hat{\delta} = \frac{1}{\lambda} \left( \delta + \frac{m_{iu}}{m_{ii}} + \beta \frac{m_{iv}}{m_{ii}} \right) \quad .$$

. Under the assumptions used here:  $plim\hat{\delta} = \frac{\delta}{\lambda} ,$   $n \neq \infty$ 

that can be written as:

$$plim\delta = \delta + \frac{\delta b\beta}{\lambda}$$
(15)

where the second term on the right-hand side is positive, therefore:

plim $\hat{\delta} > \delta$ ,

i.e.,  $\hat{\delta}$  is biased upward. Since the magnitude of the bias  $(\delta b \beta / \lambda)$  is given by parameters with real values different from zero, the estimate of  $\delta$  is also inconsistent.

APPENDIX C

•

-

Appendix to Chapter V

			Model (out	put detini	TION, TUNCT	Ional form	<b>,</b>		
		17		2)	C.	57	(	<u>17</u>	
	Number	Number of	Loans (q1),	( - )			Loans (q,),		
		or bepos	in Accounts	·427	•	Depesti Derences 142			
	Cobb-D	ouglas	Tran	slog	Cobb-De	ouglas	Tran	slog	
Parameter (Varlable)	Estimate	t-ratio	Estimate	T-ratio	Estimate	t-ratio	Estimate	t-ratio	
a <sub>0</sub> (intercept)	0.9294	1.321	24.7935	2.665*	3.7372	8.055*	-1.2233	-0.244	
$\alpha_1$ (inq <sub>1</sub> , loans)	0.1519	4+085*	-1.5198	-1.722*	0.4786	15.018*	-0.4352	-0.889	
α <sub>2</sub> (inq <sub>2</sub> , deposits)	0.4574	5.238°	-2.9849	-1.747*	0.1753	5.502*	1.3378	2.3161	
$\beta_1$ (inp <sub>1</sub> , price of Labor)	1.3163	8-129*	-0.9386	3.596*	0.7355	6-175*	2.4714	0.986	
$\beta_2$ (inp <sub>2</sub> , price of capital)	-0.1473	-2.992*	-2-4711	-2-2861	0-1150	3.068*	-0.2199	-0.337	
γ11 (Ing) <sup>2</sup>			-0.1104	-1-978†			0.0633	1.9741	
Y <sub>22</sub> (Inq <sub>2</sub> ) <sup>2</sup>			-0.1632	-0.568			-0.0438	-1.057	
γ <sub>12</sub> (Inq <sub>1</sub> Inq <sub>2</sub> )			0.1796	1-427			0.0568	1.702*	
$\delta_{11} (\ln p_1)^2$			-0.7590	-0.765			-0-1449	-0.216	
$\delta_{22} (Inp_2)^2$			0.1487	1.611*			0.0363	0.786	
δ <sub>12</sub> (Inp <sub>1</sub> Inp <sub>2</sub> )			0.5351	2•689*			0.1671	0.977	
η <sub>ll</sub> (inq <sub>l</sub> inp <sub>l</sub> )			0.2275	1.672†			0.0419	0.382	
$\eta_{12}$ (inq <sub>1</sub> inp <sub>2</sub> )			-0.0068	-0-141			0.0158	0-407	
η <sub>21</sub> (inq <sub>2</sub> inp <sub>1</sub> )			0.6444	1.621*			-0.2400	-1.493	
$\eta_{22} (\ln q_2 \ln p_2)$			-0.0685	-0.494			-0.0975	-2+313t	
R <sup>2</sup>	0.7567		0•7922		0.8208		0.8598		
F-val ue	120.54*		39.49*		324.03*		119,59*		
F-test of functional form <sup>a/</sup>			2.47*				7 <b>.</b> 60*		

Table 33.	Development Definitions	Sank: Estimat	red Parameters Forms. Depen	of the Cost Fu ident Variable:	inction, for Different Administrative Co:	nt Output sts (InC)
-----------	----------------------------	---------------	--------------------------------	-----------------------------------	--	------------------------

.

OLS estimation, N=160 for models (1) and (2), N=288 for models (3) and (4).

.

<u>a</u>/ F=((SSE<sub>C</sub> - SSE<sub>T</sub>)/10) / (SSE<sub>T</sub>/N-14), where SSE = error sum of squares Significance levels: °, .01 C denotes Cobb-Douglas form t, .05 T denotes Translog form °, .10

	Model (output definition, functional form)									
	(	D	C	23	(.	ົ້	(	<del>ان</del>		
	Number	of Loans +	Deposit Accounts		Value of Loans +		Deposit Balances			
	CODD-DO	ouglas	Iranslog		CODD-Do	ougras	iranslog			
Parameter (Varlable)	ESTIMATO	T-ratio	ESTIMATO	T-FATIO	ESTIMATO	T-Fatio	ESTIMATO	T-FATIO		
α <sub>0</sub> (Intercept)	0.6644	0.932	15.6583	1•786°	1.9225	4-114*	0.9483	0.198		
$\alpha_1$ (InQ, loans + deposits)	0•4752	6.116*	-0.8433	-0.481	0.6431	13.661*	-1.9831	-2.067†		
$\beta_1$ (Inp <sub>1</sub> , price of labor)	1.4704	8.919*	-2.4458	<del>-</del> 0.735	1.0043	8•553*	5.2828	2.217†		
β <sub>2</sub> (Inp <sub>2</sub> , price of capital)	-0.1797	-3.529*	-0•9955	-0.915	0.1306	2•984*	<del>-</del> 2•2977	<del>-</del> 2•554*		
$\gamma_{11}$ (InQ) <sup>2</sup>			-0.5788	-3.042*			-0.0037	-0.049		
$\delta_{11} (\ln p_1)^2$			-0.9617	-0.936			-1.6008	-2.252†		
$\delta_{22} (Inp_2)^2$			0.0807	0•902			0.0369	0.419		
$\delta_{12} (Inp_1Inp_2)$			0.4596	2.296†			0.6276	3.067*		
η <sub>11</sub> (inQinp <sub>1</sub> )			1.1400	2.831*			0.5154	2•255†		
$\eta_{12}$ (InQInp <sub>2</sub> )			-0.2070	-1•765°			-0.1199	-1.488		
R <sup>2</sup>	0.7325		0•7650		0•7946		0.8167			
F-value	142•43*		54.25*		364•97		137.17			
F-test of functional form <mark>a</mark>			3.45*							

Table 34. Development Bank: Estimated Parameters of the Cost Function, for the Single-Output Definition (Loans + Deposits), and Different Functional Forms. Dependent Variable: Administrative Costs (InC)

OLS estimation, N=160 in models (1) and (2), N=288 in models (3) and (4).

a/ F=[(SSE SSE_)/6]	/ (SSE <sub>T</sub> /N-10), where S	SE = error	sum of squares	Significance levels:	*,	•01
_ 0 1	·	C denotes	Cobb-Douglas form		t,	•05
		T denotes	Translog form		۰,	•10

	Model (output definition)						
	Number o Number of Depo	(1) f Loans (q <sub>1</sub> ), psit Accounts (q <sub>2</sub> )	(2) Value of Loans (q <sub>1</sub> ), Deposit Balances (q <sub>2</sub> )				
Parameter (Variable)	Estimate	t-ratio	Estimate	t-ratio			
α <sub>0</sub> (intercept)	-0-2973	-0.101	5.1313	5.652*			
$\alpha_1$ (Inq <sub>1</sub> , ioans)	-0.2284	-0.527	0•0574	0•297			
α <sub>2</sub> (Inq <sub>2</sub> , deposits)	1.3004	1.376	0.0768	0.212			
$\beta_1$ (inp <sub>1</sub> , price of labor)	0•4525	0.913	0.7055	3.233*			
$\beta_2$ (lnp <sub>2</sub> , price of capital)	0.5475	1.105	0.2945	1.350			
$\gamma_{11} (lnq_1)^2$	0.0838	1•790°	0•0887	2•362*			
$\gamma_{22} (lnq_2)^2$	-0.1563	-0.809	0.0368	0.375			
γ <sub>12</sub> (Inq <sub>1</sub> Inq <sub>2</sub> )	-0.0179	-0.184	0.0256	0.601			
$\delta_{11} (\ln p_1)^2$	0.1222	2-281†	0•1191	1.257			
$\delta_{22} (lnp_2)^2$	0.1222	2.281†	0.1191	1.257			
$\delta_{12}$ (Inp <sub>1</sub> Inp <sub>2</sub> )	-0.1222	-2.281†	-0.1191	<del>-</del> 1 •257			
η <sub>11</sub> (lnq <sub>1</sub> lnp <sub>1</sub> )	<del>-</del> 0•0536	-1•493	-0.0452	-1.007			
$\eta_{12}$ (lnq <sub>1</sub> lnp <sub>2</sub> )	0.0536	1.493	0.0452	1.007			
$\eta_{21}$ (lnq <sub>2</sub> lnp <sub>1</sub> )	0.0080	0.076	-0.0284	-0.394			
η <sub>22</sub> (Inq <sub>2</sub> Inp <sub>2</sub> )	-0.0080	-0.076	0.0284	0.394			
θ <sub>1</sub> (Inq <sub>1</sub> InLS, ioan-size Interaction)	0.0855	11.500*	-0.0068	-1.059			
$\theta_2$ (inq <sub>2</sub> inDS, deposit-size	0.0205	2•304†	-0.0143	-0.974			
R <sup>2</sup> Interaction)	0.8905		0.8786				
F-value	109.42		97•37				

Table 35. Development Bank: Estimated Parameters of the Cost Function, including Loan-Size and Deposit-Size Effects. Single-Equation Estimation with Two Output Definitions

a/ OLS estimation. Factor-price homogeneity restrictions imposed on estimation. However, none of these restrictions was statistically significant. Levels of significance: \*, •01; t, •05. °, •10.

				Model (funct	Ional for	m, deposit-s	Ize Interac	fion)	
			1)	(2			· · · · · · · · · · · · · · · · · · ·	(	4)
		Cobb-F	ouglas	Trans	100	Cobb-Dougla	s, Deposit-	Translog,	Deposit-
Par	ameter (Variable)	Estimate	t-rafio	Estimate	T-ratio	Estimate	T-ratio	Estimate	T-ratio
α	(Intercept)	-2.9043	-3.591*	76•9435	2.693*	-3.0105	-3.705*	70.7300	2.518*
α	(Ing, Ioans)	0.0369	1.594	1.2882	1.010	0.0500	1.957*	1.1854	0•948
α <sub>2</sub>	(Inq <sub>2</sub> , deposits)	0.9122	29•049 <del>*</del>	0.3836	0.199	0.9697	16.977*	0.0959	0.051
β <sub>1</sub>	(Inp <sub>1</sub> , price of labor)	0.8976	8.109*	-18.5176	-2.334†	0.9033	8.164*	-16.4716	-2.109†
β <sub>2</sub>	(inp <sub>2</sub> , price of capital)	0.2347	7•193*	-2.2939	-1 •274	0.2360	7•237*	-2.2602	-1.280
Y11	(Ing <sub>1</sub> ) <sup>2</sup>			0.0221	0.547			0•0144	0•364
γ <sub>22</sub>	(Inq <sub>2</sub> ) <sup>2</sup>			-0.0814	-1.058			-0.0709	-0.939
γ <sub>12</sub>	(Inq <sub>1</sub> Inq <sub>2</sub> )			0.1099	1.198			0.1256	1.395
δ <sub>11</sub>	(inp <sub>1</sub> ) <sup>2</sup>			2.4241	2•196†			2.1233	1.953†
δ <sub>22</sub>	(Inp <sub>2</sub> ) <sup>2</sup>			-0.0337	-0.435			-0.0269	-0.354
δ <sub>12</sub>	(Inp <sub>1</sub> Inp <sub>2</sub> )			0.2425	1.005			0.2414	1.021
η11	(Inq <sub>1</sub> Inp <sub>1</sub> )			-0.2197	-1.330			-0.2088	-1.290
η <sub>12</sub>	(Inq <sub>1</sub> Inp <sub>2</sub> )			<del>-</del> 0.0275	-0.556			0.0033	0.067
η <sub>21</sub>	(Inq <sub>2</sub> Inp <sub>1</sub> )			0.0734	0.301			0-1124	0•470
η <sub>22</sub>	(Inq <sub>2</sub> Inp <sub>2</sub> )			0•0960	1.350			0.0615	0.870
θ2	(Inq <sub>2</sub> InDS, deposit-size					-0.0097	-1.204	-0.0232	-2.856*
$R^2$	Interaction)	0.9699		0.9758		0.9701		0•9764	
F-v	alue	1487.77*		503.31*		1143•41*		489.52	
F-+	est of functional form_/			4-11*				5.09*	

Table 36.	Private Bank:	Estimated Parameters of	the Cost Function, for Different
	Functional Form	s. Dependent Variable:	Administrative Costs (InC)

.

OLS estimation, N=190. Significance levels: \*, •01; t, •05; °, •10.  $\underline{a}$ / F=[(SSE<sub>C</sub> - SSE<sub>T</sub>)/101 / (SSE<sub>T</sub>/N-14), where SSE = error sum of squares, C denotes Cobb-Douglas form, T denotes Translog form

	Variables								
Variables	Loans (lnq <sub>l</sub> )	Deposits (lnq <sub>2</sub> )	Loans x Deposits (lnq <sub>l</sub> lnq <sub>2</sub> )	(Loans) <sup>2</sup> (1nq <sub>1</sub> ) <sup>2</sup>	(Deposits) <sup>2</sup> (lnq <sub>2</sub> ) <sup>2</sup>	Loans x Price of Labor (lnq <sub>l</sub> lnp <sub>l</sub> )	Deposits x Price of Labor (lnq <sub>2</sub> lnp <sub>2</sub> )		
Loans (lnq <sub>1</sub> )	1	0.88	0.97	0.99	0.89	0.99	0.89		
Deposits (lnq <sub>2</sub> )		1	0.95	0.90	0.99	0.89	0.99		
Loans x Deposits (lnq <sub>1</sub> lnq <sub>2</sub> )			1	0.99	0.97	0.98	0.96		
(Loans) <sup>2</sup> , (lnq <sub>1</sub> ) <sup>2</sup>				1	0.91	0.99	0.91		
(Deposits) <sup>2</sup> , (lnq <sub>2</sub> ) <sup>2</sup>					1	0.91	0.99		
Loans x Price of Labor, (lnq <sub>l</sub> lnp <sub>l</sub> )						1	0.91		
Deposits x Price of Labor (lnq <sub>2</sub> lnp <sub>1</sub> )							1		

• •

# Table 37. Private Bank: Correlation Coefficients Between Selected Variables of the Cost Function. All Variables in Natural Logarithms.<sup>a</sup>/

a/ All coefficients statistically significant at 0.01 level.

	··			Correlation Coefficients with Selected Variables					
Bank,	, Variable	Mean Z	Standard Deviation X	Nominal Average Deposit-Rate Ceiling (d <sub>c</sub> )	Real Average Deposit-Rate Ceiling (d <sub>c</sub> -p)	Nominal Lending-Rate Ceiling (1)	Real Lending-Rate Ceiling (1-p)		
Devel	lopment bank								
No Pa	ominal Implicit Rate aid on Deposits (d <sub>p</sub> ) <u>a</u> /	3.62	2.58	-0.0059		0.0186			
Re Pa	eal Implicit Rate aid on Deposits (d <sub>p</sub> - p)	-4.51	5.43		0.8055*		0.8428*		
Priva	ite bank								
No Pa	ominal Implicit Rate aid on Deposits (d <sub>p</sub> )	3.73	1.20	0.2954*		0.2346*			
Re Pa	eal Implicit Rate aid on Deposits (d <sub>p</sub> - p)	-4.13	4.84		0.9315*		0.9520*		
A11									
No De	painal Average eposit-Rate Ceiling (d <sub>c</sub> )	6.56	1.65	1	-0.4306*	-0.3455*			
Re Ce	eal Average Deposit-Rate eiling (d <sub>c</sub> - p̀)	-2.51	3. ú2		1		0.8799*		
No Ce	ominal Lending-Rate eiling (1)	17.22	1.60			1	0.6132*		
Re Ce	eal Lending-Rate eiling (1 - p)	9.08	5.51				. 1		

#### Table 38. Mesns, Standard Deviations, and Correlation Coefficients for Interest-Rate Ceilings and Implicit Rates Paid on Deposits. Nominal and Real Terms. Results for the Development Bank and the Private Bank

.

\* : significant at 0.01 level a/ Total Interest Payments/Total Deposit Balances (includes demand deposits)

Source of Funds												
	Deposits			с	Central Bank			Foreign Funds			Central Bank and Foreign Funds Combined	
Year	Share in No. of Loans 2	Share in Loan Amount X	Average Loan Size Lps.1/	Share in No. of Loans X	Share in Loan Amount Z	Average Loan Size Lps.	Share in No. of Loans 2	Share in Loan Amount Z	Average Loan Size Lps.	Share in No. of Loans Z	Share in Loan Amount Z	
1971	64.5	64.7	1223	0	0		35.5	35.3	1217	35.5	35.3	
1972	32.0	53.4	2215	0	0		68.0	46.6	910	68.0	46.6	
1973	24.9	54.6	3318	3.7	0.5	208	71.4	44.9	953	75.1	45.4	
1974	19.6	51.5	3348	41.0	8.9	276	39.4	39.6	1282	80.4	48.5	
1975	19.6	39.8	2236	43.2	20.0	507	37.2	40.2	1188	80.4	60.2	
1976	30.4	56.1	2679	9.0	12.1	1942	60.6	31.8	760	69.6	43.9	
1977	31.9	64.5	4528	22.5	14.7	1464	45.7	20.8	1021	68.2	35.5	
1978	10.4	33.0	8976	76.1	35.2	1306	13.5	31.8	6608	89.6	67.0	
1979	7.7	46.2	17953	80.4	33.8	1256	11.9	20.0	5027	92.3	53.8	
1980	5.9	34.7	15107	79.2	40.5	1312	14.9	2+-8	4293	94.1	65.3	
1981	4.3	42.1	22496	77.2	36.5	1036	18.5	21.4	2641	95.7	57.9	
1982	5.0	49.8	27238	63.5	30.2	<b>99</b> 5	31.5	20.0	1729	95.0	50.2	

Table 39. Shares of Different Sources of Funds in the Portfolio of New Loans of the Development Bank, and Average Loan Size by Source, 1971-1982.

Source: BANADESA, Economic Studies Department.  $\frac{1}{1}$  1 US\$ = 2 lempiras

.

.

					Total Agriculture			
	Cro	ops	Lives	stock	(Crops + Livestock			
	Share in	Share	Share in	Share	Share in	Share		
	No. of	in Loan	No. of	in Loan	No. of	in Loan		
	Loans	Amount	Loans	Amount	Loans	Amount		
<u>Year</u>	<u> </u>	<u> </u>		&		<u> </u>		
1971	74.3	40.1	23.6	27.5	97.9	67.6		
1972	70.7	43.2	25.7	30.2	96.4	73.4		
19 <b>73</b>	67.7	46.5	27.4	29.6	95.1	76.1		
1974	83.9	62.6	13.9	19.8	97.8	82.4		
1975	88.8	74.2	8.8	13.6	97.6	87.8		
1976	86.1	61.3	12.4	13.8	98.5	75.1		
1977	79.8	79.2	14.4	12.6	94.2	91.8		
1978	89.1	85.2	9.3	7.1	98.4	92.3		
1979	91.4	76.0	7.0	6.7	98.4	82.7		
1980	96.7	72.4	1.5	3.9	98.2	76.3		
1981	94.7	57.3	3.8	8.4	98.5	65.7		
1982	93.1	68.2	5.4	8.5	98.5	76.7		

Table 40. Shares of Agriculture Loans in Portfolio of New Loans of the Development Bank, 1971-1982.

Source: BANADESA, Economic Studies Department.

·	Source of Funds								
Loans to	Depo	sits	Centra	l Bank	Foreign	Funds			
Agriculture	Number	Amount	Number	Amount	Number	Amount			
Crops <sup>b/</sup>									
Number	-0.36		0.38		<b>⊸0.32</b>				
Amount		-0.25		0.33		-0.28			
Livestock <sup>c</sup> /									
Number	0.44		-0.48		0.41				
Amount		0.31		-0.42		0.36			

Table 41.	Correlation Coefficients Between Sources
	of Funds and New Loans to Agriculture
	in the Development Bank <mark>a</mark> /

a/ All coefficients significant at .01 level.  $\frac{b}{N} = 283$   $\frac{c}{N} = 292$ 

,

	Source of Funds									
Source	Depo	sits	Centra	1 Bank	Foreign	Funds				
of Funds	Number	Amount	Number	Amount	Number	Amount				
Deposits										
Number	1.		-0.85		0.56					
Amount		1.		-0.79		0.13 <u>b</u> /				
Central Bank	-		,							
Number			1.		-0.91					
Amount				1.		-0.69				
Foreign Func	ls									
Number					1.					
Amount						1.				

Table 42. Correlation Coefficients Between Shares of Sources of Funds in the Loan Portfolio of the Development Banka

a/ N = 299, all coefficients significant at .01 level, unless specified otherwise.

b/ Significant at .05 level.

APPENDIX D

Appendix to Chapter VI

264

-
						Loan	Source					
		BANADESA	1	Pri	lvate Ban	nks	Cre	dit Unio	ns	A1	1 Source	8
	Pct. of	Mean	Median	Pct. of	Mean	Median	Pct. of	Mean	Median	Pct. of	E	
Loan Size	No. of	Value	Value	No. of	Value	Value	No. of	Value	Value	No of	Mean	Median
Category (Lps.)	Loans	Lps.	Lps.	Loans	Lps.	Lps.	Loans	Lps.	Lps.	Loans	Value	Value
Less than 1,000	9.62	830	900	0.0			30 <b>.9</b> 5	538	480	11.62	665	600
1,001 - 2,000	31.73	1,610	1,700	3.85	1,350	1,350	28.57	1,505	1,450	23.74	1,572	1,500
2,001 - 5,000	16.35	3,556	4,000	1.92	4,000	4,000	23.81	3,385	3,138	14.14	3,510	3,638
5,001 - 10,000	18.27	6,743	6,400	9.62	9,000	10,000	9.52	7,125	7,000	14.14	7,200	6,800
10,001 - 15,000	6.73	11,547	12,000	3.85	15,000	15,000	2.38	12,000	12,000	5.05	12,283	12,000
15,001 - 25,000	4.81	20,800	22,000	11.54	21,762	22,285	2.38	18,000	18,000	6.06	21,048	21,935
25,001 - 50,000	7.69	34,250	35,500	25.00	40,937	41,540	2.38	40,000	40,000	11.11	38,463	<b>39,</b> 500
50,001 - 100,000	3.85	83,925	87,350	30.77	76,759	77,500	0.0	<u></u>		10.10	78,193	80,000
More than 100,000	0.96	110,000	110,000	13.46	179,379	182,400	0.0			4.04	170,707	176,200
All Loans	100.0	11,101	4,000	100.0	62,082	47,622	100.0	3,748	1,650	100.0	22,930	5,290

Table 43. Distribution of Loans by Loan Size, Mean and Median Values by Loan-Size Category, by Source of Loan. Total Sample

-

	Loan Size Category (Lps.)							
	Less th	en 5,000	5,001	- 25,000	More th	an 25,000	Тс	tal
Farm Size	Row	Column	Row	Column	Row	Column	Row	Column
Category (Has.)	Pct.	Pct.	Pct.	Pct.	Pct.	Pct.	Pct.	Pct.
Less than 5	100.0	23.33	0.0	0.0	0.0	0.0	100	13.46
5.1 - 10	89.47	28.33	10.53	6.45	0.0	0.0	100	18.27
10.1 - 20	80.0	20.0	20.0	9.68	0.0	0.0	100	14.42
20.1 - 50	39.13	15.0	56.52	41.94	4.35	7.69	100	22.12
50.1 - 100	50.0	13.33	25.0	12.90	25.0	30.77	100	15.38
100.1 - 200	0.0	0.0	75.0	19.35	25.0	15.38	100	7.69
More than 200	0.0	0.0	33.33	9.68	66.67	46.15	100	8.65
Total	57.69	100	29.81	100	12.50	100	100	100

Table 44.	Farm-Size Distribution of Loans Granted by th	e
	Development Bank, by Loan Size. <u>a</u> /	

 $\underline{a}$  / N = 104

.

.

		Lua	n Size C	ategory (I	.ps.)			
	Less t	hen 5,000	5,001	- 25,000	More th	an 25,000	Тс	tal
Farm Size	Row	Column	Row	Column	Row	Column	Row	Column
Category (Has.)	Pct.	Pct.	Pct.	Pct.	Pct.	Pct.	Pct.	Pct.
Less than 5	50.0	33.33	0.0	0.0	50.0	2.78	100	3.85
5.1 - 10	0.0	0.0	50.0	7.69	50.0	7.69	100	3.85
10.1 - 20	0.0	0.0	33.33	7.69	66.67	5.56	100	5.77
20.1 - 50	0.0	0.0	0.0	0.0	100.0	16.67	100	11.54
50.1 - 100	20.0	66.67	20.0	15.38	60.0	16.67	100	19.23
100.1 - 200	0.0	0.0	50.0	38.46	50.0	13.89	100	19.23
More than 200	0.0	0.0	21.05	30.77	78.95	41.67	100	36.54
Total	5.7	100	25.0	100	69.23	100	100	100

Table 45.	Farm-Size Distribution of Loans Granted by	У
	Private Banks, by Loan Size <sup>a</sup>	-

 $\underline{a}$  / N = 52

267

		ĹOa	n Size C	ategory (I	.ps.)		·	
	Less t	hen 5,000	5,001	- 25,000	More th	nan 25,000	Тс	otal
Farm Size	Row	Column	Row	Column	Row	Column	Row	Column
Category (Has.)	Pct.	Pct.	Pct.	Pct.	Pct.	Pct.	Pct.	Pct.
Less than 5	100.0	37.14	0.0	0.0	0.0	0.0	100	30.95
5.1 - 10	66.67	5.71	33.33	16.67	0.0	0.0	100	7.14
10.1 - 20	100.0	22.86	0.0	0.0	0.0	0.0	100	19.05
20.1 - 50	66.67	22.86	33.33	66.67	0.0	0.0	100	28.57
50.1 - 100	100.0	2.86	0.0	0.0	0.0	0.0	100	2.38
100.1 - 200	100.0	8.57	0.0	0.0	0.0	0.0	100	7.14
More than 200	0.0	0.0	50.0	16.67	50.0	100.0	100	4.76
Total	83.33	100	14.29	100	2.38	100	100	100

Table 46. Farm-Size Distribution of Loans Granted by Credit Unions, by Loan Size<sup>a</sup>

a/N = 42

		•	MOU	έT.	<b>.</b> .		
eter	(1	)	(2	)	(3	)	
pendent Variable)	Estimate	t-ratio	Estimate	t-ratio	Estimate	t-ratio	
intercept) lnG, quarantee/loan amount)	2.6804 0.0855	2.581* 1.426*	3.2154	3.310*	1.3820	1.292	
lnW, labor costs) <u>a</u> / lnL, loan amount)	0.4299	5.408*	0.0402 0.3412	0.472 3.936*	0.0989 0.4830	1.179 4.403*	
lni, interest rate) Slni, small loans x	-0.9384	-3.414*	-0.8715	-3.643	-0.8163	-3.281*	
Interest rate) D <sub>l</sub> lni, private banks x					0.3210	3.190*	
interest rate)					-1.5407	-1.944*	
D <sub>1</sub> , private banks)	0.4762	1.557	0.4797	1.813°	4.3920	2.054*	
$D_2^{-}$ , credit unions)	-0.6346	-2.122†	-0.8283	-3.229*	-0.7405	-3.006*	
U <sub>l</sub> , basic grains loans)	0.3495	1.218	0.3345	1.334	0.3529	1.469	
U <sub>2</sub> , export crops loans)	-0.3341	-0.965	-0.3661	-1.230	-0.2452	-0.856	
$U_3^2$ , livestock loans)	0.3521	1.010	0.4679	1.523	0.5117	1.734	
are	0.4397		0.4654		0.5161		
ue	14.12*		19.37*		18.77*		
	153		187		187		
	<pre>pendent Variable) intercept) InG, guarantee/loan amount) InW, labor costs)<sup>A</sup>/ InL, loan amount) Ini, interest rate) Sini, small loans x</pre>	InterceptInterceptintercept)2.6804InG, guarantee/loan amount)0.0855InW, labor costs)a/InL, loan amount)0.4299Ini, interest rate)-0.9384Slni, small loans x interest rate)-0.9384Slni, small loans x interest rate)0.4762Ol, private banks)0.4762Ol, private banks)0.4762Ol, private banks)0.3495Ol, private banks)0.3521Dasic grains loans)0.3521Ol, private banks)0.4397Ite14.12*153153	Left (1)Estimate t-ratioEstimate t-ratioEstimate t-ratioEstimate t-ratioLand (10, 20, 20, 20, 20, 20, 20, 20, 20, 20, 2	Stell       Image: Constraint of the system       Constrest of the system <td>(17Estimatependent Variable)Estimatet-ratioEstimatet-ratioEstimatet-ratiointercept)<math>2.6804</math><math>2.581*</math><math>3.2154</math><math>3.310*</math>lnG, guarantee/loan amount)<math>0.0855</math><math>1.426*</math><math>0.0402</math><math>0.472</math>lnW, labor costs)<math>0.4299</math><math>5.408*</math><math>0.3412</math><math>3.936*</math>lni, interest rate)<math>-0.9384</math><math>-3.414*</math><math>-0.8715</math><math>-3.643</math>Slni, small loans x interest rate)<math>-0.6346</math><math>-2.122*</math><math>-0.8283</math><math>-3.229*</math><math>0_1</math>, private banks)<math>0.4762</math><math>1.557</math><math>0.4797</math><math>1.813^\circ</math><math>0_2</math>, credit unions)<math>-0.6346</math><math>-2.122*</math><math>-0.8283</math><math>-3.229*</math><math>0_1</math>, basic grains loans)<math>0.3495</math><math>1.218</math><math>0.3345</math><math>1.334</math><math>0_2</math>, export crops loans)<math>-0.3341</math><math>-0.965</math><math>-0.3661</math><math>-1.230</math><math>0_3</math>, livestock loans)<math>0.3521</math><math>1.010</math><math>0.4679</math><math>1.523</math>are<math>0.4397</math><math></math><math>0.4654</math><math></math>16<math>14.12*</math><math>19.37*</math><math>153</math><math>187</math></td> <td>(17)(17)Estimate t-ratioEstimateEstimate t-ratioEstimateintercept)2.68042.581*3.21543.310*1.3820Inde, guarantee/loan amount)0.08551.426*Inde, loan amount)0.08551.426*Inde, loan amount)0.042995.408*0.34123.936*0.4830Interest rate)0.04020.4720.0989Int, loan amount)0.42995.408*0.34123.936*0.4830Interest rate)0.04020.4720.0989Int, interest rate)-0.9384-3.414*-0.8715-3.643-0.81630.1Interest rate)-1.54070.1-1.54070.10.47621.5570.47971.813*4.39200.1-0.6346-2.122t-0.8283-3.229*-0.74050.1-0.3341-0.965<th c<="" td=""></th></td>	(17Estimatependent Variable)Estimatet-ratioEstimatet-ratioEstimatet-ratiointercept) $2.6804$ $2.581*$ $3.2154$ $3.310*$ lnG, guarantee/loan amount) $0.0855$ $1.426*$ $0.0402$ $0.472$ lnW, labor costs) $0.4299$ $5.408*$ $0.3412$ $3.936*$ lni, interest rate) $-0.9384$ $-3.414*$ $-0.8715$ $-3.643$ Slni, small loans x interest rate) $-0.6346$ $-2.122*$ $-0.8283$ $-3.229*$ $0_1$ , private banks) $0.4762$ $1.557$ $0.4797$ $1.813^\circ$ $0_2$ , credit unions) $-0.6346$ $-2.122*$ $-0.8283$ $-3.229*$ $0_1$ , basic grains loans) $0.3495$ $1.218$ $0.3345$ $1.334$ $0_2$ , export crops loans) $-0.3341$ $-0.965$ $-0.3661$ $-1.230$ $0_3$ , livestock loans) $0.3521$ $1.010$ $0.4679$ $1.523$ are $0.4397$ $$ $0.4654$ $$ 16 $14.12*$ $19.37*$ $153$ $187$	(17)(17)Estimate t-ratioEstimateEstimate t-ratioEstimateintercept)2.68042.581*3.21543.310*1.3820Inde, guarantee/loan amount)0.08551.426*Inde, loan amount)0.08551.426*Inde, loan amount)0.042995.408*0.34123.936*0.4830Interest rate)0.04020.4720.0989Int, loan amount)0.42995.408*0.34123.936*0.4830Interest rate)0.04020.4720.0989Int, interest rate)-0.9384-3.414*-0.8715-3.643-0.81630.1Interest rate)-1.54070.1-1.54070.10.47621.5570.47971.813*4.39200.1-0.6346-2.122t-0.8283-3.229*-0.74050.1-0.3341-0.965 <th c<="" td=""></th>	

Table 47.	Estimated Coefficients of the Transaction-Costs Function, w	ith
	Alternative Proxies for Firm Size. Dependent Variable: InT	

OLS estimation. S,  $D_1$ ,  $D_2$ ,  $U_1$ ,  $U_2$ , and  $U_3$  are dummy variables. Levels of significance: \*, 0.01 +, 0.05 °, 0.10 <u>a</u>/ lnW was set equal to zero in observations with no hired labor.

Table 48. Transaction Costs per Lempira  $(\tau)$ : Estimated Values of  $\tau$ , Changes in  $\tau$  and in Total Borrowing Costs (i+ $\tau$ ) with Increases in the Explicit Interest Rate (i). Results by Loan Use and Loan Size for the Development Bank

Loan Use/Loan Size	Estimated Value of Transaction Costs per Lempira 2 7, %	Estimated Total Borrowing Costs (i+î), %	Change in τ with a one-point increase in the interest rate δτ/δi, pct. points	Change in Total Borrowing Costs with a one-point increase in the interest rate ð(i+t)/ði, pct. points
Basic Grains				
Lps. 2,000	3.84	16.84	-0.164	0.836
Lps. 20,000	0.51	13.51	-0.033	0.967
Export Crops				
Lps. 2,000	2.13	15.31	-0.091	0.909
Lps. 20,000	0.28	13.28	-0.018	0.982
Livestock				
Lps. 2,000	4.19	17.19	-0.179	0.821
Lps. 20,000	0.56	13.56	-0.036	0.964

Source: Extended-model results presented in table 6.5. Estimates evaluated at geometric means of farm area (35 Has.) and interest rate (13%).

.

Table 49. Transaction Costs per Lempira ( $\tau$ ): Estimated Values of  $\tau$ , Changes in  $\tau$  and in Total Borrowing Costs (i+ $\tau$ ) with Increases in the Explicit Interest Rate (i). Results by Loan Use and Loan Size for Private Banks

.

Loan Use/	'Loan Size	Estimated Value of Transaction Costs per Lempira $\hat{\tau}$ , %	Estimated Total Borrowing Costs (i+î), %	Change in τ with a one-point increase in the interest rate δτ/δi, pct. points	Change in Total Borrowing Costs with a one-point increase in the interest rate ð(i+t)/ði, pct. points
Basic Gra	ins				
Lps.	2,000	7.77	20.77	-1.595	-0.595
Lps.	20,000	1.04	14.04	-0.236	0.764
Export Cr	ops				
Lps.	2,000	4.30	17.30	-0.882	0.118
Lps.	20,000	0.57	13.57	-0.131	0.869
Livestock	:				
Lps.	2,000	8.47	21.47	-1.739	-0.739
Lps.	20,000	1.13	14.13	-0.257	0.743

Source: Extended-model results presented in table 6.5. Estimates evaluated at geometric means of farm area (35 Has.) and interest rate (13%).

. .